

OPEN ACCESS

Precision luminosity measurements at LHCb

To cite this article: The LHCb collaboration 2014 JINST 9 P12005

View the article online for updates and enhancements.

Related content

- Absolute luminosity measurements with the LHCb detector at the LHC The LHCb Collaboration
- Measurement of visible cross sections in proton-lead collisions at $s_{\rm NN}=5.02$ TeV in van der Meer scans with the ALICE detector The ALICE collaboration
- Characterisation and mitigation of beaminduced backgrounds observed in the ATLAS detector during the 2011 protonproton run

Recent citations

- <u>Large Hadron Collider momentum</u> <u>calibration and accuracy</u> E. Todesco and J. Wenninger
- <u>Prospects for Quarkonium Measurements</u> in p-A and A-A Collisions at the LHC Michael Winn
- <u>Beam imaging and luminosity calibration</u> M. Klute *et al*

RECEIVED: October 2, 2014 ACCEPTED: November 11, 2014 PUBLISHED: December 5, 2014

Precision luminosity measurements at LHCb



inst

The LHCb collaboration

E-mail: rosen.matev@cern.ch

ABSTRACT: Measuring cross-sections at the LHC requires the luminosity to be determined accurately at each centre-of-mass energy \sqrt{s} . In this paper results are reported from the luminosity calibrations carried out at the LHC interaction point 8 with the LHCb detector for $\sqrt{s} = 2.76$, 7 and 8 TeV (proton-proton collisions) and for $\sqrt{s_{NN}} = 5$ TeV (proton-lead collisions). Both the "van der Meer scan" and "beam-gas imaging" luminosity calibration methods were employed. It is observed that the beam density profile cannot always be described by a function that is factorizable in the two transverse coordinates. The introduction of a two-dimensional description of the beams improves significantly the consistency of the results. For proton-proton interactions at $\sqrt{s} = 8$ TeV a relative precision of the luminosity calibration of 1.47% is obtained using van der Meer scans and 1.43% using beam-gas imaging, resulting in a combined precision of 1.16%. This represents the most precise luminosity measurement achieved so far at a bunched-beam hadron collider.

KEYWORDS: Pattern recognition, cluster finding, calibration and fitting methods; Instrumentation for particle accelerators and storage rings - high energy (linear accelerators, synchrotrons)

ARXIV EPRINT: 1410.0149

Contents

1	Intro	oduction	1
2	Expe	erimental setup and data-taking conditions	3
3	Bunc	ch current normalization	5
	3.1	Bunch population measurement	7
	3.2	Ghost charge	8
	3.3	Total uncertainty	10
4	Rela	tive luminosity calibration	11
	4.1	Interaction rate determination	11
	4.2	Systematic uncertainties	15
5	Forn	nalism for the luminosity of colliding beams	22
	5.1	Beam overlap measurement methods	23
	5.2	Luminosity in the case of purely Gaussian beams	24
	5.3	Double Gaussian shape model	26
6	Bean	n-gas imaging method	26
	6.1	Data-taking conditions and event selection	27
	6.2	Vertex position resolution	30
		6.2.1 Resolution for beam-beam interaction vertices	30
		6.2.2 Resolution for beam-gas interaction vertices	32
		6.2.3 Resolution function for a sample of vertices	34
	6.3	Measurement of the overlap integral	35
	6.4	Generic simulation	39
	6.5	Evidence of non-factorizable beam shapes	40
	6.6	Results and systematic uncertainties	43
	6.7	Beam-gas imaging results at lower energies	50
7	Van	der Meer scan method	53
	7.1	Experimental conditions	53
	7.2	Overlap integral model	56
	7.3	Cross-section determination	58
	7.4	Rate measurement	60
	7.5	Length scale	62
	7.6	Beam-beam effects	65
	7.7	Fit model uncertainty	68
	7.8	Reproducibility	70
	7.9	Results	75
	7.10	Summary of other van der Meer scan calibrations	75

The LHCb collaboration

1 Introduction

The determination of the cross-section of a given subatomic process at high energy colliding-beam experiments is generally performed by the measurement of an interaction rate. To determine such a cross-section on an absolute scale, a measurement of the colliding-beam luminosity must be performed. The requirement for the accuracy on the value of the cross-section is usually driven by the precision of theoretical predictions for the process. At the LHCb experiment [2] the cross-section measurements for the production of vector bosons (*Z* and *W*) [3, 4] and the exclusive two-photon production of muon pairs [5] motivate an accuracy of order 1-2% for the luminosity calibration.

The instantaneous luminosity *L* is defined by the relation between the reaction rate *R* and the process cross-section σ

$$R = L\sigma. \tag{1.1}$$

The instantaneous luminosity for a colliding bunch pair can be written as [6-8]

$$L = N_1 N_2 \, v_{\rm rev} \, \Omega \,, \tag{1.2}$$

where N_1 and N_2 are the populations of the colliding bunches of beam 1 and beam 2, v_{rev} is the revolution frequency and the beam overlap integral Ω embodies the passage of the two bunches with spatial particle density distributions $\rho_1(x, y, z, t)$ and $\rho_2(x, y, z, t)$ accross each other. In the limit of ultra-relativistic particles (velocity close to the speed of light, $v \approx c$), crossing at small angle, the beam overlap integral is given by

$$\Omega = 2c \int \rho_1(x, y, z, t) \rho_2(x, y, z, t) dx dy dz dt.$$
(1.3)

Methods for absolute luminosity determination can be classified as being either direct or indirect. Indirect methods are e.g. the use of the optical theorem to make a simultaneous measurement of the elastic and total cross-sections [9, 10], or the comparison with a process for which the absolute cross-section is known, either from theory or by a previous direct measurement. Direct methods derive the luminosity from the measurements of the colliding beam parameters. The analysis described in this paper relies on two direct methods to determine the absolute luminosity calibration: the "van der Meer scan" method (VDM) [11–15] and the "beam-gas imaging" method (BGI) [11, 16], the latter making use of unique capabilities of the LHCb experiment. The VDM method exploits the ability to move the beams in both transverse coordinates with high precision and thus to scan the overlap integral of the colliding beams at different relative beam positions while measuring a relative rate. This method, which was first applied at the CERN ISR [12], is also being used by the other LHC experiments [17–19]. The BGI method is based on reconstructing vertices of interactions between beam particles and gas nuclei in the beam vacuum to measure the angles, positions and shapes of the individual beams without displacing them. The shapes obtained with these data are constrained by the distribution of vertices measured with beam-beam interactions. In both methods, data taken with the LHCb detector located at interaction point (IP) 8 are used in conjunction with data from the LHC beam instrumentation.

At the LHC, from 2009 to 2013, several luminosity calibration measurements were performed with a gradually improving precision. Different nucleon-nucleon centre-of-mass energies \sqrt{s} and different beam species were used: protons on protons (pp), lead on lead (Pb-Pb) and protons on lead (*p*Pb or Pb*p*, where the first/second beam species applies to beam 1/beam 2 in the standard LHC definition [20], see figure 1). First LHC luminosity calibrations were obtained by LHCb using pp collision data collected at the end of 2009 at $\sqrt{s} = 900 \,\text{GeV}$ [21] and in 2010 at $\sqrt{s} = 7 \text{ TeV} [11, 22, 23]$ with an accuracy that was limited by the systematic uncertainties associated with the normalization of the colliding bunch populations [24, 25]. Recent detailed studies of the LHC beam current transformers (BCTs) significantly reduced these uncertainties [26–28], thus facilitating an improvement of the final precision of the luminosity calibration. In this paper results are reported from luminosity calibration experiments carried out at the LHC IP8 with the LHCb detector from 2011 to 2013, for $\sqrt{s} = 2.76$, 7 and 8 TeV in pp collisions and for $\sqrt{s_{NN}} = 5$ TeV in pPb and Pbp collisions. In addition to performing luminosity calibration measurements, LHCb provided related beam-gas interaction measurements as a service to the other LHC experiments. This included the measurement of the total charge outside the nominally filled slots ("ghost charge", see section 3) and of the single beam size as a function of time during the VDM scans of these other experiments.

The precision of the luminosity calibration in the LHCb experiment is now limited by the systematic uncertainties of the beam overlap determination. These systematic uncertainties are different, to a large extent, for the VDM and BGI methods. Therefore, the comparison provides an important cross check of the results. The calibration measurements obtained with the VDM and BGI methods are found to be consistent and are averaged for the final result.

Since the absolute calibration can only be performed during specific running periods, a relative normalization method is needed to transport the results of the absolute luminosity calibration to the complete data-taking period. To this end, several observables are used, each one corresponding to an effective visible cross-section σ_{vis} . The corresponding cross-section is calibrated for each variable using the measurements of the absolute luminosity during specific data-taking periods. The integrated luminosity for an arbitrary period of data taking is then obtained from the accumulated counts of a calibrated visible cross-section.

In the present paper we first describe briefly the LHCb experimental setup and data-taking conditions in section 2, emphasizing the aspects relevant to the analysis presented here. Section 3 is devoted to the normalization of the bunch population, while the methods used for the relative normalization technique are given in section 4. In section 5 we introduce the luminosity formalism for colliding beams. The determination of the luminosity with the BGI method is detailed in section 6 and with the VDM scan method in section 7. The combination of the results and conclusions are given in section 8.



Figure 1. Schematic view of the current LHCb detector. LHC beam 1 (beam 2) enters from the left (right) side of the figure. The labels indicate sub-detectors: vertex locator (VELO), RICH1, RICH2 (ring imaging Cherenkov detectors 1 and 2), TT (tracker Turicensis), T1, T2, T3, (tracking stations 1, 2 and 3), SPD/PS (scintillating pad detector / preshower detector), ECAL (electromagnetic calorimeter), HCAL (hadron calorimeter), and M1, M2, M3, M4, M5 (muon stations 1, 2, 3, 4, and 5) (drawing from ref. [1]).

2 Experimental setup and data-taking conditions

The LHCb detector (figure 1) is a single-arm forward spectrometer with a polar angular coverage of approximately 15 to 300 mrad in the horizontal (bending) plane, and 15 to 250 mrad in the vertical plane. It is designed for the study of particles containing b or c quarks and is described in detail elsewhere [2].

The apparatus contains tracking detectors, ring-imaging Cherenkov detectors, calorimeters, and a muon identification system. The tracking system comprises the vertex locator (VELO) surrounding the beam interaction region, a tracking station upstream of the dipole magnet and three tracking stations located downstream of the magnet. Particles traversing the spectrometer experience a bending-field integral of around 4 Tm.

The VELO plays an essential role in the application of the VDM and BGI luminosity calibration methods at LHCb. It consists of two horizontally retractable halves, each having 21 modules of radial and azimuthal silicon-strip sensors in a half-circle shape (figure 2). Two additional stations (*Pile-Up System*, PU) upstream of the VELO tracking stations are mainly used in the hardware trigger. The VELO has a large acceptance for beam-beam interactions owing to its many layers of silicon sensors and their close proximity to the beam line. During nominal operation, the distance between the closest sensor strip and the beams is only 8.2mm. During injection and beam adjust-



Figure 2. Sketch of the VELO sensor positions. The luminous region is schematically depicted with a filled ellipse. Its longitudinal extent, RMS $\sigma = 53 \text{ mm}$, is indicative. Sensors measuring the *R* (ϕ) coordinates are shown as blue (red) lines. The LHC beam of ring 1 (2) enters from the left (right) on this sketch. The coordinate system is defined in section 5 (drawing from ref. [1]).

ments, the two VELO halves are kept apart in a retracted position 30mm away from the beams. They are brought to their nominal position close to the beams during stable beam periods only. More details about the VELO can be found in ref. [29].

The LHCb trigger system [30] consists of two separate levels: a hardware trigger, which is implemented in custom electronics, and a software trigger, executed on a farm of commercial processors. The hardware trigger is designed to have an accept rate of 1 MHz and uses information from the PU sensors of the VELO, the calorimeters and the muon system. These detectors send information to the hardware decision unit, where selection algorithms are run synchronously with the 40 MHz LHC bunch crossing. For every nominal bunch-crossing slot (*i.e.* each 25 ns) the hardware decision unit sends its information to the LHCb readout supervisor, which distributes the synchronous hardware trigger decision to all front-end electronics. For every positive hardware decision the full event information of all sub-detectors is sent to the processor farm and is made available to the software trigger algorithms.

For luminosity calibration and monitoring, a trigger strategy is adopted to select beam-beam inelastic interactions and interactions of the beams with the residual gas in the vacuum chamber. Events are collected for the four bunch-crossing types: two colliding bunches (bb), one beam 1 bunch with no beam 2 bunch (be), one beam 2 bunch with no beam 1 bunch (eb) and nominally empty bunch slots (ee). Here "b" stands for "bunch" and "e" stands for "empty". The first two categories of crossings produce particles in the forward direction and are triggered using calorimeter information. An additional PU veto is applied for be crossings. Crossings of the type eb produce particles in the backward direction, are triggered by demanding a minimal hit multiplicity in the PU, and are vetoed by calorimeter activity. The trigger for ee crossings is defined as the logical OR of the conditions used for the be and eb crossings in order to be sensitive to background from both beams. In addition to these specific triggers, a decision based on a hardware trigger sensitive to any activity in the PU and calorimeter is available. The latter hardware trigger configuration is used for most measurements described in this paper. Events are then further selected by the software trigger based on track and vertex reconstruction using VELO hits. During VDM scans specialized trigger configurations are defined that optimize the data taking for these measurements (see section 7).

The reconstruction of interaction vertices (also called "primary vertices", PVs) is performed using standard LHCb algorithms [31]. The initial estimate of the PV position is based on an iter-

ative clustering of tracks. For each track the distance of closest approach (DOCA) with respect to all other tracks is calculated and tracks are clustered into a PV candidate if their DOCA is less than 1 mm. An initial position of the PV is obtained from the weighted average of the points of closest approach between all track pairs, after removing outliers. The final PV coordinates are determined by iteratively improving the position determination with an adaptive, weighted, least-squares fit. Participating tracks are assigned weights depending on their impact parameter with respect to the PV. The procedure is repeated for all possible track clusters, excluding tracks from previously reconstructed PVs, retaining only those with at least five tracks. For the analysis described here only PVs with a larger number of tracks are used since they provide better position resolution. For the study of beam-gas interactions only PVs with at least ten tracks are used and at least 25 tracks are applied as described below.

The full list of luminosity calibrations discussed in this paper is summarized in table 1. The table is divided into five sections following the different nucleon-nucleon centre-of-mass energies and beam species involved. A first measurement with intentionally enlarged beta functions at the IP ($\beta^* = 10$ m) was performed in October 2011 with pp collisions at $\sqrt{s} = 7$ TeV. Several fills in 2012 were dedicated to luminosity calibration for pp collisions at $\sqrt{s} = 8$ TeV, although only the measurements in July and November were performed with large β^* . The April measurements were performed in non-optimal conditions, with focused beams ($\beta^* = 3 \text{ m}$) and with a tilted crossing plane (a non-zero vertical half crossing angle ϕ_v), and are therefore primarily used for the VDM calibration method and to cross-check the effects on the BGI method of the finite vertex resolution. Calibrations for *p*Pb and Pb*p* were conducted in January 2013 at $\sqrt{s_{NN}} = 5$ TeV with VDM scans only. Further pp calibrations were performed at $\sqrt{s} = 2.76$ TeV in February 2013, exclusively using the BGI method. The number of bunches per beam is also given in the table. No active gas injection was used to enhance the beam-gas rates and the end of 2011, though a first rate increase was obtained in October 2011 by degrading the beam vacuum by switching off the VELO ion pumps. Thus, three configurations of the VELO vacuum state have been used, one where the vacuum pumps are operating (normal state), one where the VELO ion pumps were switched off, and one where, in addition to running with pumps off, neon gas was injected into the VELO vacuum chamber (see section 6). All pp BGI calibration measurements of 2012 and 2013 took advantage of gas injection. During VDM calibration scans, gas injection was always off. In all pp calibration runs discussed here the initial bunch populations ranged between 0.6 and 1.1×10^{11} particles. For the pPb and Pbp runs they varied between 1 and 2×10^{10} elementary charges (for both beam species). Calibration experiments with the VDM method included a variety of beam displacement sequences. The details of these individual experiments are given in the section devoted to the VDM analysis (section 7). In fills 3503, 3537 and 3540, no luminosity calibration was performed at IP8, though the LHCb experiment provided ghost charge and beam size measurements for the benefit of the luminosity calibrations conducted in other LHC experiments.

3 Bunch current normalization

Various detector systems are used to determine with high precision the population of particles in each colliding and non-colliding bunch in the LHC. The longitudinal structure of the LHC beams

Table 1. Dedicated LHC calibration fills during which LHCb performed the luminosity calibrations described in this paper or ghost charge and beam size measurements for other LHC experiments. In most calibration measurements the number of bunches per beam was the same for beam 1 and beam 2. For the *p*Pb and Pb*p* fills where this was not the case two numbers are given, the first for the number of beam 1 bunches, the second for the number of beam 2 bunches. The number of colliding bunches at LHCb is indicated in parentheses (fifth column). Half crossing angles ϕ_x and ϕ_y , and β^* are given as nominal values. The VELO vacuum state during BGI measurements is indicated in the column "Gas injection". A state "off" means that gas injection was turned off and the VELO ion pumps were turned off, which resulted in a residual vacuum pressure about a factor four higher than nominal. A state "on" indicates that neon gas was being injected into the beam vacuum. During VDM measurements the state was always "off".

Period	Fill	$\phi_x(\phi_y)$	β^*	Bunches	Gas	Luminosity		
		(µrad)	(m)	per beam	injection	calibration		
Fills with $p p$ at $\sqrt{s} = 8$ TeV								
Apr 2012	2520	236 (90)	3	48 (6)	on	BGI		
Apr 2012	2523	236 (90)	3	52 (24)	on	BGI, VDM		
Jul 2012	2852	456 (0)	10	50 (16)	on	BGI,		
Jul 2012	2853	456 (0)	10	35 (16)	on	BGI, VDM		
Jul 2012	2855	456 (0)	10	48 (6)	on	BGI		
Jul 2012	2856	456 (0)	10	48 (6)	on	BGI		
Nov 2012	3311	456 (0)	10	39 (6)	on	BGI		
Nov 2012	3316	456 (0)	10	39 (6)	on	BGI		
Fills with <i>p</i>	<i>p</i> at \sqrt{s}	$=7 \mathrm{TeV}$						
Oct 2011	2234	270 (0)	10	36 (16)	off	BGI, VDM		
Fills with <i>p</i>	p at \sqrt{s}	$= 2.\overline{76 \text{TeV}}$						
Feb 2013	3555	855 (0)	10	100 (22)	on	BGI		
Feb 2013	3562	855 (0)	10	39 (6)	on	BGI		
Feb 2013	3563	855 (0)	10	39 (6)	on	BGI		
Fills with p Pb at $\sqrt{s_{NN}} = 5 \text{ TeV}$								
Jan 2013	3503	456 (0)	2	272+338 (38)	off	other experiments		
Jan 2013	3505	456 (0)	2	272+338 (38)	off	VDM		
Fills with Pb p at $\sqrt{s_{NN}} = 5$ TeV								
Feb 2013	3537	456 (0)	2	314+272 (22)	off	other experiments		
Feb 2013	3540	456 (0)	2	314+272 (22)	off	other experiments		
Feb 2013	3542	456 (0)	2	338 (39)	off	VDM		

is shaped by the 400MHz radio frequency (RF) system. Both LHC rings are filled with bunches at locations ("RF buckets") defined by the RF system and are organized in "slots", which contain each ten consecutive buckets. Ideally, only one of these buckets is filled with a bunch, called the "main bunch", the other nine are nominally empty. Only a subset of the slots are filled in a given filling scheme. In each filled slot, the main bunch occupies the same bucket number. In practice, a small fraction (typically $< 10^{-3}$) of the charge in a slot occupies nominally empty buckets and are called "satellite" bunches. Additionally, also the nominally empty slots may contain charges. The total charge outside the filled slots is called "ghost charge".

3.1 Bunch population measurement

To measure the population in the main bunches, specific instruments are used to determine the overall circulating charge, the relative charge in the filled bunches, the fraction of the charge in the satellite buckets of the filled slots and the fraction of ghost charge. Four independent directcurrent current-transformers (DCCTs), two per ring, are used to measure the total beam current circulating in each LHC ring. The DCCT is designed to be insensitive to the time structure of the beam [32]. Two fast bunch current transformers (FBCTs), one per ring, provide a relative measure of the individual charges on a slot-by-slot basis [33]. The FBCT is designed to produce a signal proportional to the charge in each 25 ns LHC bunch slot. The captured particles of an LHC bunch are contained within an RF bucket of 1–1.5 ns length at ± 2 standard deviations [34]. Since 2012, one longitudinal density monitor (LDM) [35, 36] per LHC ring is available for detecting synchrotron radiation photons emitted by particles deflected in a magnetic field. The LDMs are used to obtain the longitudinal beam charge distribution with a time resolution of about 90 ps to resolve the charge distribution in individual RF buckets. Finally, the ghost charge fraction is obtained by counting beam-gas interactions with the LHCb detector in nominally empty (ee) compared to the rates in nominally filled (bb, be and eb) bunch crossings.

Previous LHC luminosity calibration experiments showed that one of the dominant uncertainties arises from the normalization of the bunch population product N_1N_2 . As a consequence, a detailed study of the normalization was carried out using data from the LHC beam current transformers (BCTs) and from the LHC experiments. A dedicated analysis procedure was defined and bunch population uncertainties were quantified for the 2010 LHC luminosity calibration measurements [24, 25]. The precision was limited by the understanding of the BCT data at that stage. Since then, a number of additional tests were carried out that significantly improved the understanding of the bunch current measurements. Careful calibration measurements and systematic studies of the DCCTs improved the dominant uncertainty by an order of magnitude [26, 37]. Uncertainties on the beam current product for the 2011–2013 measurements are well below 1% and are given in more detail below.

The accuracy of the relative bunch populations determined with the FBCT is cross-checked against results from other measurements, such as those obtained from the ATLAS BPTX button pick-up [38] and those derived from the LHCb beam-gas interaction rates [27]. The sum of the FBCT signals of all nominally filled bunch slots is normalized to the total number of particles measured by the DCCTs after subtraction of the ghost charge and satellite charges,

$$N_{j,i} = \frac{I_{\text{DCCT},j}}{v_{\text{rev}}Z_j e} \cdot (1 - f_{\text{ghost},j}) \cdot \frac{S_{\text{FBCT},j,i}}{\sum_i S_{\text{FBCT},j,i}} \cdot (1 - f_{\text{sat},j,i}), \qquad (3.1)$$

defining $N_{j,i}$ as the bunch population of the nominally filled RF bucket of bunch slot *i* of beam *j*, and $I_{\text{DCCT},j}$ as the current measured by the DCCTs and $Z_j e$ the charge of a beam particle (82*e* for Pb beams). The sum runs over all nominally filled slots and the $S_{\text{FBCT},j,i}$ are the signals measured by the FBCT of ring *j*. The ghost charge fraction is denoted $f_{\text{ghost},j}$ and the fraction of the charge in satellite bunches $f_{\text{sat},j,i}$ for beam *j* and slot *i*.

Ghost charge fractions for the 2011–2013 LHC luminosity calibration fills range up to about 2.5%. As mentioned above, these measurements are performed with the LHCb detector. The results and methods are described in detail in section 3.2.

Satellite charges have been observed in various ways with the LHC detectors by detecting longitudinally displaced collisions (see for example ref. [24]). The total satellite population fraction ($f_{\text{sat},j,i}$) in a bunch slot is usually less than a percent compared to the associated main bunch population. Nevertheless, it needs to be quantified to obtain a precise measurement of the bunch population that actually contributes to the luminosity.

3.2 Ghost charge

The determination of the ghost charge from the beam-gas interaction rate measurements was pioneered in a previous LHCb luminosity calibration [39]. The results presented here benefit from the larger number of beam-gas events obtained with neon gas injection in the beam vacuum chamber, which allows the uncertainty to be reduced and provides a more detailed determination of the charge distribution over the LHC ring in a shorter time. Systematic uncertainties are further reduced by a better trigger efficiency calibration. The ghost charge measurement is based on the same data sample as used for the BGI analysis. The trigger requirements are described in section 6.

To ensure that each vertex is a result of a beam-gas interaction and is assigned to the correct beam, several selection criteria are applied [37] that are based on the track directions (all forward for beam 1, all backward for beam 2), on the transverse position (to exclude interactions with material in the vicinity of the beams), on the longitudinal position and on the vertex track multiplicity.

The LHCb data acquisition is synchronised with the LHC RF system with a granularity of 25 ns. The sampling phase of the detectors relative to the LHC clock is optimized to provide the highest efficiency for nominally filled RF buckets, but the trigger efficiency may vary across the 25 ns bunch slot. Since the ghost charge is distributed over all RF buckets inside the 25 ns slots, the trigger efficiency must be known for all possible phases. A first efficiency measurement was performed in 2010 [11, 39], resulting in a ghost charge uncertainty of about 20% per beam. A new dedicated measurement was performed in 2012 with the aim of reducing this uncertainty by acquiring data for more clock phases and by using neon gas injection to increase the statistical accuracy. The efficiency is determined by measuring dead-time corrected beam-gas interaction rates from non-colliding bunches at different clock phases and comparing them with the standard phase (zero clock shift). The absolute rate is measured as function of clock shift in 2.5 ns steps. The beam intensity decay observed during the measurement is taken into account.

If a beam-gas interaction occurs near the bunch slot edges, that is, the originating charge is near the previous or next clock cycle, the resulting VELO sensor signals may be sufficiently long that they are also seen in the neighbouring clock cycle. Therefore, depending on where the charges are located within the 25 ns bunch slot, some vertices are counted twice and thus bias the ghost charge or trigger efficiency measurement. To take this double-counting effect into account, the efficiency is measured including all beam-gas events or, alternatively, excluding double-counted vertices. In addition, the efficiency is measured for different vertex track multiplicity thresholds (from 8 to 12 tracks) to account for the slightly different trigger conditions used for this measurement as compared to later BGI measurements. The results of the trigger efficiency calibration are shown in figure 3 and the values averaged over the 25 ns clock cycle are summarized in table 2.

The increased rate of beam-gas interaction data acquired with neon gas injection enables a measurement of the charge distribution over the ring circumference. In figure 4 the ghost charge per 25 ns slot is shown as function of slot number (BCID) using data from fill 2520 as an example.



Figure 3. Relative beam-gas trigger efficiency as function of LHCb detector clock shift with respect to the LHC reference timing, (left) including or (right) excluding double-counted beam-gas interaction vertices. The efficiency is shown relative to the value at the nominal clock setting (*i.e.* zero shift). The shaded areas indicate the variation between the results for thresholds corresponding to 8 and 12 tracks. The data points, appearing in groups of three, indicate measurements applying the 8, 10 and 12 track thresholds.

Table 2. Relative beam-gas trigger efficiency for the ghost charge measurement assuming a constant charge distribution within a bunch slot.

Beam	Efficiency average ϵ_j				
	including double-counting	excluding double-counting			
1	1.05 ± 0.03	0.93 ± 0.02			
2	0.90 ± 0.01	0.86 ± 0.01			

Ghost charges are observed around the nominally filled bunches and are mostly absent further than about 20 slots away from filled bunches.

Ghost charge fractions during LHC luminosity calibration fills are measured in four-minute time bins. For each time bin the ghost charge fraction is evaluated with both counting methods: including and excluding double-counted vertices and applying the corresponding average trigger efficiency of table 2. If all charges are evenly spread within their bunch slot, each evaluation would provide a different result before efficiency correction, but the same result after efficiency correction. After efficiency correction the differences between the two evaluations are small. This observation is in agreement with the LDM measurements [40], which show that the ghost charge tends to be spread evenly over all RF buckets of a bunch slot. The LDM information on the charge distribution within the nominally empty bunch slots is not used in the results except for fill 3542 during which the trigger was not configured to perform this measurement. The average of the two efficiency-corrected evaluations is taken as final value for the ghost fraction, while their difference is taken as systematic uncertainty. The trigger efficiency uncertainty taken from table 2 is added in quadrature with the systematic uncertainty. A summary of all ghost charge measurements performed for the special luminosity fills in 2011, 2012 and 2013 is provided in table 3.

With the exception of intermediate energy fills at $\sqrt{s} = 2.76$ TeV, ghost charge fractions are



Figure 4. Histogram of ghost charge distribution as a function of LHC bunch slot number (BCID) in fill 2520 for beam 1 (green) and beam 2 (yellow). The BCID position of nominally filled bunches is indicated as small vertical blue and red lines for beam 1 and beam 2, respectively. The ghost charge distribution is shown for the (top) ring circumference and (bottom left) first 400 and (bottom right) last 400 BCIDs. Ghost charges are mostly absent in regions without nominally filled bunches. Note that only ee BCIDs are displayed.

stable within $\pm 10\%$ during a fill and the total beam intensity can be corrected with good accuracy using an average value for a fill. In this case the RMS over the fill, given in table 3, should be taken into account in the uncertainty. On the contrary, for the intermediate-energy fills, an increase in the ghost charge fraction over time warrants a time dependent correction to the total beam intensity. As an example, the difference in ghost charge evolution seen between high- and intermediate-energy fills is shown in figure 5 comparing the long fill 2855 at $\sqrt{s} = 8$ TeV and fill 3563 at $\sqrt{s} = 2.76$ TeV.

3.3 Total uncertainty

A summary of the bunch population product uncertainties is given in table 4 for each luminosity calibration fill. The systematic uncertainties for the ghost charge corrections of the two beams described in the previous section are assumed to be fully correlated with each other, *i.e.* the final



Figure 5. Ghost charge fractions for (left) fill 2855 and (right) fill 3563. Fill 2855 with $\sqrt{s} = 8 \text{ TeV}$ shows a constant or slightly decreasing ghost charge fraction throughout the fill lasting about 9 hours. Fill 3563 ($\sqrt{s} = 2.76 \text{ TeV}$) shows an important increase of ghost charge over a period of 4 hours.

ghost charge uncertainty on the bunch population product is the linear sum of the ghost charge systematic uncertainty of each beam.

The satellite fractions provided by the LDM [40] are measured at the beginning and at the end of the fill. Here, the average of these two measurements is used. The average satellite fractions for all colliding bunches and fills with $\beta^* = 10$ m at $\sqrt{s} = 8$ TeV are 0.25% and 0.18% for beam 1 and beam 2, respectively. The uncertainty on the satellite fraction correction is taken as the full difference between the fractions measured at the beginning and end of fill. Assuming the uncertainties are fully correlated between the two beams, the uncertainty on the population product due to the satellite fraction correction is taken as the linear sum of the average uncertainties per beam, and is given as the average per fill in table 4.

The beam population product normalization uncertainty is dominated by the DCCT measurement. All fills listed in table 4 are subject to the same procedure to evaluate the beam population product uncertainty. For fills with $\beta^* = 10$ m and $\sqrt{s} = 8$ TeV, the average uncertainty on the bunch population product weighted with the number of measurements amounts to 0.22% at 68% confidence level.

4 Relative luminosity calibration

Absolute luminosity calibrations are performed during short periods of data-taking. To be able to determine the integrated luminosity for any data sample obtained during long periods, the interaction rate of standard processes is measured continuously. The effective cross-section corresponding to these standard processes is determined by counting the visible interaction rates during the specific periods when the absolute luminosity is calibrated.

4.1 Interaction rate determination

The luminosity is proportional to the average number of visible proton-proton interactions per beam-beam crossing, μ_{vis} . The subscript "vis" is used to indicate that this particular definition of

Table 3. Measurements of ghost charge fractions for all luminosity calibration fills in 2011, 2012 and 2013. The systematic uncertainty is assumed to be fully correlated between the two beams. Therefore, the final systematic uncertainty on the beam intensity product due to the ghost charge correction is a linear sum of the ghost charge systematic uncertainty of each beam. Proton-lead fills were acquired without neon gas injection and have a larger statistical uncertainty. For fill 3542 the ghost charge was only measured using the LHC LDMs.

Fill	Beam 1				Beam 2			
	$f_{\rm ghost,1}$	RMS	uncertainty		$f_{\rm ghost,2}$	RMS	uncertainty	
	(%)	in fill	syst.	stat.	(%)	in fill	syst.	stat.
Fills with	pp at \sqrt{s}	= 8 TeV						
2520	0.30	0.01	0.02	0.001	0.35	0.01	0.01	0.002
2523	0.50	0.02	0.03	0.001	0.44	0.01	0.01	0.001
2852	0.62	0.01	0.04	0.002	0.53	0.01	0.01	0.002
2853	0.40	0.01	0.02	0.002	0.28	0.01	0.01	0.002
2855	0.22	0.01	0.01	0.001	0.23	0.01	0.01	0.001
2856	0.24	0.01	0.01	0.001	0.22	0.02	0.01	0.001
3311	0.15	0.03	0.01	0.001	0.06	0.01	0.01	0.001
3316	0.19	0.01	0.01	0.001	0.06	0.01	0.01	0.001
Fills with pp at $\sqrt{s} = 7$ TeV								
2234	0.84	0.10	0.05	0.012	0.76	0.13	0.02	0.015
Fills with	pp at \sqrt{s}	= 2.761	TeV					
3555	0.58	0.14	0.04	0.001	0.33	0.03	0.01	0.001
3562	0.78	0.30	0.05	0.003	0.52	0.22	0.01	0.003
3563	1.28	0.55	0.08	0.002	0.88	0.35	0.02	0.002
Fills with <i>p</i> Pb at $\sqrt{s_{NN}} = 5$ TeV								
3503	0.18	0.04	0.01	0.005	0.50	0.13	0.01	0.011
3505	0.29	0.05	0.02	0.007	0.66	0.12	0.02	0.015
Fills with Pbp at $\sqrt{s_{NN}} = 5$ TeV								
3537	0.50	0.12	0.03	0.010	0.88	0.11	0.02	0.015
3540	0.73	0.09	0.05	0.019	0.17	0.05	0.01	0.014
3542	n.a.				n.a.			

interaction rate does not need to have a simple physics interpretation. Any interaction rate that can be measured under stable conditions can be used as such a relative luminosity monitor. The interaction rates are acquired and stored together with the physics data as "luminosity data". During further processing of the data the relevant luminosity information is kept in the same storage entity. Thus, it remains possible to select only part of the full data set for analysis and still keep the capability to determine the corresponding integrated luminosity.

Triggers, which initiate the full readout of the LHCb detector, are created for a random choice of beam crossings at a fixed average frequency. These are called "luminosity triggers". During normal physics data-taking, the overall rate is chosen to be 1000 Hz. Of this rate, 70% is assigned to slots where two bunches cross (bb), 15% to slots with only a beam-1 bunch (be), 10% to those

Fill	Sources of u	Sources of uncertainty on bunch population product					
	DCCTs	Ghost fractions	Satellite fractions				
Fills with pp a	at $\sqrt{s} = 8 \mathrm{TeV}$						
2520	0.26	0.029	n.a.				
2523	0.22	0.043	n.a.				
2852	0.19	0.049	0.097				
2853	0.24	0.032	0.019				
2855	0.21	0.019	0.021				
2856	0.21	0.020	0.031				
3311	0.22	0.011	0.011				
3316	0.23	0.013	0.011				
Fills with pp at $\sqrt{s} = 7$ TeV							
2234	0.24	0.064	0.25 [28]				
Fills with pp at $\sqrt{s} = 2.76$ TeV							
3555	0.51	0.047	0.230				
3562	3562 0.22		0.020				
3563	0.23	0.101	0.024				
Fills with <i>p</i> Pb at $\sqrt{s_{NN}} = 5 \text{ TeV}$							
3505	0.31	0.137	0.070				
Fills with Pbp at $\sqrt{s_{NN}} = 5 \text{ TeV}$							
3542	0.34	0.192	0.092				

Table 4. Relative uncertainties (in percent) on colliding-bunch population products for all relevant fills.

with only a beam-2 bunch (eb) and the remaining 5% to slots that are empty (ee). The events taken for crossing types other than bb are used for background subtraction and beam monitoring.

Interaction rates are measured by processing the random luminosity triggers and these rates are stored in a small number of "luminosity observables". The set of luminosity observables comprises the number of vertices and tracks reconstructed in the VELO, the number of muons reconstructed in the muon system, the number of hits in the PU and in the SPD in front of the calorimeters, and the transverse energy deposition in the calorimeters. The number of vertices in the VELO that fall within a limited region around the nominal interaction point and VELO tracks crossing this region are counted separately. Some of these observables are directly obtained from the hardware trigger decision unit, others are the result of partial event reconstruction in the software trigger or in the off-line software. Observables used in this analysis are summarized in table 5.

The luminosity for a given data set can be determined by integrating the values of observables that are proportional to the instantaneous luminosity and by applying the corresponding absolute calibration constant. However, this procedure sets stringent requirements on the stability of the observable and on its linearity in the presence of multiple interactions. Alternatively, one may determine the relative luminosity from the fraction of "empty" or invisible events in bb crossings which we denote by P_0 . An invisible event is defined by applying an observable-specific threshold, below which it is considered that no pp interaction is seen in the corresponding bunch crossing. For

Table 5. Definition of luminosity observables used in the analysis. The fiducial volume used here	is a
cylinder of radius $< 4 \text{ mm}$ around the z axis and bound by $ z < 300 \text{ mm}$. It is used to cut either on the p	oint
of closest approach of a track relative to the z axis or on the position of a vertex.	

Name	Description	Origin
Track	Number of tracks reconstructed in the VELO in a	software reconstruction
	fiducial volume	
TrackNR	Number of tracks reconstructed in the VELO not	software reconstruction
	restricted to a fiducial volume	
Vertex	Number of vertices reconstructed in the VELO in a	software reconstruction
	fiducial volume	
VertexNR	Number of vertices reconstructed in the VELO not	software reconstruction
	restricted to a fiducial volume	
Muon	Number of muon tracks reconstructed in the muon	hardware trigger unit
	system	
PU	Number of hits counted in the PU	hardware trigger unit
SPD	Number of hits counted in the SPD	hardware trigger unit
ECalo	Energy deposition in the calorimeters	hardware trigger unit
Calo	Both SPD and ECalo over threshold	software reconstruction

a colliding bunch pair, the number of interactions per bunch crossing follows a Poisson distribution with mean value proportional to the luminosity, hence the luminosity is proportional to $-\ln P_0$. In the absence of backgrounds, the average number of visible pp interactions per crossing can be obtained from the fraction of empty bb crossings by $\mu_{vis} = -\ln P_0^{bb}$. This "zero-count" method is both robust and easy to implement [41]. The choice of a low visibility threshold ensures a better behaviour under gain or efficiency variations of the observable than the straightforward linear summing method. In addition, any non-linearity encountered with multiple events does not play a role when counting empty slots.

Assuming equal particle populations in bb, be, and eb bunches and no particles in ee slots, backgrounds are subtracted using

$$\mu_{\rm vis} = -\left(\ln P_0^{\rm bb} - \ln P_0^{\rm be} - \ln P_0^{\rm eb} + \ln P_0^{\rm ee}\right), \qquad (4.1)$$

where $P_0^i(i = bb, ee, be, eb)$ are the probabilities to find an empty event in a bunch-crossing slot for the four different bunch-crossing types. In eq. (4.1) it is implicitly assumed that all bunches of the same type have the same properties. The consequences of this approximation will be discussed in section 4.2. The P_0^{ee} contribution is added because it is also contained in the P_0^{be} and P_0^{eb} terms. The purpose of the background subtraction, eq. (4.1), is to correct the count-rate in the bb crossings for the detector response, which is due to beam-gas interactions and detector noise. In principle, the noise background is measured during ee crossings. In the presence of parasitic beam protons in ee bunch positions (ghost charge), it is not correct to evaluate the noise from P_0^{ee} . In addition, the detector signals are not fully confined within one 25 ns bunch-crossing slot for some of the observables. The empty (ee) bunch-crossing slots immediately following a bb, be or eb crossing slot contain detector signals from interactions occurring in the preceding slot ("spill-over"). However, because the filling schemes used for the data-taking described here did not contain adjacent filled



Figure 6. Mean value of (left) *ECalo* and (right) *Track* observables in ee crossings. Each histogram entry represents the average over a run² in 2012 and is weighted with the corresponding integrated luminosity. For measurements using the *ECalo* and *Calo* observables we discard the runs for which the *ECalo* pedestal mean is lower than 4.75 (dashed vertical line).

slots, the spill-over background is negligible in the bb, be and eb crossings. Since the detector noise for the selected observables is small (see section 4.2) the term $\ln P_0^{\text{ee}}$ in eq. (4.1) is neglected.

The results of the zero-count method based on the number of tracks and vertices reconstructed in the VELO are found to be the most stable. An empty event is defined to have < 2 tracks in the VELO. A VELO track is defined by at least three *R* clusters and three ϕ clusters on a straight line in the VELO detector. The number of tracks reconstructed in the VELO restricted to a fiducial region is chosen as the reference observable.

4.2 Systematic uncertainties

The zero-count method is valid if an event is considered empty when the value of the observable is exactly zero. However, if the observable is affected by noise such that its value is never zero, the threshold discriminating empty events has to be increased. This is the case for the *ECalo* and *Calo* observables, used as a cross-check, for which a positive threshold must be chosen. The introduced bias depends on the noise distribution, the one-interaction spectrum and the average number of interactions per crossing. While the latter was kept approximately constant during the 2012 data-taking period, the *Calo* noise distribution was changing due to ageing of the hadron calorimeter. In the second half of 2012, the HCAL gain was adjusted more frequently, thus keeping the noise distribution more stable.

The noise distribution is measured in ee crossings. Histograms of the mean value of the noise are shown for the *ECalo* and the *Track* observable in figure 6. For the *ECalo* observable, two peaks are observed in the pedestal distribution. This is attributed to a change of operating conditions, which is not easily corrected for. Therefore, for cross-checks using the *ECalo* and *Calo* observables we discard the runs for which the *ECalo* pedestal mean is lower than 4.75. The remaining larger fraction of runs spans the full year and is subsequently used for assessment of systematic uncertainties. The *Track* observable has typically less than 2.5 tracks per 100 ee crossings, which induces a negligible bias. The systematic uncertainty due to noise is negligible.

 $^{^{2}}$ A "run" is a consistent set of data, which usually spans about an hour of data taking, mainly used as an administrative unit.



Figure 7. Beam-gas background fraction $(-\ln P_0^{be} - \ln P_0^{eb})/\mu_{vis}$ for the *Track* observable during the 2012 running period.



Figure 8. Ratio of the measured μ_{vis} values without (*TrackNR*) and with (*Track*) a fiducial volume cut during the 2012 running period. The observed deviation from unity is used as an estimate of the systematic uncertainty due to potentially unaccounted background.

Equation (4.1) assumes that the proton populations in the be and eb crossings are the same as in the bb crossings. With a population spread of typically 10% and a beam-gas background fraction for the reference observable < 0.3% compared to the *pp* interactions (see figure 7) the effect of the spread is small and therefore neglected.

The measured μ_{vis} values can be contaminated by other backgrounds than beam-gas interactions, *e.g.* collisions between satellite and main bunches, and interactions with material in the VELO. We reduce such effects by applying a fiducial volume cut to the *Track* observable. To assess the magnitude of potentially unaccounted background, a comparison is made between the μ_{vis} values measured with (*Track*) and without (*TrackNR*) the fiducial volume cut, see figure 8. The observed discrepancy is used as an estimate of the systematic uncertainty due to beam-beam background.

The longitudinal position of the luminous region depends on the transverse beam separation in the crossing plane. In 2012 the instantaneous luminosity at LHCb was kept approximately constant by varying the separation of the two beams at the crossing point [42] (so called "luminosity levelling"). Due to imperfections in the procedure, the beams were displaced in a direction not exactly orthogonal to the crossing plane. As a result, there was a significant variation in the longitudinal position of the luminous region ξ_{lz} , as shown in figure 9. The reconstruction efficiency for tracks



Figure 9. Longitudinal position of the luminous region ξ_{lz} during the 2012 running period. The bottom plot shows a subset consisting of a few fills (distinguished with alternating open and solid markers).

and vertices is not uniform along z at the scale of the observed variations. Therefore, a correction needs to be applied to the observed μ_{vis} values that are measured using VELO observables. The *Calo* observable is not affected.

From simulation we determine I(0|z), the probability to obtain an empty event while having one interaction at z,

$$I(0|z) \equiv P(\text{empty event} | \text{ one interaction at } z), \qquad (4.2)$$

see figure 10 (left). Defining f(z) as the probability density of the longitudinal vertex distribution, the probability $\overline{I}(0|f)$ to have an empty event while one interaction occurred is

$$\bar{I}(0|f) = \int I(0|z) f(z) dz.$$
(4.3)

The absolute normalization of I(0|z) and $\overline{I}(0|f)$ depends on the underlying interaction generator. However, the normalization does not affect the luminosity measurement if used consistently. To avoid scaling μ_{vis} with factors largely different from unity, the correction is made with respect to a reference value $\overline{I}(0|ref)$ that corresponds to a Gaussian probability density g(z) of the longitudinal vertex distribution centred at zero and having an RMS of $\sigma_{lz} = 50$ mm,

$$\bar{I}(0|\mathrm{ref}) = \int I(0|z) g(z) dz.$$
(4.4)

The observed values, $\mu_{\rm vis}^{\rm raw}$, are proportional to $1 - \bar{I}(0|f)$. Thus, the corrected values are given by

$$\mu_{\rm vis} = \frac{1 - \bar{I}(0|\rm ref)}{1 - \bar{I}(0|f)} \,\mu_{\rm vis}^{\rm raw} \,. \tag{4.5}$$



Figure 10. (Left) probability to see an event given an interaction at *z* and (right) correction factor as function of the longitudinal position ξ_{lz} of the luminous region with size $\sigma_{lz} = 50$ mm. Only data for one magnet polarity are shown as it is almost identical to that for the other polarity.



Figure 11. RMS of μ_{vis} across bb bunch crossings relative to the mean value for the 2012 running period.

The z distribution of the vertices is well approximated with a Gaussian function. Examples of the correction factors for a Gaussian vertex distribution with $\sigma_{lz} = 50$ mm are shown in figure 10 (right).

An efficiency correction is implemented as a factor, according to eq. (4.5), evaluated as an average over about one-month running periods. To take into account a possible inaccuracy of the efficiency obtained by simulation, a comparison is made with an unaffected observable (*Calo*). The systematic uncertainty due to the residual dependence of the ratio $\mu_{\text{Track}}/\mu_{\text{Calo}}$ on the longitudinal position of the luminous region ξ_{lz} is estimated as follows. First, the data are divided in 5 mm bins in ξ_{lz} and the median of the ratio in each bin is calculated. The difference of the median with respect to that at $\xi_{lz} = 0$ is assumed to be due to imperfect correction of the *Track* observable. The relative difference is then averaged over the full data set taking into account the luminosity content. Finally, the resulting difference of 0.19% is taken as the systematic uncertainty due to the imperfect correction for the efficiency of the observable.

The numbers of protons, beam sizes and transverse offsets at the interaction point vary across bunches. Thus, the μ_{vis} value varies across bb crossings. An estimate of the spread of μ_{vis} values is the RMS divided by the mean across bunch crossings, as shown in figure 11. Due to the nonlinearity of the logarithmic function, ideally one first needs to compute μ_{vis} values for different bunch crossings and then take the average. However, for short time intervals the number of samples is insufficient to make an unbiased measurement per bunch crossing using the zero-count method, while μ_{vis} may not be constant when the intervals are too long due to *e.g.* loss of bunch population and emittance growth.

During physics data-taking, the bandwidth reserved for luminosity triggers is limited. Therefore, a statistically significant measurement of the luminosity cannot be obtained for each bunch crossing individually by integrating over periods shorter than about 30 minutes. The bias and systematic uncertainty introduced by this limitation is evaluated with a simulation. To reflect the luminosity integration for physics data, the number of visible interactions is counted in short time intervals ignoring the spread of μ_{vis} values across bunch crossings. A set of 30 consecutive short intervals is used to accumulate a sufficient number of events per BCID. Then, a correction for the spread is calculated and applied as described below.

For the following discussion backgrounds are not considered since their effect on the correction is negligible. Let n_{ti} and k_{ti} denote the number of random triggers and the number of empty events, respectively, in bb crossing slot *i* and short time interval *t*. Where the index *t* is omitted, an implicit sum is assumed over the set *T* of consecutive short intervals. Similarly, in case the index *i* is omitted, a summation over all bb bunch crossings is assumed. A correction factor is calculated for every set *T* and is applied to each short period $t \in T$

$$\kappa_T = \frac{\langle -\ln\frac{k_i}{n_i} \rangle}{-\ln\frac{k_n}{n}},\tag{4.6}$$

where the average in the numerator is taken over *i*. The corrected estimate of the number of visible interactions is

$$N_{\text{vis},T} = \kappa_T f_{\text{trig}} \sum_{t \in T} \left(-n_t \ln \frac{k_t}{n_t} \right), \qquad (4.7)$$

where f_{trig} is the probability that a bb crossing is randomly triggered.

A simulation study is performed to compare the bias of the estimated number of interactions before and after the correction procedure. The rate of triggers and the number of bunch crossings is chosen to reflect the typical running conditions. The μ values across bunch crossings are sampled from a normal distribution. A luminosity half-life of two hours is assumed. The bias is calculated as function of mean μ value and relative RMS, and is shown in figure 12.

To estimate the residual bias of the correction technique on the data, we perform a simulation for each long period T. First, the μ_{vis} value is estimated for each short period t and each bunch crossing i with

$$\mu_{ti} = -\frac{\ln\frac{k_t}{n_t} \ln\frac{k_i}{n_i}}{\ln\frac{k}{n}},\tag{4.8}$$

which has the desirable property that it coincides with the projection estimates (μ_t and μ_i) when the true μ value does not change over time or across colliding bunches. Then, for each (t,i)



Figure 12. Bias of the estimated number of interactions as function of (left) mean μ value and (right) relative RMS. The bias is fitted with a straight line and an even quadratic polynomial as function of $\langle \mu \rangle$ and relative RMS, respectively. A quadratically increasing bias as function of relative RMS is present for large $\langle \mu \rangle$ values before the correction (dashed blue line). For 2012 data taking conditions, the typical $\langle \mu \rangle$ for the *Track* observable is 1.7 and the typical relative RMS is 8%.



Figure 13. (Left) correction applied to the estimated values of μ_{Track} for variations of its value across bunch crossings and (right) residual relative bias of the estimated number of visible interactions after the correction. Each histogram entry represents a run in 2012 and is weighted with the corresponding integrated luminosity.

pair, k_{ti} is sampled from a binomial distribution with success probability $e^{-\mu_i}$ and number of trials equal to n_{ti} . As for the actual data, eqs. (4.6) and (4.7) are used to estimate the number of visible interactions. Finally, the bias is obtained from the difference of the estimated and the true number of visible interactions, averaged over 25 independent repetitions of the simulation. Histograms of the average values of κ_T and the residual relative bias for each run are shown in figure 13. The relative integrated bias over the full data set is assigned as a systematic uncertainty (0.14%).

In addition, a cross-check is made using the *Muon* observable, which is less sensitive to the spread owing to its low μ_{vis} values ranging from 0.07 to 0.15.³ The ratio μ_{Track}/μ_{Muon} as a function of the relative RMS of μ_{vis} across bunch crossings is fitted with an even quadratic polynomial. The 0.5% of runs with extreme values of the ratio are excluded from the fit. The maximum relative difference between the predicted value at the mean spread and at zero spread gives an estimate of the residual bias. Since it has an opposite sign with respect to the residual bias obtained from the

 $^{^{3}}$ Due to a change of threshold mid-2012, the *Muon* observable visible cross-section changed significantly. Therefore, the periods before and after the change are treated independently.



Figure 14. Ratio of the relative luminosities using the *Track* and the *Calo* observables during the 2012 running period. Only data for runs that are longer than 30 min are plotted. The variation of the ratio after subtraction of the variation due to statistical fluctuations is shown with a shaded area spanning $\pm 1\sigma$ around the mean.

Table 6. Top: systematic uncertainties of the relative luminosity measurement (in %). Bottom: integrated effect of the applied corrections (in %).

		pp		pPb	Pbp
Source	8 TeV	7 TeV	2.76 TeV	5 TeV	5 TeV
Beam-beam background	0.13	0.24	0.13	0.95	0.73
Efficiency of the observable	0.19	0.07	0.12	0.09	0.11
Bunch spread	0.14	0.09	0.10	0.03	0.03
Bunch spread (cross-check)	0.09	0.44			
Stability	0.12	0.13	0.14	0.39	0.35
Total	0.31	0.53	0.25	1.03	0.82
Correction					
Efficiency of the observable	-0.54	-0.11	-0.12	-0.09	-0.11
Bunch spread	+0.72	+0.99	+0.10	+0.03	+0.03

simulation, the result is taken as an additional systematic uncertainty (0.09%).

The stability of the reference observable is demonstrated in figure 14, which shows the ratio of the relative luminosities determined with the zero-count method using the *Track* and the *Calo* observables. These two observables use different sub-detectors and have different systematic uncertainties. The variation of the ratio unexplained by statistical fluctuations is assigned as a systematic uncertainty to the relative luminosity measured using the *Track* observable. A similar cross-check with the ratio of the relative luminosities using the *Track* and the *Vertex* observables shows negligible discrepancy.

The systematic uncertainties of the relative luminosity measurement are summarized in table 6. By summing the effect of the different sources in quadrature, we conclude that the relative luminosity measurement introduces a systematic uncertainty of 0.31% for the pp run at 8 TeV. The quoted uncertainty applies when the full dataset is used; for specific choices of partial datasets a different value may apply. In the case of the 2013 running conditions (proton-lead and pp at 2.76 TeV), the corrections due to the movement of the luminous region and the bunch spread are small. Since the available data sample size is insufficient to reliably perform the corresponding cross-checks, the full amount of each correction is assigned as an uncertainty. The beam-beam background uncertainty is estimated to be up to 1% for the proton-lead data taking, owing to the very low μ values (0.01–0.02) of these runs. A higher uncertainty of about 0.5% due to the bunch spread is estimated for the 2011 data taking. This is explained by worse conditions in the beginning of the year, when the spread of μ across bunches reached 30%, which leads to a correction of up to 7% for some fills. The 2011 data taking at 7 TeV was affected by parasitic collisions due to a vanishing net crossing angle for one of the magnet polarity settings. This background ranges between 0.2% and 0.7% and a correction is applied averaging over time intervals of a few weeks each, during which data were taken under similar conditions. The average correction amounts to about 0.4% and since only about half of the 2011 running period is affected, an uncertainty due to parasitic collisions of 0.2% is assigned on the full period. In addition, the estimated uncertainty due to beam-beam background from 2012 is added in quadrature to obtain 0.24% uncertainty for 2011. The stability of the effective process is estimated using only data that is not affected by parasitic collisions.

5 Formalism for the luminosity of colliding beams

In a cyclical collider, such as the LHC, the average instantaneous luminosity of one pair of colliding bunches can be expressed as [6]

$$L = N_1 N_2 v_{\text{rev}} \sqrt{|\mathbf{v}_1 - \mathbf{v}_2|^2 - \frac{|\mathbf{v}_1 \times \mathbf{v}_2|^2}{c^2}} \int \rho_1(x, y, z, t) \rho_2(x, y, z, t) \, dx \, dy \, dz \, dt \,, \tag{5.1}$$

where we have introduced the velocities \mathbf{v}_1 and \mathbf{v}_2 of the particles (in the approximation of zero emittance the velocities are the same within one bunch). The particle densities $\rho_j(\mathbf{r},t)$ (j = 1,2) at position $\mathbf{r} = (x, y, z)$ and time *t* are normalized such that their individual integrals over all space are unity at all times. For highly relativistic beams colliding with a small half crossing-angle ϕ , the Møller factor $\sqrt{(\mathbf{v}_1 - \mathbf{v}_2)^2 - (\mathbf{v}_1 \times \mathbf{v}_2)^2/c^2}$ reduces to $2c\cos^2\phi \simeq 2c$ and one recovers eqs. (1.2) and (1.3). The LHCb system of coordinates, which is used here, is chosen as a right-handed cartesian coordinate system with its origin at the nominal interaction point IP8. The *z* axis points towards the LHCb dipole magnet along the nominal average beam-line, the *x* axis lies in the horizontal plane, with x > 0 pointing approximately toward the centre of the LHC ring, and the *y* axis completes the right-handed system. This system almost coincides with the LHC coordinate system. Small angles due to the known LHC plane inclination and other magnetic lattice imperfections have negligible influence on the measurement of the overlap integral as only the crossing angles are relevant, not the individual beam directions.

Up to a normalization factor, $\rho_{bb}(x, y, z, t) = \rho_1(x, y, z, t) \rho_2(x, y, z, t)$ is the distribution of interactions from the luminous region in the laboratory frame. If both ρ_1 and ρ_2 factorize as a product of a longitudinal and a transverse density (relative to the direction of motion of the bunch), the spatial distribution integrated over time⁴ can be expressed as

$$\rho_{bb}(x, y, z) = n(z) \rho_1(x, y, z) \rho_2(x, y, z)$$
(5.2)

⁴When the time dependence is dropped, an integration over time is implied: $\rho(x, y, z) = \int \rho(x, y, z, t) dt$.

where n(z) is a shape factor which depends on z only. This relation between the distributions of beam-beam and beam-gas interactions is used in the BGI analysis.

Determining the luminosity or the reference cross-section requires measuring the bunch population products $N_1 N_2$, as discussed in section 3, and evaluating the overlap integral Ω . We briefly describe the principles of the two methods that are used in this paper to determine the latter.

5.1 Beam overlap measurement methods

The first method was introduced by van der Meer to measure the luminosity of the coasting beams at the Intersecting Storage Rings (ISR) [12]. The method was further extended to measure the luminosity of a collider with bunched beams [13] and is the main method used to determine the luminosity at the other LHC experiments. The key principle of the VDM scan method is to express the overlap integral in terms of rates that are experimental observables as opposed to measuring the bunch density functions. Experimentally, the method consists in moving the beams across each other in two orthogonal directions. The overlap integral can be inferred from the rates measured at different beam separations, provided the beam displacements are calibrated as absolute distances.

A reaction rate *R* per bunch crossing is measured that is proportional to the luminosity and depends on the two orthogonal transverse separations of the two beams Δx and Δy . Measuring this rate relative to the revolution frequency v_{rev} (approximately 11245 Hz at the LHC) defines the parameter μ , which is the average number of reactions per bunch crossing. In the case where the spatial distributions of the beams can be factorized in the two coordinates *x* and *y*, it is sufficient to measure μ (and thus *R*) as a function of Δx (at a fixed Δy_0) and as a function of Δy (at a fixed Δx_0). One can show that the interaction cross-section is then given by

$$\sigma = \frac{\int \mu(\Delta x, \Delta y_0) \, d\Delta x \cdot \int \mu(\Delta x_0, \Delta y) \, d\Delta y}{N_1 N_2 \, \mu(\Delta x_0, \Delta y_0)} \,. \tag{5.3}$$

The pair of separation values $(\Delta x_0, \Delta y_0)$ is called the working point and is typically chosen to be as close as possible to the point where the luminosity is at its maximum. However, eq. (5.3) is valid for any values of Δx_0 and Δy_0 . It can be shown that it is also valid in the presence of non-zero crossing angles [14].

The VDM method has the advantage of using a measured rate as its only observable, which is experimentally simple. The experimental difficulties of the VDM method arise mostly from the fact that the beams must be moved to perform the measurement. The exact displacements Δx and Δy in eq. (5.3) steered by the LHC magnets are calibrated at each interaction point in a so-called length scale calibration (LSC). While the resulting corrections are typically of the order of 1%, some non-reproducibilities have been observed between two consecutive scans without being able to identify the cause. Another difficulty originates from beam-beam effects. When the beams are displaced, a change in β^* (dynamic beta effect) and a beam deflection may be produced, which both influence the observed rate. The resulting corrections to the visible cross-section depend on the LHC optics, the beam parameters and filling scheme, and must be evaluated at each interaction point (see section 7.6).

In addition, when performed with one vertical and one horizontal scan, the VDM method is valid only under the assumption that the distributions along the transverse variables x and y are independent, *i.e.* that the x (y) shape measured at a working point Δy_0 (Δx_0) does not depend on

the working point position. As will be shown in the analysis described here, this assumption is not valid at the required precision.

An alternative to the VDM scan method for measuring the luminosity is provided by the BGI method [16], which was first applied at the LHCb experiment in 2009 [21] and 2010 [11]. The principle of this method is to evaluate the overlap integral by measuring all required observables in eq. (1.3) using the spatial distribution of beam-gas and beam-beam interaction vertices. The details of the measurement are discussed in section 6. Measuring the shapes of stationary beams avoids changes due to beam-beam effects and other, non reproducible, effects due to beam steering. Furthermore, at the LHC the BGI measurements at a given IP (here at LHCb in IP8) can be made in parallalel to the VDM scans of other LHC experiments and can therefore be made more frequently.

On the other hand, while the β^* and crossing angles used at the LHC do not impact the VDM method to first order, the BGI measurement relies on the vertex measurement to determine the bunch shape. Therefore, an increased β^* is preferable to avoid limitations introduced by the detector resolution. At LHCb, in 2012, *pp* physics data were acquired at $\beta^* = 3$ m, while the most precise BGI luminosity calibrations fills were carried out with $\beta^* = 10$ m. The knowledge of the crossing angle is also important since the luminosity reduction due to the crossing angle has been as large as 20%. A non-vanishing crossing angle is necessary to avoid interactions between the main bunch and out-of-time charges captured in the next RF bucket, which occur near $z = \pm 37.5$ cm. Such displaced collisions, if present, must be disentangled from beam-gas interactions. They can be completely avoided by introducing a sufficiently large crossing angle. The VDM measurement can exclude interactions occurring away from the interaction point and is therefore less affected by these satellite collisions.

The VDM and BGI methods are complementary, in the sense that their systematic uncertainties on the overlap integral are highly uncorrelated, and a luminosity calibration performed with both methods in the same fill permits their systematic uncertainties to be constrained further. At present this can only be done at the LHCb experiment.

The analyses of the VDM and BGI luminosity calibration measurements presented here indicate that the observed luminosity profiles and vertex distributions are not consistent with Gaussian bunch distributions. It is found that a sum of two Gaussian functions ("double Gaussian" shape model) is sufficient to describe the x and y shapes of each bunch as well as the resulting luminous region. However, the joint two-dimensional transverse distribution of the bunches is found to be non-factorizable in the transverse coordinates. Therefore, as explained in section 5.3, the transverse shape of the bunches is modelled with a sum of four two-dimensional Gaussian functions, which is in general non-factorizable.

In order to explain the full analysis of the present work, which involves a detailed fit model with a sum of Gaussian terms, it is useful to consider first the formalism for the ideal case of pure Gaussian beams and then describe the two-dimensional (non-factorizable) Gaussian model used in this work.

5.2 Luminosity in the case of purely Gaussian beams

The overlap integral in eq. (5.1) can be calculated analytically when the single beam distributions ρ_j (j = 1,2) are the product of three Gaussian functions, each one depending on a single spatial coordinate $m = \hat{x}_j$, \hat{y}_j , or \hat{z}_j . The beam reference frames \hat{x}_j , \hat{y}_j , \hat{z}_j are right-handed systems and

the longitudinal axis \hat{z}_j is assumed to be parallel to the velocity vector of the bunch \mathbf{v}_j . It is also assumed that the \hat{y}_j axes of the two colliding bunches are parallel to the y axis of the laboratory frame. The beam crossing plane, defined by the velocity vectors \mathbf{v}_1 and \mathbf{v}_2 , is here assumed to coincide with the xz plane. This condition was not respected only for the April 2012 fills. The relevant modifications of the formulae below are discussed in section 7.2. We assume the bunches are centred at $\mathbf{r}_j = (\xi_{xj}, \xi_{yj}, \xi_{zj})$ at time t = 0, with a particle density function described by a normalized Gaussian function

$$\rho_{mj}(m) = \frac{1}{\sqrt{2\pi}\sigma_{mj}} e^{-\frac{1}{2}\left(\frac{m-\xi_{mj}}{\sigma_{mj}}\right)^2} \quad \text{for beam } j = 1,2 \text{ and coordinate } m = \hat{x}, \hat{y}, \hat{z}, \tag{5.4}$$

where σ_{mj} denotes the RMS of the corresponding Gaussian function.

Assuming that $\rho_j(\mathbf{r},t) = \rho_j(\mathbf{r} - \mathbf{v}_j t, 0)$, one can show that the overlap integral becomes

$$\Omega = \frac{e^{-\frac{\Delta x^2}{2\Sigma_x^2} - \frac{\Delta y^2}{2\Sigma_y^2}}}{2\pi\Sigma_x\Sigma_y},$$
(5.5)

where the following quantities have been introduced

$$\Sigma_x^2 = 2\sigma_z^2 \sin^2 \phi + 2\sigma_x^2 \cos^2 \phi \quad \text{with} \quad 2\sigma_x^2 = \sigma_{\hat{x}1}^2 + \sigma_{\hat{x}2}^2$$

$$\Sigma_y^2 = 2\sigma_y^2 \quad 2\sigma_y^2 = \sigma_{\hat{y}1}^2 + \sigma_{\hat{y}2}^2 \quad (5.6)$$

$$2\sigma_z^2 = \sigma_{\hat{z}1}^2 + \sigma_{\hat{z}2}^2$$

and $\Delta m = \xi_{m1} - \xi_{m2}$ (with m = x, y) are the transverse beam separations evaluated at the moment t = 0 when the colliding bunches are at the same *z* position. In the LHCb experiment, this *z* position (called z_{rf}) is defined by the LHC RF timing and needs not coincide with the location z = 0 of the LHCb laboratory frame nor with the geometrical crossing point of the two beam trajectories.

The longitudinal position ξ_{lz} of the luminous region is related to the beam separation Δx and longitudinal bunch crossing point z_{rf} with

$$\xi_{lz} - z_{\rm rf} = \frac{\sin\phi\cos\phi\left(\sigma_x^2 - \sigma_z^2\right)}{\Sigma_x^2}\Delta x.$$
(5.7)

The index *l* indicates here a property of the luminous region, as opposed to a single beam property.

One can also show that the longitudinal size σ_{lz} of the luminous region is related to the convolved bunch length σ_z by

$$\frac{1}{\sigma_{l_z}^2} = \frac{2\sin^2\phi}{\sigma_x^2} + \frac{2\cos^2\phi}{\sigma_z^2}.$$
(5.8)

Therefore, if one has a measurement of the transverse bunch size of the individual beams, of the crossing angle and of the longitudinal size of the luminous region, one can evaluate the longitudinal bunch convolution.

5.3 Double Gaussian shape model

A factorizable transverse beam distribution with double Gaussian projections has the density

$$\rho(\hat{x}, \hat{y}) = \rho(\hat{x}) \rho(\hat{y}) = \prod_{m=\hat{x}, \hat{y}} [w_m g(m; \xi_m, \sigma_{mn}) + (1 - w_m) g(m; \xi_m, \sigma_{mw})]$$

= $\sum_{i_x i_y} w_{i_x i_y} g(\hat{x}; \xi_{\hat{x}}, \sigma_{\hat{x}i_x}) g(\hat{y}; \xi_{\hat{y}}, \sigma_{\hat{y}i_y}),$ (5.9)

where $g(m; \mu, \sigma)$ indicates a normalized Gaussian function of the variable *m* with parameters μ and σ . By convention, the narrow (n) and wide (w) components in each projection have widths σ_{mn} and σ_{mw} , and weights w_m and $1 - w_m$. The weights $w_{i_x i_y}$ in the sum representation are defined as

$$w_{i_x i_y} = \begin{bmatrix} w_{nn} & w_{nw} \\ w_{wn} & w_{ww} \end{bmatrix} = \begin{bmatrix} w_x w_y & w_x (1 - w_y) \\ (1 - w_x) w_y & (1 - w_x) (1 - w_y) \end{bmatrix}.$$
 (5.10)

The wide and narrow components are assumed to have the same mean, as supported by the data. Moreover, it is assumed that the 3-dimensional bunch distribution factorizes in a transverse ($\rho(\hat{x}, \hat{y})$) and a longitudinal component, where the latter is modelled with a Gaussian function.

Non-factorizability can be introduced into the model in eq. (5.9) by modifying the weights $w_{i_x i_y}$ from eq. (5.10). For instance, in an extreme case, one can have $w_{nw} = w_{wn} = 0$ and $w_{nn} + w_{ww} = 1$, which corresponds to a sum of two 2-dimensional Gaussian functions. To allow for a gradual transition between this extreme case and the case of factorizable beams, it is useful to define the weights as a linear combination

$$\begin{bmatrix} w_{nn} & w_{nw} \\ w_{wn} & w_{ww} \end{bmatrix} = f \begin{bmatrix} w_x w_y & w_x (1 - w_y) \\ (1 - w_x) w_y & (1 - w_x) (1 - w_y) \end{bmatrix} + (1 - f) \begin{bmatrix} \frac{w_x + w_y}{2} & 0 \\ 0 & 1 - \frac{w_x + w_y}{2} \end{bmatrix},$$
(5.11)

where the coefficient f parametrizes the factorizability. In the fully non-factorizable case (f = 0) there is no distinction between the x and y weights, thus the parameters w_x and w_y are (arbitrarily) combined in a single weight.

As a result of the single beam model from eq. (5.9), the shape of the luminous region and the overlap integral are described by a weighted sum of 16 components. Explicitly, the beam overlap integral is given by

$$\Omega = \sum_{I} w_I \Omega_I = \sum_{I} w_{i_x i_y, 1} w_{j_x j_y, 2} \Omega_I , \qquad (5.12)$$

where *I* denotes the set of indices i_x, i_y, j_x, j_y , while $w_{i_x i_y, 1}$ and $w_{j_x j_y, 2}$ are the weights from eq. (5.11) for beam 1 and beam 2. Each partial overlap integral Ω_I is evaluated with eq. (5.5).

6 Beam-gas imaging method

In this section, the BGI methodology and calibration results are presented in detail. A description of the data taking conditions (trigger settings, vacuum conditions) and of the event selection are given. Studies of the vertex position resolution and the unfolding method are presented. The resolution is determined from data, separately for beam-beam and beam-gas interactions. An analysis of the resolution-corrected vertex position distribution is then performed, which uses both beam-gas

vertices and beam-beam vertices to perform a global fit of the beam parameters (angles, luminous region length, longitudinal crossing position, transverse beam shapes). A double Gaussian model is used which also allows for a non-factorizability of the x and y distributions. Simulation is used to verify the soundness of the fit procedure. Several checks are made, based on data, to quantify systematic uncertainties. The list of dedicated luminosity calibration fills discussed in this paper for 2011, 2012 and 2013 can be found in table 1. We focus on the 8 TeV pp data set taken in 2012, because it gives the most precise results. For the other data sets the analysis is similar and only the differences with the former are discussed in section 6.2.

6.1 Data-taking conditions and event selection

For dedicated luminosity calibrations the LHC is filled with a low number of bunches, of the order of 50 per beam or less, and a large gap between bunches ($\sim 1 \mu s$) is maintained. Under these conditions the vacuum pressure at the interaction point is $\sim 10^{-9}$ mbar, producing a beam-gas trigger rate of about 0.5 Hz per 10¹¹ protons.⁵ Performing a BGI measurement with such low rates necessitates integration of a measurement over a period of up to 8 hours. Significant limitations in the precision are caused by the low event rate, beam drift and by emittance growth over the integration time. In order to mitigate this, in 2011 the VELO vacuum pumps located close to the interaction point were switched off, thus increasing the beam-gas rate by about a factor of four. To increase the rate further and to take full advantage of the BGI capabilities, the use of a gas injection system was proposed [16], developed and commissioned in the LHCb experiment [37].

A first gas injection test with circulating beams was performed in November 2011. When activating the system, neon gas is injected in the VELO, thus raising the pressure from about 10^{-9} mbar to slightly above 10^{-7} mbar. Once the injection is stopped, the nominal pressure of $\sim 10^{-9}$ mbar is recovered within 20 minutes. The pumps are switched off during the gas injection. The effect of gas injection on the pressure and beam-gas interaction rate is shown in figure 15. For *pp* collisions at $\sqrt{s} = 8$ TeV, the recorded beam-gas event rates per 10^{11} protons were 98 Hz for beam 1 and 82 Hz for beam 2. The corresponding rates of the hardware trigger were about 2.1 kHz and 1.3 kHz for beam 1 and beam 2, respectively.

All luminosity calibrations acquired in 2012 and 2013 took advantage of a simplified activity trigger in all bunch crossing types. For these fills the hardware trigger used information from the SPD and PU sensors and from the calorimeters. The software trigger dedicated to the BGI measurement accepted events with a multiplicity larger than ten tracks. The vertex position *z* was required to lie within a range of -2000 < z < 400 mm for beam 1-gas events and 0 < z < 2000 mm beam 2-gas events. Of the triggered events in bb crossings with a vertex in the range -300 < z < 300 mm only a fixed fraction was accepted ("prescaled") to keep their total rate below 15 kHz, close to the maximum rate that can be recorded. Interactions with a transverse vertex position more than 4 mm from the nominal beam line were not accepted in order to reject interactions with the material of the RF foil of the VELO [29].

All events acquired are reconstructed offline to determine their vertex position. A vertex has a number of tracks N_{Tr} associated with it, each track having either a forward or backward direction with respect to beam 1. Their multiplicities are defined as N_{fwd} and N_{bwd} , respectively. Forward

⁵During fills dedicated to luminosity the bunch population ranges between $0.7 \cdot 10^{11}$ and $1.1 \cdot 10^{11}$.



Figure 15. Effect of the gas injection (fill 2520), with (left) the pressure increase in the VELO and (right) the trigger rate increase. The VELO ion pumps are off. The pressure is measured with Penning gauges in the vacuum vessel within 50 cm of the interaction point. The indicated value when neon is injected is to be multiplied by approximately a factor four to account for the lesser gauge sensitivity to neon gas. The absolute pressure increase cannot be accurately determined as the gas composition in unknown before the start of injection. The beam-gas rate for triggered events increases from about 2 to 72 Hz/bunch for beam 1 and 1.3 to 51 Hz/bunch for beam 2.

tracks correspond to particles moving towards the LHCb spectrometer and benefit from additional information such as energy and transverse momentum, which can improve the vertex resolution. The VELO acceptance for forward tracks vanishes when they originate from $z \gtrsim 600 \,\text{mm}$. Backward tracks are only recorded by the VELO and can be detected when they are produced from a vertex with longitudinal position $z \gtrsim -95 \,\text{mm}$. The longitudinal vertex distribution for all bunchcrossing types before applying further selection criteria is shown in figure 16. The acceptance limits for beam-gas events from beam 1 and beam 2 are visible as a sharp drop in rate. It can be seen that the acceptance drops in two stages for beam 1 at about 200 mm and 400 mm; this effect is related to the positions of VELO sensors in the forward region. The distribution of beam-beam interactions in the luminous region is reduced due to the prescale factor, which also affects beam-gas interactions in bb crossings located in $|z| < 300 \,\text{mm}$. The xz and yz vertex distributions for non-colliding bunches are shown in figure 17 for the first 1000 vertices per beam for two fills.

The BGI method relies on differentiation of beam-gas and beam-beam vertices. The reconstructed vertex is required to contain at least 10 tracks, $N_{\text{Tr}} \ge 10$. To exclude interactions with material in the VELO RF foil, the vertex position must be within a radial distance of 2 mm from the beam line. The longitudinal position of a beam-gas vertex must satisfy $-1000 \text{ mm} \le z \le 500 \text{ mm}$ for beam 1-gas vertices and $0 \text{ mm} \le z \le 1000 \text{ mm}$ for beam 2-gas vertices. In bb crossings, these ranges are reduced to $-1000 \text{ mm} \le z \le -250 \text{ mm}$ and $250 \text{ mm} \le z \le 1000 \text{ mm}$, respectively. For beam 1 (beam 2)-gas vertices all tracks are required to be in the forward (backward) direction. Beam-beam interactions are selected requiring at least two tracks in both directions in addition to a minimum requirement of 25 on the vertex track multiplicity.

The measurement of the beam angles combines beam-gas interaction vertices from colliding and non-colliding bunches. For the measurements of the overlap integral, beam-gas and beam-



Figure 16. Longitudinal distribution of vertices for the various bunch crossing types acquired in 40 minutes during fill 2852. Crossing types ee (green), be (blue) and eb (red) contain only beam-gas events while the bb crossing type (black) contains beam-beam vertices in the central region and beam-gas vertices over the whole range. The effect of the prescale factor on the beam-beam interaction rate is visible. The exclusion region of ± 300 mm for beam-gas vertices in bb crossings is visible as the step given by the prescale factor applied to these vertices inside this range.



Figure 17. Position of beam-gas vertices projected in the xz and yz planes. The first 1000 vertices per beam are shown. The crossing angles are visible in both planes (left) for fill 2520 and in the xz plane only (right) for fill 2852. The angles are measured by fitting a straight line through all vertices and indicated as solid lines. The beams were offset in the yz plane during this period of fill 2852; beam 1 is slightly above beam 2 in the y plane (right plot, bottom).

beam events from colliding bunch pairs are used. Beam-gas interactions in non-colliding and empty bunch crossings determine the ghost charge fractions and are used in the beam-gas vertex resolution determination.



Figure 18. Distributions for beam-beam vertices selected for the resolution measurement with (left) the longitudinal position distribution and (right) the track multiplicity distribution. The requirement to have at least two forward and two backward tracks in each vertex causes the sharp efficiency drop close to z = -100 mm. Fill 2520 is selected as example for this figure.

6.2 Vertex position resolution

The knowledge of the vertex position resolution is a central ingredient for the measurement of the absolute luminosity as the observed vertex distribution is a convolution of the physical beam or luminous region with the detector resolution. To reduce the impact of the resolution on the measurement of the overlap integral, the beam optics for dedicated luminosity calibration fills had a β^* value of 10 m (compared to 3 m used for physics production runs) and an increased transverse emittance. This resulted in a beam width about twice as large as the transverse vertex resolution.

The vertex resolution used for the BGI analysis is understood as the standard deviation of the distribution of the residual distance in one coordinate between the true vertex position and its measured position. The resolution depends on the number of tracks associated with a vertex, the longitudinal position and whether the vertex originates from a beam-gas collision with only forward or only backward tracks, or from a beam-beam collision with both forward and backward tracks. Although the value of the resolution in *z* is about ten times worse than that in the transverse directions, its effect can be neglected owing to the much larger luminous region length (~ 60 mm). Therefore, only the resolution in the transverse *x* and *y* directions is considered here. The resolution measurement method (described below) has been verified with simulated events. Our studies show that a better resolution is predicted by simulation compared to the measurement from data. The difference could be explained by the imperfect alignment of the VELO sensors.

6.2.1 Resolution for beam-beam interaction vertices

The longitudinal distribution of selected beam-beam interaction vertices and the corresponding distribution of the number of tracks per vertex are shown in figure 18.

Without external knowledge of the true position of primary vertices, the residual distance to the true position and therefore the resolution cannot be measured directly. Instead, one can measure the residual distance between two reconstructed vertices originating from the same collision by



Figure 19. Measurement of the position difference between two sub-vertices from beam-beam interactions (fill 2520). The left-hand panel shows the distribution of distances in *x* between the two measurements resulting from primary vertex splitting, selecting the case with exactly 48 tracks in both sub-vertices. The data are fitted with a Gaussian function. The right-hand panel shows as a colour code the standard deviation $\sigma_{\Delta v_x}$ of measured distances in *x* between the two vertices as a function of the vertex track multiplicities. The measured RMS of 0.0129 mm for two vertices with 48 tracks as shown on the left plot corresponds to the blue square at a multiplicity of 48 for both vertex 1 and vertex 2.

dividing the tracks forming one vertex into two randomly chosen samples ("split vertex method"). Defining the absolute vertex positions v_1 and v_2 (v = x or y) resulting from the primary vertex splitting and $\Delta v = v_1 - v_2$ the distance between the two measurements, the Gaussian width $\sigma_{\Delta v}$ of all Δv distances is a convolution of both vertex resolutions and depends on the vertex track multiplicity $N_{\text{Tr},1}$ and $N_{\text{Tr},2}$ of each vertex

$$\sigma_{\Delta\nu}^2 = \sigma_{\text{res},\nu 1}^2(N_{\text{Tr},1}) + \sigma_{\text{res},\nu 2}^2(N_{\text{Tr},2}), \qquad (6.1)$$

where $\sigma_{\text{res},v}(N_{\text{Tr}})$ is the resolution for the vertex track multiplicity N_{Tr} . Indices 1, 2 denote here the two vertices resulting from the splitting. It is observed that the position of the original primary vertex is not identical to the average position of the two sub-vertices. To minimize any systematic bias due to the resolution measurement procedure, the analysis of the beam parameters is performed using the average position of the two sub-vertices rather than the position of the original primary vertex.

The dependence on N_{Tr} is determined using the full *z* range of the luminous region. In a second step, the variation of the resolution as function of *z* is addressed by introducing *z*-dependent factors. Measured distances Δv are sorted according to the possible combinations $i = (N_{\text{Tr},1}, N_{\text{Tr},2})$ of the two vertex track multiplicities. For each combination *i* the distribution of all Δv_i measurements are fitted with a Gaussian function to determine the width of the distribution $\sigma_{\Delta v_i}$. An example of such a distribution is shown in figure 19 (left) together with the result of the fit. In this particular case, the number of tracks in both sets being equal, the measured distribution width $\sigma_{\Delta v_i}$ is directly related to the vertex resolution $\sigma_{\text{res},v}(N_{\text{Tr}})$ using eq. (6.1). All measured distribution widths $\sigma_{\Delta x_i}$ as function of $(N_{\text{Tr},1}, N_{\text{Tr},2})$ combinations are shown in figure 19 (right).



Figure 20. Results of the beam-beam vertex position resolution measurement in fill 2520 with (left) the resolution as a function of the track multiplicity and (right) the distribution of parametrization residuals. The markers in the left plot show the direct determination of the individual resolution for a given vertex track multiplicity for the *x* coordinate. Results for the *y* coordinate are similar. The curve shows the results of the resolution parametrization fitted using eq. (6.2). In this fill the fit gives the values $A_x = 0.110$ mm, $B_x = 0.669$, $C_x = 0.0011$ mm, $A_y = 0.101$ mm, $B_y = 0.640$, $C_y = 0.0006$ mm for the parameters defined in eq. (6.2). The entries in the right plot are weighted by the number of vertices observed with a given vertex track multiplicity.

The resolution for a given vertex track multiplicity $\sigma_{res,\nu}(N_{Tr})$ is obtained by fitting all distribution widths $\sigma_{\Delta\nu_i}$ with a least squares minimization constrained by eq. (6.1). Results for the resolution as a function of vertex track multiplicity are shown in figure 20. As can be seen in figure 18 (right), the number of vertices with more than 120 tracks vanishes, limiting the resolution measurement up to about 60 tracks per vertex. The resolution for vertices with a larger number of tracks is obtained by extrapolation of a parametrization,

$$\sigma_{\mathrm{res},\nu}(N_{\mathrm{Tr}}) = \frac{A}{N_{\mathrm{Tr}}^B} + C.$$
(6.2)

The factor *A*, the power *B* and constant term *C* are measured by fitting all $\sigma_{\Delta v_i}$ measurements. Typical values are given for fill 2520 in the caption of figure 20. Results for the resolution parametrization functions are shown in figure 20 and are in good agreement with results from the direct determination of individual resolutions for each vertex track multiplicity. No bias is observed and the statistical fluctuations are less than 0.1 µm. Beam-beam vertex resolution parametrization results are evaluated for each fill independently.

The variation of the vertex resolution for beam-beam interactions as a function of z is determined by comparing the residual distribution with the parametrization. A correction factor F_z is introduced for each z bin such that the residuals are minimized. The resulting values of F_z range between 0.98 and 1.04 [37].

6.2.2 Resolution for beam-gas interaction vertices

In previous measurements [21], the beam-gas vertex resolution had to be extrapolated from beambeam resolution measurements due to an insufficient number of beam-gas events. With neon gas



Figure 21. Beam-gas vertex resolution in x as a function of vertex track multiplicity for (left) beam 1 and (right) beam 2. Fill 2855 is displayed here as an example. Single markers indicate one resolution measurement for a given vertex track multiplicity. The continuous curves indicate the results of the fits using the parametrization in the same z ranges. The statistical uncertainties are shown by error bars. The beam 1 (beam 2) resolution is determined separately in five (four) z regions. The results for the y resolution are similar.

injection, the increased rate allows a direct measurement of the beam-gas vertex resolution to be made for both beams on a fill-by-fill basis. The measurement principle is similar to that used for the beam-beam vertex resolution. The main differences reside in the vertex selection and in the z dependence of the resolution measurement.

The detector acceptance and extrapolation distance vary considerably within the $\pm 1 \text{ m } z$ range used for the BGI analysis, leading to different distributions of vertex track multiplicity and vertex resolution. Therefore, beam-gas interaction vertex resolutions were measured separately in five *z* ranges for beam 1 ([-1200, -800], [-800, -500], [-500, -250], [-250, 0], [0, 250] mm) and four for beam 2 ([0, 250], [250, 500], [500, 800], [800, 1200] mm). Results are shown in figure 21. The distributions of the residuals between the direct resolution measurements and the resolution parametrization are shown in figure 22. There is no significant bias in the parametrization and the statistical spread of about 0.2 µm can be neglected. The *z* dependence is evaluated with a procedure similar to that used for beam-beam interactions.

The beam-gas vertex resolution can be tested by measuring the single beam width at different z positions. While the physical beam width is unknown, its relative change as a function of z can be predicted from the machine optics and is expected to behave as

$$\sigma_{\text{beam}}(z) = \sigma_{\text{beam}}(0) \sqrt{1 + \left(\frac{z}{\beta^*}\right)^2},$$
(6.3)

assuming the waist position is at z = 0. This is called the "hourglass" effect. In fill 2520, measurements were performed with $\beta^* = 3$ m optics, providing smaller beam sizes and a stronger hourglass effect than in other luminosity calibration fills. Therefore, the beam width measurements for this fill are more sensitive to the resolution. A deviation from the expected *z* dependence of the beam width was observed and additional resolution factors f_z are introduced based on this deviation.


Figure 22. Distributions of residuals between the parametrization and the direct fits for (left) beam 1 and (right) beam 2. Entries are weighted by the number of vertices observed with a given vertex track multiplicity. Fill 2855 is shown here as example.

These corrections factors are measured for each fill separately. With a β^* value of 10m the hourglass effect is negligible (less than 1% at 1m). Similarly the beam width to resolution ratio is also larger and the correction factors have a smaller impact on the measured width. A final verification of single beam width measurements as function of z is shown in figure 23 after applying the correction factors. The resolution has been measured independently for all dedicated fills.

6.2.3 Resolution function for a sample of vertices

The average position of the two sub-vertices from a split primary vertex is used to measure the beam shapes. The observed vertex distribution is a convolution of the density function such as eq. (5.9) with the detector resolution. The resolution function of a sample of vertices for a given vertex type (beam 1, beam 2, beam-beam) is the superposition of the resolution parametrizations based on the number of tracks and the *z* position of each of the vertices in the sample. This resolution function of the sample, R_m , can be parametrized in each coordinate, m = x, *y*, by a sum of a limited number N_g of normalized Gaussian functions g_{k_m} with weights c_{k_m} and widths σ_{res,k_m}

$$R_m = \sum_{k_m=1}^{N_{\rm g}} c_{k_m} g_{k_m}(\sigma_{{\rm res},k_m}).$$
(6.4)

The resolution is unbiased, *i.e.* the functions are centred at the origin. By dividing the full range of the distribution of the resolution estimates of all vertices in the sample into N_g equal-sized bins, the widths and weights are determined by taking the centre and population of the bins, respectively. The weights c_{k_x} and c_{k_y} are the relative resolution weights of the effective resolution functions for x and y. Choosing $N_g = 3$ gives a good description of the sample resolution-function; a larger number of Gaussian functions does not change the results.

Since the transverse distributions of single beams and of the luminous region are expressed as a superposition of Gaussian functions, describing the resolution function also as a sum of Gaussian functions results in an analytical expression for their convolution. In this approximation, the observed transverse distribution for one of the beam components is written as a superposition of



Figure 23. Resolution-deconvolved single beam widths σ_x and σ_y as a function of *z* for fill 2520 ($\beta^* = 3$ m). Values with and without correction factors are shown. Each data point for a given *z* position is an average of all normalized widths from non-colliding bunches. The curved blue line shows the expected evolution of the beam width due the β^* hourglass effect, the shaded blue surface indicates the boundaries corresponding to a 10% uncertainty on β^* . All points are normalized to the values nearest to z = 0.

convolved Gaussian density functions. Each intervening Gaussian width (σ_{mn} , σ_{mw}) is replaced by a resolution-convolved width in the coordinate m = x, y:

$$\sigma_{k_m n}^* = \sqrt{\sigma_{\text{res},k_m}^2 + \sigma_{mn}^2} \quad \text{and} \quad \sigma_{k_m w}^* = \sqrt{\sigma_{\text{res},k_m}^2 + \sigma_{mw}^2}.$$
(6.5)

A similar treatment can be applied to the transverse distribution of the luminous region, using in this case the products of single beam density functions as given by eq. (5.2) [37].

6.3 Measurement of the overlap integral

The knowledge of the three-dimensional bunch shapes $\rho_j(x, y, z)$ is required to evaluate the overlap integral defined in eq. (1.3). To determine the value of the parameters described in eq. (5.11), a fitting procedure is performed that proceeds in several steps. First, the directions of the single

beams are obtained from the positions of beam-gas interaction vertices using all bunches in the filling scheme simultaneously. These beam directions determine the crossing angles and are used to project vertex positions in relatively large z ranges onto a reference plane to ease the fitting procedure. The rest of the procedure is applied only to colliding bunches, taking each bunch-pair individually. To avoid problems with beam drifts and emittance growth, the data are grouped into data-taking periods of about 20 minutes. In a second step, transverse properties of single beams are obtained using the assumption that the shapes in the two transverse coordinates are factorizable. The parameter values obtained are used as initial values for the following step. The third step consists of a fit to beam-gas vertices of bunch pairs of both beams simultaneously, together with beam-beam interaction vertices in their luminous region. This fit is performed in two transverse coordinates separately, still assuming factorizability. In the following, fourth, step the two beams and their luminous region are fitted in both transverse coordinates simultaneously with the full two-dimensional model. The initial parameters of the latter fit are provided by the fit performed in the previous step. Finally, the z positions of the beam-beam vertices are used to determine the longitudinal properties of the bunches. At this point, all parameters needed to evaluate the overlap integral are determined. The individual steps of the procedure are described in more detail below.

The beam angles $\alpha_{m,j}$ for beam j = 1, 2 and axis m = x, y in the laboratory reference frame are measured using beam-gas vertex positions. While the luminosity measurement is based on vertices in the colliding bunch pairs, vertices originating from non-colliding bunch crossings are valuable to measure the crossing angles as they cover the full *z* range owing to the absence of beam-beam background. In a first pass, a straight-line least-square fit is performed to all beam-gas interaction vertices weighting the positions according to their resolution and to an initial estimate of the beam width. An example of a crossing angle measurement using vertex positions directly is shown in figure 17. In a second pass, events are binned in 50 mm *z* intervals with centre z_c . Their transverse vertex position $v_{j,m}$ is projected to z_c using as initial estimates values of $\alpha_{m,j}$ obtained in the first pass. A weighted straight-line fit is then performed through the transverse positions obtained by Gaussian fits to the distributions in the *z* intervals. The statistical uncertainty in the angles is about $10^{-2} \mu rad$. The half crossing angle, which is the angle of interest to measure the overlap integral, is defined as $\phi_m \equiv (\alpha_{m,1} - \alpha_{m,2})/2$.

In the second step of the fitting procedure, the transverse shapes of each individual bunch are analysed using beam-gas interaction data from both beams. The data are divided into different z ranges to combine only data of similar resolution. Three slices for the beam-gas samples in the ranges -1000 mm < z < -250 mm (beam 1) and 250 mm < z < 1000 mm (beam 2) are chosen. The vertices within a slice are projected along the beam direction onto a plane perpendicular to the z axis. This coordinate translation neglects the hourglass effect by assuming a constant beam shape along the beam axis. This is justified since, with $\beta^* = 10 \text{ m}$, the beam is broadened by 0.5% over 1 m. The transverse beam-shape model is fitted to the data in the three slices simultaneously using the double-Gaussian density model of eq. (5.9) and convolution with the resolution model.

For the third and fourth steps, the beam-beam data are divided into z slices and a new fit is performed using beam-gas and beam-beam data simultaneously. The transverse distribution of the luminous region changes as function of z. Since the number of events is sufficient, the analysis is simplified by taking many slices so that in each of these slices the fit model can be approximated by the function value at the centre of the z range. Thus, for the beam-beam vertices, the range -100

mm < z < 110 mm is divided into 18 slices. The properties of the luminous region follow from the single beam properties, thus no new shape parameters are introduced, while the values obtained from the single beam fits are used as initial values. Free amplitude parameters are introduced for each single beam and luminous region z slice to take into account the combined effect of trigger and reconstruction efficiency and absolute rates. The third step consists of a 1-d fit assuming a factorizable description in the x and y coordinate. As is shown below, a two-dimensional model is needed for the transverse shapes of the beams that can accommodate non-factorizable distributions in the x and y coordinates. Such a fit is performed in the fourth step, where a 2-d fit is performed to both coordinates simultaneously.

The model for the transverse distributions in the two fit passes is similar. The single-beam density function is defined in terms of two Gaussian functions for each coordinate. A factorizability parameter f_j (for beam j) is used as defined in eq. (5.11). For the 1-d fits the values of f_j (j = 1, 2) are fixed to 1, decoupling the two transverse coordinates. Following this model, described in section 5.3, the observed transverse vertex distribution per beam for a given z range is fitted with the resolution-convolved width defined in eq. (6.5). The fit parameters are, per beam j, the Gaussian parameters $w_{m,j}$, $\xi_{m,j}$, $\sigma_{mn,j}$ and $\sigma_{mw,j}$ for both axes m = x, y, the factorizability parameter f_j , and a free amplitude $A_{j,k}$ per z slice k. Figure 24 shows 1-d global fit results for the first measurement of the first bunch pair of fill 2852 as example.

The next step introduces values of the factorizability parameters f_j different from unity, and therefore a combined fit coupled in the two transverse coordinates is mandatory. The large number of parameters (18 beam parameters + 24 amplitudes) for the 2-d global fit requires good starting values. The results of the 1-d global fits are used as starting values for the final global fits except for the starting values for the factorizability parameters f_j , which are set to 0.5. An example of a global fit result displaying only one z slice per beam and one luminous region slice is shown in figure 25 for the first bunch and first measurement performed with gas injection and a $\beta^* = 10$ m lattice (fill 2853). Evidence for a significant non-factorizability of the beam shape is discussed further in section 6.5. The χ^2/ndf , with ndf the number of degrees of freedom of the fit, is typically between 1 and 1.1 for the 2-d fit. The non-factorizability of the beams in the transverse coordinates can affect the overlap integral by up to 3%.

Finally, to be able to calculate the overlap integral from eq. (5.5), the convolved bunch length σ_z and the longitudinal position z_{rf} , where the bunch centres coincide longitudinally, must be known. The transverse offsets Δm of eq. (5.5) have to be evaluated at z_{rf} , which is defined by the LHC RF phase. The values of the parameters σ_z and z_{rf} can be obtained from an analysis of the longitudinal vertex distribution of beam-beam interactions using the relations given in eqs. (5.7) and (5.8). A fit is performed to the luminous region distribution of beam-beam interaction vertices for the same data sets as the transverse fits. Following eqs. (5.2) and (5.9), the luminous region distribution $\rho_{bb}(z)$ is represented by the sum of sixteen Gaussian contributions. Complete factorization of the *z* dependence of the bunch distribution is assumption each contribution to the luminous region has a length, amplitude and longitudinal offset, which depend only on the transverse single beam parameters and the quantities z_{rf} , the combination $\sigma_z^2 = \sigma_{z1}^2 + \sigma_{z2}^2$ in eq. (5.6) and an arbitrary overall amplitude parameter A_I . The quantities z_{rf} and $\sigma_{z1}^2 + \sigma_{z2}^2$ are common to all Gaussian contributions. The relative weights of the sixteen contributions follow from eq. (5.12)



Figure 24. Results of a 1-d global fit of one colliding bunch pair and for the *x* coordinate (fill 2852). The first (second) row shows the results of the three beam 1 (beam 2) *z* slices. The third and fourth row show the results of the central five beam-beam *z* slices (the remaining 13 slices used in the fit are not shown for better readability). The double Gaussian fit results are shown as solid red lines. The dashed green lines show the effective *x* resolution functions. The resolution-corrected distributions are shown as solid blue lines.



Figure 25. Results of the global 2-d fit for one bunch pair in fill 2853 as an example, with (left) the central *z* slice of beam 1, (middle) the central *z* slice (out of 21) of beam 2 and (right) the central *z* slice of the luminous region. The fit result of the resolution-corrected beam shape is shown as a three-dimensional shape with amplitude indicated on the vertical scale. The data are displayed as a contour plot above the fit result with an arbitrary colour scale with red indicating the maximum. The pulls of the fit are given on the top with a colour scale ranging from -3 (dark red) to +3 (dark blue).

according to the fraction of luminosity they carry and do not introduce new parameters. Thus, the longitudinal distribution of the luminous region is fitted with only three free parameters and makes use of transverse fit parameters fixed by the global transverse fit. Because the reconstruction efficiency in the VELO is not constant over the full luminous region, the *z* distribution is corrected for the efficiency obtained from simulation.

At this point, all parameters are measured and the overlap integral can be calculated following eqs. (5.5) and (5.6). The statistical uncertainty is evaluated by sampling the multivariate normal distributions of the parameters using the fit results as mean values and the covariance matrix provided by the last fitting step. The resulting statistical uncertainty on the overlap integral evaluated per bunch pair and in 20 minute periods is typically less than 0.5%.

6.4 Generic simulation

The BGI method relies on an accurate beam shape description. To test whether the fitting procedure described in the previous section gives unbiased results, simulated data sets are created with a Monte Carlo method. The development of the more complex 2-d fit model described in section 5.3 was motivated by the possible non-factorizability of the beams in the *x* and *y* directions and the observation that the 2-d properties of the beams could, in principle, be measured with beam-gas interactions. This capability was first tested with simulated data. Results showing evidence for beam non-factorizability are presented in section 6.5.

Datasets of simulated vertices are generated for single beams and the luminous region as follows. Single beam vertex positions v_m (in the axis m = x, y) are generated by sampling eq. (5.9) at a fixed z = 0 position for both beams. A random z position $v_{z,j}$ is then assigned to each vertex in the range -1000 < z < -250 mm for beam 1 and 250 < z < 1000 mm for beam 2. The transverse coordinates of the vertices are translated to $v_{z,j}$ according to the beam direction. The z dependence of the reconstruction efficiency is implemented with a linear reduction of vertices as function of z. Per simulated dataset, about 5×10^4 vertices are generated per beam, similar to the number of events acquired per bunch pair during 20 minutes of data taking.

The shape of the luminous region and the overlap integral are sampled over space and time using the single beam transverse distributions ρ_j (j = 1, 2) defined in section 5 and the longitudinal bunch shape ρ_{zj} , which is assumed to be a single Gaussian function. A sufficiently large sampling volume $S_{vol} = \Delta x \Delta y \Delta z \Delta t$ is chosen ($\pm 0.8 \text{ mm}$ in x and y, $\pm 250 \text{ mm}$ in z and $\pm 1.2 \text{ ns}$ in t). A number N_s of random samples of (x, y, z, t) $\in S_{vol}$ are generated uniformly over the volume. Vertices are retained according to the probability density

$$\rho_{bb}^{s}(x, y, z, t) = \rho_{1}(x, y, z, t)\rho_{z1}(z - ct)\rho_{2}(x, y, z, t)\rho_{z2}(z + ct), \qquad (6.6)$$

where the z and t dependence in $\rho_{1,2}$ just expresses a translation along the beam direction. Following the usual rejection sampling method, a randomly sampled vertex is retained if a uniform random number u assigned to it in a range $[0,A_{bb}]$ satisfies $u \le \rho_{bb}^s(x,y,z,t)$, with the arbitrary constant $A_{bb} \ge \max(\rho_{bb})$. The numerical value of a generated overlap integral is calculated using the fraction N_{bb} of vertices retained compared to the total number of samples N_s generated in the volume S_{vol} with

$$\Omega = 2c \frac{N_{bb}}{N_s} S_{\text{vol}} A_{bb} \,. \tag{6.7}$$

Each simulated primary vertex generated with the above method is assigned a track multiplicity according to the distributions found in data. Using the resolution parametrization measured with data as described in section 6.2, each vertex is assigned a measurement deviation in x and y by sampling a normal Gaussian distribution.

The generated datasets are stored with the same format and are processed with the same algorithms as used for the data. The fitting algorithms are tested with different beam parameters and are validated with simulated datasets before being applied to the data. An example generated with beam parameters similar to those found in the data is shown in figure 26. Detailed studies validate that the simulation input parameters can consistently be recovered with the fitting procedure.

6.5 Evidence of non-factorizable beam shapes

Discrepancies of the order of 3% are observed in visible cross-section measurements performed with the BGI method in the four July 2012 fills when fitted with a model factorizable in the *x* and *y* coordinates. Since the beam-gas interaction vertices provide a complete transverse view of the beams, the factorizability hypothesis can be tested.

A set of simulated data samples with non-factorizable beams ($f_{1,2} = 0$) is fitted with the 2-d global fit model. A factorizable version of the model is obtained by fixing the factorizability parameters at unity. A fit with these parameters left free can describe also non-factorizable distributions. The effect of the pulls showing the difference between a factorizable and non-factorizable model is shown in figure 27. The fit converges towards the correct value of $f_j = 0$ when left free. In addition, the (x, y) distribution of pulls reveals a clearly visible cross-like structure when fitted with a factorizable model, which cannot fully describe the beam shape. The measurement performed on one data set acquired in July 2012 is also shown in figure 27 as an example. The fit correctly



Figure 26. Test of the fit model on simulated data with (left) the transverse view of vertex density of the central z slice at z = 0 and (right) the comparison of simulated data with predictions for different x and y slices of the luminous region central z region. The colour scale is relative with red indicating the highest number of vertices per bin. The dashed lines indicate the x and y slices in the distribution shown on the right. The markers indicate the simulated data while solid lines show the prediction. Both simulated data and predictions are convolved with the resolution. The beam parameters are simulated with values similar to those observed in the data.

describes the beam shapes and the fit pulls are more uniformly distributed. The 2-d fit model converges towards non-factorizable beams and the pulls of the fit assuming a factorizable beam display the same structure as in the distributions simulated with non-factorizable beams.

Data sets have been generated with the parameter $f_j = 0$ and another set of samples with $f_j = 1$. Results for the measured factorizability parameter f_j are shown in figure 28 both for simulated samples and for data. In general, the fit algorithm can reliably measure the value of $f_{1,2}$ except when the beams are close to a single Gaussian shape, where the parameters $f_{1,2}$ have little or no meaning. The distribution of values of f_j in the fits to the data shows a clear dominance of non-factorizable bunch shapes. The factorizability parameter is only meaningful if the beam has a double Gaussian shape in both transverse coordinates. If the beam shape is single Gaussian in one plane only, the beam is by definition factorizable in the model used and f_j can not be measured. The ability to measure f_j depends thus on the "strength" of the double Gaussian shape of the beams defined here as

$$S_{j,m} = 1 - \frac{\sigma_{\min,m,j}}{\sigma_{\mathrm{rms},m,j}}, \qquad (6.8)$$

with

$$\sigma_{\text{rms},m,j}^2 = w_{m,j}\sigma_{mn,j}^2 + (1 - w_{m,j})\sigma_{mw,j}^2$$
(6.9)

for beam j = 1,2 and plane m = x, y. The indices n and w denote the narrow and wide width Gaussian component, respectively, while the width σ_{main} is the width of the Gaussian (n or w) which carries the largest weight. A single Gaussian shape has therefore a vanishing strength parameter.

With few exceptions, all bunch pairs measured in April and July 2012 have a double Gaussian shape and $S_{i,m}$ is significantly larger than zero for both beams and planes. The November fills





Figure 27. Measurement of transverse beam shape factorizability. The upper six graphs are the results of a simulation. From left to right: fit pulls of a beam *z* slice for beam 1, beam 2 and luminous region. The *z* slices [-500, -250] mm for beam 1, [250, 500] mm for beam 2 and [-5, 5] mm for beam-beam are shown here out of the 24 *z* slices. The beams are generated assuming non-factorizability, $f_{1,2} = 0$; the same dataset is used in both rows but the data are fitted with two different models. Top row: the fit assumes factorizable beams ($f_{1,2} = 1$), which is equivalent to the 1-d model. Bottom row: fit with the additional beam factorizability parameters $f_{1,2}$. The fit converges to the correct parameter valuess $f_{1,2} = 0$. The lower six graphs are the results of a fit to data (fill 2855, BCID 1335), which converges to non-factorizable beam shapes in this example.

3311 and 3316 are clearly different from the earlier fills, as all bunches have $S_{j,m}$ values smaller than 0.04.



Figure 28. Fit results for factorizability parameters $f_{1,2}$ for (left) the simulation and (right) all measurements performed in 2012 at $\sqrt{s} = 8$ TeV. In the simulation, different double-Gaussian beam parameters were used and the values of $f_{1,2}$ were set to 0 or 1. For the data, only bunch pairs with a double Gaussian strength $S_{j,m} > 0.02$ and with a fit uncertainty on the factorizability parameter of $\delta(f_j) < 1$ are displayed. The top (bottom) panels refers to beam 1 (beam 2).

6.6 Results and systematic uncertainties

Instantaneous luminosity values for each colliding bunch pair are evaluated using eq. (1.1) and (1.2) with the overlap integral Ω measured with data integrated over intervals of about 20 minutes. The luminosity measurements per colliding bunch pair are used to evaluate the visible cross-section for specific observables (section 4) with

$$\sigma_{\rm vis} = \frac{\mu_{\rm vis} \, \nu_{\rm rev}}{L} = \frac{\mu_{\rm vis}}{N_1 N_2 \, \Omega} \,, \tag{6.10}$$

where μ_{vis} is the visible average interaction rate for the reference observable. In the BGI measurement, the *Vertex* observable is chosen as reference owing to its time stability and low background. Cross-section results for the *Vertex* observable for all dedicated luminosity calibration fills in 2012 at $\sqrt{s} = 8$ TeV with $\beta^* = 10$ m and nominally head-on beams are shown in figure 29, together with a comparison between 1-d and 2-d fits of the transverse bunch properties. One observes that the statistical uncertainties are small and that the results using the 2-d fits are consistent between fills. Comparing results from both fit methods shows the importance of measuring the shapes in two dimensions to take the non-factorizability of the description between the two transverse coordinates into account; the 1-d fits do not display the same consistency. One notes also that the difference between the 1-d and 2-d fit results ranges between 1 and 3% in the July fills, for which the bunches clearly had double Gaussian shapes, while it is < 1% in the November fills, for which the bunches were almost single Gaussian (see section 6.5).

The bunch shapes change over the course of a fill due to emittance growth and other factors, such as beam-beam effects or beam position drifts. Any change in the beam shape influences the overlap integral. Furthermore, the bunch population product decays over time, reducing the luminosity. In contrast, the cross-section is a physical observable and must be stable over time. The



Figure 29. Cross-section results for head-on beams with $\beta^* = 10$ m at $\sqrt{s} = 8$ TeV using the *Vertex* observable, (left) the 2-d fit and (right) the relative difference between the 2-d and 1-d fit. Each data point is a cross-section measurement from a colliding bunch pair using integrated data over about 20 minutes. The measurements are sorted by time and BCID. The fills are indicated in the figure and are separated by dashed vertical lines. Two dotted horizontal lines indicate a $\pm 1\%$ deviation from the central value. The error bars show the statistical uncertainties of the overlap integral.

consistency of the measured cross-section, together with the corresponding overlap integral, rate and bunch population product is shown in figure 30 for one colliding bunch pair for two different fills. While the intensity product decay is typically smooth, the rate fluctuations follow the variations of the overlap integral. The figure also shows that the variation of the overlap integral in adjacent 20 minute intervals is very small (much less than a percent), indicating that the effect of *e.g.* emittance growth and beam drifts during the short intervals is negligible. In the following, sources of systematic uncertainties and their effect on the measurement precision will be described.

The impact of the beam-beam interaction vertex resolution on the cross-section depends on the transverse size of the luminous region. Comparing the results obtained with different ratios of the resolution to this transverse size permits evaluating the corresponding systematic uncertainty. The measurement of the width of the luminous region can be biased by *e.g.* a misalignment in the VELO sensors, which would correlate the position of the sub-vertices and broaden the beam without being detected by the resolution measurement. Considering head-on beams by setting $\Delta x = \Delta y = 0$ in eq. (5.5), one sees that $\Omega \propto (\Sigma_x \Sigma_y)^{-1}$. Assuming similar size for both beams in each transverse coordinate, one can derive from the formalism for pure Gaussian beams that the quantity

$$R = \frac{2\sigma_{\text{res},lx}^2}{4\sigma_{lx}'^2 + \sigma_z^2 \sin^2 \phi_x} + \frac{\sigma_{\text{res},ly}^2}{2\sigma_{ly}'^2}$$
(6.11)

is representative of the importance of the beam-beam resolution in the cross-section measurement. Here, the σ'_{lm} represent the measured values of σ_{lm} and $\sigma_{res,lm}$ the beam-beam resolutions (m = x, y). A value of R = 0 means that the resolution is negligible compared to the beam size. Crosssection measurements with different R values are obtained by using different cuts on the vertex track multiplicity and by using data acquired with a β^* value of 3 and 10m. This procedure gives four sets of results with different values of R.



Figure 30. Evolution of normalized cross-section and related values for a colliding bunch pair for BCID 1335 of (left) fill 2855 and (right) of fill 3311. The cross-section values are normalized to their average while other parameters are normalized to the first data point. Fill 2855 lasted more than eight hours with a beam intensity product decrease of about 10%. Fill 3311 lasted about 4h30 with a beam intensity product decrease of less than 2%; in this fill the luminosity reduction is mostly caused by emittance growth.



Figure 31. Illustration of the evaluation of the systematic uncertainty due to the beam-beam resolution. Measured visible cross-section values for different event samples are shown as a function of *R*. The *R* parameter is calculated with eq. (6.11). Each data point is an average of all measurements from a fill. The error bars are the RMS of the cross-section and *R* values per fill. Plain markers are measurements performed with $\beta^* = 10 \text{ m}$ (6 fills) and open markers are performed with $\beta^* = 3 \text{ m}$ (2 fills). The samples with $R \approx 0.5$ are selected to enhance the effect of the resolution. The dashed line visualizes the determination of the systematic uncertainty as described in the text.

The results for the six fills with $\beta^* = 10$ m and two fills with $\beta^* = 3$ m are combined in figure 31. Measurements performed with the larger β^* and the best resolution (high track multiplicity

cut) provide the smallest *R* value, while a worse resolution combined with the smaller β^* results in a larger *R* value. The clear correlation between the cross-section and the *R* value visible in figure 31 is an indication that the resolution is not perfectly understood. The cross-section obtained for similar *R* values and different β^* are similar. This shows that the difference between the measurements at $\beta^* = 3 \text{ m}$ and $\beta^* = 10 \text{ m}$ can be attributed to the resolution description. The leftmost group of data points are the results presented in figure 29. Those results are obtained with the high track multiplicity cut and with a $\beta^* = 10 \text{ m}$ beam optics. This combination provides the best measurement conditions and those measurements, called here σ_c are used as central value for the final results. The cross-section σ_e obtained by extrapolating the cross-section to R = 0 based on the $\beta^* = 10 \text{ m}$ fills with the low and high track multiplicity cuts (blue and magenta leftmost measurement groups) is used to evaluate the uncertainty due to the beam-beam resolution. The difference of $\Delta \sigma = \sigma_c - \sigma_e = 0.93\%$ between the central value σ_c and the extrapolated value σ_e is taken as systematic uncertainty. This is the largest single contribution to the uncertainty of the cross-section result.

As discussed in section 6.2, a set of correction factors to the beam-gas interaction vertex resolution have been determined to reach consistent beam width measurements at different z positions. The necessity to include correction factors is an indication of additional systematic uncertainties. The overlap integrals have also been evaluated without the resolution correction factors. The full difference between the results with and without correction factors amounts to 0.55%, which is used as the corresponding systematic uncertainty.

A misalignment of the VELO detector can correlate the positions of the sub-vertices and degrade the vertex resolution. This effect is included in the systematic uncertainties assigned to the resolution as described above. However, detector misalignment can also affect the crossing-angle measurement. Various versions of the detector alignment are produced, all of which provide acceptable and comparable results in the overall alignment quality. The different alignment versions arise from introducing different sets of constraints to satisfy internal consistency checks. The same data set is reconstructed with all alignment versions and measurements of the half crossing angle are performed. In the *x* direction, differences of the order of 10 μ rad are found, an order of magnitude larger than the statistical uncertainty. The crossing angle uncertainty in the *y* axis is about 3 μ rad and has a negligible impact on the luminosity. The luminosity uncertainty from the crossing angle correction depends on the bunch width and length and is different for each bunch pair. The full range of 0.45% in the various measured cross-sections for different alignment versions is taken as systematic uncertainty related to this source. The different alignment versions have negligible impact on the bunch shape measurements.

The bunch shape model defined in section 5.3 can describe all observed bunch shapes with a χ^2 /ndf close to one. Nevertheless, the accuracy of the fit procedure as well as the capacity to describe different shapes is verified. The simulation (section 6.4) is used to generate datasets of vertex distributions with different input parameters in order to test the capability to reproduce the input values by the fit procedure. In addition, the systematic uncertainty arising from fitting a double Gaussian model to different simulated shapes is estimated. Single, double or triple Gaussian shapes, or a "Super Gaussian shape"⁶ [43], are tried.

⁶A generalized Gaussian shape with a power parameter different from 2.



Figure 32. Overview of cross-section measurements for head-on periods in fills with $\beta^* = 10 \text{ m}$ at $\sqrt{s} = 8 \text{ TeV}$ with (left) the histogram combining all data from figure 29 and (right) the central value of cross-sections averaged per fill. The dashed vertical line indicates the median value of all measurements. The error bars indicate the standard deviation of all measurements from a fill. The statistical uncertainty of each average is smaller than the marker size.

A given set of input parameters is used to generate a sample of statistically independent datasets, each of which is analysed using the same algorithms as for real data. All shapes give a χ^2 /ndf close to unity and can be considered to be well described by the fit model. The results of the measurements of the simulated data are compared to the input parameters on the basis of the value of the overlap integral. The difference in the results indicates a 0.5% systematic uncertainty due to the fit model and accuracy of the fitting procedure. The double Gaussian fit model used to describe the transverse bunch shapes does not allow a description of a possible third component. For example, a fraction of the protons measured in a bunch could be present in a wider Gaussian shape that would not be measured with a double Gaussian function. In this case, the tails of the measured distributions would have a larger population fraction than expected. The fraction of vertices in the tails beyond the double-sided 99 percentile predicted by the fit, is checked for all measurements with $\beta^* = 10$ m and for the simulated datasets. The tail population is about 2% for the single beams and about 1.5% for the luminous region (while 1% is expected). This tail population, however, is also observed in the simulated datasets, indicating that the higher tail population is a result of the fitting procedure. Therefore, the corresponding bias is already included in the fit model uncertainty given above, and no additional systematic uncertainty is assigned.

Measurements of cross-sections for all colliding bunch pairs in the fills of 2012 with $\beta^* = 10$ m are shown in figure 32. The measurements use the *Vertex* observable as reference and have an RMS spread of 0.54%. Cross-section results are also shown on a fill-by-fill basis in figure 32; the indicated error bar is the RMS of all measurements of the corresponding fill. Since the statistical uncertainty per measurement is significantly less than 0.5%, the observed spread of 0.54% is due to the combination of statistical fluctuations and additional systematic effects. The full RMS is taken as systematic uncertainty on the cross-section. This uncertainty covers uncorrelated bunch-by-bunch uncertainties such as the shape description, which is influenced by the fit model and detector resolutions, or uncertainties in the bunch population measurements.



Figure 33. Determination of the bunch length of a colliding bunch pair with (left) the vertex reconstruction efficiency as function of *z* and (right) the fit to the longitudinal vertex distribution after efficiency correction. The statistical uncertainty on the corrected points is dominated by the limited amount of simulated data. This component of the uncertainty is not shown in the right-hand plot. The data are selected requiring vertices with at least two forward and two backward tracks. The efficiency is normalized to the value at z = 0 to keep similar amplitudes between the raw and corrected data; the absolute scale of the correction does not change the fit results. This example displays data for BCID 1335 in fill 2855. The raw data (black dots) show the distortion resulting from the requirement of having at least two tracks in both directions.

Some structure remains visible in the measurement shown in figure 29, pointing towards additional systematic uncertainties or a less than perfect bunch shape description. Correlations between the cross-sections and the major variables entering into the cross-section measurement (interaction rate, bunch population product, overlap integral, luminous region z position, crossing angle corrections) have been checked and are found to be negligible.

As described in section 6.3, the convolved bunch length $\sigma_{z1}^2 + \sigma_{z2}^2$ and bunch crossing position z_{rf} are measured with a fit to the longitudinal distribution of the luminous region. For each colliding bunch pair measurement, all vertices in the range |z| < 250 mm are selected, regardless of the track directions. This selection reduces the effect of the *z* dependence in the VELO region and limits the distortion of the luminous region shape. On the other hand, some beam-gas interaction vertices can be included in the distribution. An additional measurement of $\sigma_{z1}^2 + \sigma_{z2}^2$ and z_{rf} is performed by requiring that vertices contain at least two forward and two backward tracks to exclude beam-gas interactions. This requirement distorts its longitudinal shape as backward tracks with an origin $z \gtrsim -95$ mm are not detected by the VELO. The reconstruction efficiency ϵ_{vtx} obtained from simulation and corresponding to this requirement is shown in figure 33 and is used to correct the raw distribution. An example of a fit to the longitudinal distribution of the luminous region with this requirement is also shown in the figure for one bunch-pair measurement. All cross-section measurements have been analysed with both track requirements. The difference between the results obtained with the two methods is 0.04% and is taken as systematic uncertainty related to the reconstruction efficiency.

Two dedicated BGI measurements were performed with beams displaced vertically with repect to each other by 180µm in fills 2852 and 2853 with a $\beta^* = 10$ m optics and were preceded and



Figure 34. Measurement of convolved bunch length $\sigma_{z1}^2 + \sigma_{z2}^2$ with displaced beam during fills 2852 and 2853. The fills are indicated in the plot and separated by a dashed vertical line. The measurements are sorted by time and BCID. About 3.5 hours separate the first head-on period in fill 2853 from the period with offset beams explaining the higher values of the bunch length observed during that period. The next head-on period follows the period with offset beams without time interval.

followed by periods with no displacement. The beam offset reduced the overlap integral by a factor of about four. The convolved bunch length $\sigma_{z1}^2 + \sigma_{z2}^2$ measured during those fills is shown in figure 34. Periods where the beams were offset show the same $\sigma_{z1}^2 + \sigma_{z2}^2$ values as when the beams were head-on. With a relative displacement of 180µm, the luminosity is dominated by the interaction of the wide beam components of both beams. On the other hand, the luminosity is dominated by the interaction of the narrow beam components when the beams collide head-on. The equality of the convolved bunch length values for head-on and offset collisions is an indication that the wide and narrow bunch components share the same length, justifying the assumption that the *z* coordinate is factorizable. The observed difference of 0.05% in the cross-section for the periods with offset beams compared to the head-on beams is taken as systematic uncertainty related to the convolved bunch length $\sigma_{z1}^2 + \sigma_{z2}^2$ measurement.

The beam-gas interaction rate is proportional to the residual gas pressure at the interaction point. The BGI method measures the beam shapes with beam-gas interaction vertices, and therefore assumes that the pressure is uniform in the transverse directions. The relative error induced by a pressure gradient is estimated by evaluating the effect of a distortion on the overlap integral considering a constant pressure along the *y* axis and a pressure gradient along the *x* axis at the experimentally determined limit. A measurement was performed in 2010 during fill 1422 to verify the homogeneity of the pressure in the *x* direction by displacing the beams by 0.3 mm and has been used in the luminosity measurement for 2010 [11]. The relative uncertainty on Ω introduced by the pressure gradient is at most 0.03%.

The VELO transverse dimensions, which fix the absolute scale of the vertex transverse distribution measurements, were checked in the laboratory at different temperatures on individual silicon sensors. The relative uncertainty on the cross-section determination due to the transverse scale uncertainty is estimated to be at most 0.05%.

The FBCT response is approximatively linear with respect to bunch intensities. However, small deviations from linearity lead to a non zero offset when extrapolating to zero bunch intensity. This offset can be inferred from the combination of all cross-sections measured during a fill, taking advantage of the fact that the cross-section is independent of the choice of the bunch pair. An analysis of the FBCT non-linearity is performed as outlined in ref. [27]. A fit is performed with the offsets and the improved cross-section value as free parameters. As expected, the χ^2/ndf values are improved by this procedure. All cross-section results obtained after applying the FBCT offset corrections for fills with $\beta^* = 10$ m and $\sqrt{s} = 8$ TeV give a distribution with an RMS of 0.48% instead of 0.54% without the correction. The central value is changed by 0.04%. This small deviation is not applied to the final result, but is taken as systematic uncertainty related to a potential FBCT offset. The systematic uncertainties introduced by the bunch population determination are described in section 3.

A difference of 0.2% is observed between the background-subtracted interaction rate measurement with the restricted *Vertex* observable and the corresponding unrestricted observable. This difference is attributed to the background subtraction and assigned as systematic uncertainty. The reference cross-section used for physics data taking is based on the *Track* observable, which is affected by beam-gas interaction background when the neon gas injection system is used for the BGI calibration data taking. To transport the visible cross-section based on the *Vertex* observable to that based on the *Track* observable, their ratio is measured in periods without gas injection. A variation of 0.2% is observed, which is taken as systematic uncertainty. Using the relation between the *Vertex* and *Track* observable of $\mu_{\text{Track}}/\mu_{\text{Vertex}} = 1.106$, the final calibration result is $\sigma_{\text{Track}} = 60.62 \pm 0.87$ mb. A summary of all uncertainties is provided in table 7. The values shown for the other energies will be discussed below.

6.7 Beam-gas imaging results at lower energies

The analysis presented in this paper focuses on the calibration of the reference cross-section at $\sqrt{s} = 8 \text{ TeV} (pp)$ since it gives the most precise results. Comparable results are obtained with the BGI method for the 2013 *pp* calibration at $\sqrt{s} = 2.76 \text{ TeV}$ and the 2011 *pp* calibration at $\sqrt{s} = 7 \text{ TeV}$.

The $\sqrt{s} = 2.76$ TeV calibration was performed in three dedicated fills (3555, 3562 and 3563). Trigger conditions were identical to those used in July and November 2012 and the neon gas injection system was active. The luminosity measurement and evaluation of systematic uncertainties follows the same procedure as with $\sqrt{s} = 8$ TeV data. Cross-section results for $\sqrt{s} = 2.76$ TeV are shown in figure 35. The RMS of the measurements is 1.3%.

A half crossing angle of 885 µrad was chosen to avoid collisions between satellite and main bunches. Differences observed in the data as compared to 2012 data are discussed below. The beams have a double Gaussian shape and are significantly non-factorizable (f_j is close to zero). However, the factorizability is more difficult to measure than with $\sqrt{s} = 8$ TeV data and the uncertainties on the f_j parameter are larger. The beam-beam resolution has a small impact on the cross-section compared to 2012 (uncertainty of 0.40%), owing to the comparatively large transverse bunch size. On the other hand, the uncertainty related to the beam-gas vertex position resolution

Table 7. Relative systematic uncertainties on the reference cross-section for the BGI calibrations at \sqrt{s} values of 8, 7 and 2.76 TeV (in %). The uncertainties are divided into groups affecting the measurement of the overlap integral (section 6), the bunch population product (section 3) and the rate measurement (section 4).

Source of uncertainty	8 TeV	7 TeV	2.76 TeV				
Overlap integral measurement							
Beam-beam resolution	0.93	1.00	0.40				
Beam-gas resolution	0.55	2.80	1.31				
Detector alignment	0.45	0.30	0.90				
Fit model	0.50	1.50	0.50				
Measurement spread	0.54	1.00	1.30				
Reconstruction efficiency	0.04	0.20	0.17				
Bunch length and z_{rf}	0.05	0.10	0.10				
Pressure gradient	0.03	0.03	0.03				
VELO transverse scale	0.05	0.05	0.05				
Statistical uncertainty	0.01	0.25	0.03				
Bunch population uncertainties							
FBCT non-linearity	0.04	0.05	0.05				
DCCT population product	0.22	0.24	0.28				
Ghost charge	0.02	0.10	0.07				
Satellite charge	0.06	0.25	0.08				
Rate measurement							
Background subtraction	0.20	0.05	0.20				
Ratio of observable <i>Track</i> to <i>Vertex</i>	0.20	n.a.	0.20				
Final $\delta\sigma_{\mathrm{Track}}/\sigma_{\mathrm{Track}}$	1.43	3.50	2.20				

has significantly larger impact than in 2012, due to the lower number of vertices in the luminous region. The cross-section difference measured with and without all beam-gas resolution correction factors amounts to 1.31%. The uncertainty related to the detector alignment of 0.9% is estimated at twice the value obtained in 2012 because the crossing angle correction is about twice as large. The convolved bunch length $\sigma_{z1}^2 + \sigma_{z2}^2$ measurement plays a more important role compared to the $\sqrt{s} = 8$ TeV data due to the larger crossing angle correction. An uncertainty of 0.1% is assigned to the determination of $\sigma_{z1}^2 + \sigma_{z2}^2$ and z_{rf} and 0.17% to the reconstruction efficiency. The FBCT offset fit changes the cross-section by 0.05% and reduces the overall RMS to 1.1%. The systematic uncertainty related to the ghost charge amounts to 0.07%. Uncertainties for the observable background subtraction and fit model are taken from the 2012 measurements. The reference cross-section for the $\sqrt{s} = 8$ TeV data, the reference cross-section used for physics data taking is based on the *Track* observable. Using the relation between the *Vertex* and *Track* observable of $\mu_{\text{Track}}/\mu_{\text{Vertex}} = 1.135$, the final calibration result is $\sigma_{\text{Track}} = 52.7 \pm 1.2$ mb.

In the 2011 *pp* calibration, the main differences with the situation in 2012 are the absence of the neon gas-injection system and a different trigger configuration. Therefore, the beam-gas interaction rate is a factor 20 lower than in 2012. To partly compensate for the lower rate, measurements



Figure 35. Cross-section results with head-on beams with $\beta^* = 10$ m at $\sqrt{s} = 2.76$ TeV for the *Vertex* observable, with (left) the individual measurements and (right) the histogram of the values. The individual measurements are made per colliding bunch pair and 20 minutes time integration and are sorted by time and BCID. The different fills are separated by a dashed vertical line in the figure. Two dotted horizontal lines show the $\pm 1\%$ deviation from the central value. The measurement spread has a 1.3% RMS. The median value is indicated by a dashed vertical line in the right-hand plot.



Figure 36. Cross-section results at $\sqrt{s} = 7$ TeV in 2011 for the *Track* observable, with (left) the individual measurements and (right) the histogram of the values. The individual measurements are made per colliding bunch pair, for a one hour integration time, and are sorted by time and BCID. The different fills are separated by a dashed vertical line in the figure. Two dotted horizontal lines show the $\pm 1\%$ deviation from the central value. The measurement spread has a 2.0% RMS. The median value is indicated by a dashed vertical line in the right-hand plot.

are performed in one-hour periods, potentially introducing effects of emittance growth and beam drifts. In addition, data are available in one dedicated calibration fill only. Cross-sections results for $\sqrt{s} = 7$ TeV are shown in figure 36. The RMS of the measurements is 2.0%.

Collisions between satellite and main bunches were observed at $z = \pm 375$ mm due to the low half crossing angle of 270 µrad. Therefore, beam-gas vertices in the region -550 mm < z < 550 mm are discarded to exclude beam-beam vertices in the single beam selection. This requirement, however, reduces the number of vertices available for the single beam measurement. Furthermore, the remaining vertices are measured with a worse resolution leading to a bigger impact on the beam size measurement. The limited amount of beam-gas vertices reduces the accuracy of the resolution determination and the correction factors described in section 6.2.2 cannot be determined. An uncertainty of 10% is assumed for the beam-gas vertex resolution leading to an uncertainty of 2.8% on the cross-section calibration.

The ratio of the resolution width to the beam size is smaller in 2011, potentially reducing the uncertainty resulting from the beam-beam resolution. However, the same uncertainty as for $\sqrt{s} = 8$ TeV is conservatively assumed for the impact of the beam-beam vertex resolution. Given the smaller crossing angle, the uncertainty related to the detector alignment effects is estimated at 0.3%. The beam factorizablility is more difficult to measure in 2011 due to the lower number of vertices. The beams are mostly non-factorizable (f_i is typically close to zero).

Owing to the low beam-gas induced background, the reference cross-section can be based on the *Track* observable directly without using the ratio *Vertex* to *Track*. A global satellite fraction correction of 0.78% is applied to the result shown in figure 36. The correction cannot be applied on a per bunch basis as the LDM instrument was not operational yet for this fill. A summary of all uncertainties is provided in table 7. The final calibration result for $\sqrt{s} = 7$ TeV based on the *Track* observable is $\sigma_{\text{Track}} = 63.0 \pm 2.2$ mb. The difference with respect to the 8 TeV calibration result is not significant given the uncertainties, which are mostly uncorrelated.

This value is consistent with the result obtained with the BGI method described in a previous LHCb publication [11]. The improvement in the present result is due to the better bunch population measurement, while the uncertainties in the overlap integral are similar.

7 Van der Meer scan method

7.1 Experimental conditions

VDM scans at $\sqrt{s} = 8$ TeV (*pp*) were performed in LHCb during dedicated LHC fills in the 2012 running period, one in April and one in July. The VDM scan method is presented in detail for these calibrations. The calibrations at $\sqrt{s} = 7$ TeV (*pp*) and at $\sqrt{s_{NN}} = 5$ TeV (*p*Pb and Pb*p*) are summarized in section 7.10. The list of dedicated luminosity calibration fills can be found in table 1 and the scan parameters are listed in table 8. Four *x*-*y* scan pairs were performed in April and six in July. In all scans the two beams were moved symmetrically, typically covering a $\pm 6\sigma_n$ range of beam separations, where σ_n is the nominal beam width assuming nominal values of β^* and transverse normalized emittance $\epsilon_n = 3.75 \,\mu$ m rad. The last scan pair in each fill had a nominal working point at a relatively large offset $\Delta x_0, \Delta y_0 \approx +2\sigma_n$. These offset scan pairs are only used for cross-checks and are not considered in the cross-section determination because of their high sensitivity to systematic effects (*e.g.* beam orbit drift, factorizability, linear correlations). The first and fourth scans in April were performed along x'' and y'' axes that are rotated with respect to the principal axes of the LHC optics. This particular scan pair is analysed, but not used in the final result.

conditi	ions availa	able for th	e measuremen	t				
Pair	Scans	Axes	Offset	Number	Separation	Duration		
				of steps	range	step (s)	total (min)	
pp at $\sqrt{s} = 8$ TeV, Apr 2012, Fill 2523								
1	2/3	x/y	0/0	31/31	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	12/12	
2*	1/4	x''/y''	0/0	33/31	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	13/12	
3	5/6	x/y	0/0	17/17	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	7/7	
4*	7/8	y/x	$\sim 2\sigma_n/2\sigma_n$	17/17	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	7/7	
pp at $\sqrt{s} = 8$ TeV, Jul 2012, Fill 2853								
1	1/2	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	12/12	
2	3/4	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	12/12	
3	5/6	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	12/12	
4	7/8	x/y	0/0	27/27	$\pm 6\sigma_n/\pm 6\sigma_n$	15	12/12	
5*	9/10	x/y	0/0	51/51	$\pm 6\sigma_n/\pm 6\sigma_n$	15	21/21	
6*	11/12	y/x	$\sim 2\sigma_n/2\sigma_n$	27/27	$\pm 6\sigma_n/\pm 6\sigma_n$	15	12/12	
pp at $\sqrt{s} = 7$ TeV, Oct 2011, Fill 2234								
1	1/2	x/y	0/0	31/31	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	15	13/13	
2	3/4	x/y	0/0	17/17	$\pm 6\sigma_n/\pm 6\sigma_n$	15	8/8	
3	5/6	x/y	0/0	17/17	$\pm 6\sigma_n/\pm 6\sigma_n$	15	8/8	
<i>p</i> Pb at $\sqrt{s_{NN}} = 5$ TeV, Jan 2013, Fill 3505								
1	1/2	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 6\sigma_{\rm n}$	30	17/17	
2	3/4	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 5\sigma_{\rm n}$	30	17/17	
Pbp	at $\sqrt{s_{NN}}$ =	= 5 TeV, 1	Feb 2013, Fill	3542				
1	1/2	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 5\sigma_{\rm n}$	30	17/17	
2	3/4	x/y	0/0	27/27	$\pm 6\sigma_{\rm n}/\pm 5\sigma_{\rm n}$	30	17/17	

Table 8. Parameters of VDM scans. The scan pairs marked with an asterisk are not used in the determination of the central value of the cross-section (as explained in the text). The step duration indicates the period of stable conditions available for the measurement.

Beam movements recorded with LHC beam position monitors (BPMs) upstream and downstream of LHCb are shown in figure 37. The BPM measurements are not used in the analysis as their time stability is insufficiently accurate. During a scan the beam separation values are calculated on the basis of a detailed model of the LHC magnets and are set by the accelerator control system. Dedicated length scale calibration scans are performed in each VDM calibration fill to experimentally verify and calibrate the beam displacements using the precisely known geometry of the VELO (see section 7.5).

During VDM scan sessions, a large fraction of the available trigger bandwidth was allocated to randomly triggered bunch crossings; 20kHz were devoted to the crossings with collisions, 2kHz to the crossing slots where only one of two beams was present, and 0.5kHz to the empty crossing slots. In addition, starting from the July 2012 session, beam-gas events were recorded simultaneously. Due to the small beam-gas interaction rate of a few hundred Hz in total, the collected events are only used for a cross-check of the beam positions.

The average decay time of the bunch population product N_1N_2 was 36 (70) hours in the April



Figure 37. Beam positions recorded with the LHC beam position monitors around the LHCb interaction point in the (left) April and (right) July VDM sessions. The initial values are set to zero. The top (bottom) panel shows horizontal (vertical) positions. In all scans the beams are moved symmetrically. The beam movements around 6:30 (22:40) in April (July) correspond to the length scale calibration by the constant separation method.

(July) session. The value of N_1N_2 changed by about 1% during a scan pair. Therefore, the rates are normalized by N_1N_2 of each colliding bunch pair at every scan point by defining a specific average number of interactions per crossing

$$\mu_{\rm sp} = \frac{\mu_{\rm vis}}{N_1 N_2} \,. \tag{7.1}$$

To reduce the noise associated with the $N_{1,2}$ measurements, the data for each beam are approximated by a smoothing spline [44].

In addition to the bunch population changes, the luminosity stability may be limited by changes in the bunch profiles, *e.g.* by emittance growth. The luminosity stability is checked several times during the scans when the beams were brought back to their nominal position. The evolution of μ_{sp} , averaged over bunch pairs, is shown in figure 38. The average luminosity decay time is measured to be 29 (58) hours in the April (July) session, which is largely due to the decay in the bunch population product. The average luminosity drop caused by emittance growth (drop of μ_{sp}) amounts to 1.2% (0.5%) during the entire calibration session in April (July). The scan points have been taken from lower to higher Δx , Δy values, therefore, the luminosity drop due to emittance growth effectively enhances the rate at negative values and reduces the rate at positive values, so that the net effect on the integrals in eq. (5.3) cancels to first order since the curves are symmetric. Therefore, the systematic error due to the emittance growth is considered to be negligible.



Figure 38. Evolution of the specific average number of interactions per crossing (μ_{sp}) at the nominal headon beam positions during the (left) April and (right) July VDM sessions. In each scan the nominal point was measured three times and the average over bunches is plotted with open markers. The value at the actual zero beam separation is predicted for each scan pair (excluding offset and tilted scans) and shown with a horizontal band. The difference with respect to the values at zero nominal separation is due to the working point not being exactly at zero beam separation. The beam positions were not re-optimized during the sessions.

As will be shown in section 7.8, the measurement using the fifth scan pair in July is more sensitive to beam orbit drifts, because it was performed at a slightly offset working point and took longer than usual (42 instead of 24 minutes). Since the orbit drifts cannot be precisely corrected for, this scan pair is not used for the central value of the result but is accounted for in the systematic uncertainty. In total, for the cross-section measurement, two *reference* scan pairs are used from the April session and four from the July session.

7.2 Overlap integral model

The rate measurements from the orthogonal VDM scans performed provide no information on the factorizability of the single beams. However, the cross-section measurement is sensitive to the latter. In order to impose constraints on the factorizability, the BGI measurements performed in the same fills on the same bunch pairs can be used. To facilitate such approach, it is advantageous to use an identical model of the single beam densities for both BGI and VDM analyses. In this section, a model for the overlap integral as function of beam separation is developed based on the double Gaussian beam shape description from section 5.3. It should be noted that the application of the classic VDM method in the case of factorizable beam distributions (see eq. (5.3)) only requires a good empirical description of the VDM scan data.

The formalism discussed in section 5.2 is only valid in the case of a horizontal beam crossing plane. However, in the April calibration the crossing plane was rotated along z, which necessitates the following extended treatment. In this more general case, the exponent in the overlap integral for Gaussian beams from eq. (5.5) gains a dependence on the product of the transverse beam separations $\Delta x \Delta y$ and the normalization factor is modified. It is useful to extend the definition of the effective convolved widths of the luminous region Σ_m (m = x, y) from eq. (5.6) (taking $\cos \phi_m \approx 1$) as

$$\Sigma_m^2 = \sigma_{m1}^2 + \sigma_{m2}^2 + (\sigma_{z1}^2 + \sigma_{z2}^2) \tan^2 \phi_m, \qquad (7.2)$$

so that Σ_y also has a term that depends on the corresponding half crossing angle in the *yz* plane ϕ_y . The general expression for the overlap integral (in the case of pure Gaussian beams) is then given by

$$\Omega(\Delta x, \Delta y | \Sigma_x, \Sigma_y, \lambda) = \sqrt{1 - \lambda^2} \exp\left(\frac{\lambda \Delta x \Delta y}{\Sigma_x \Sigma_y (1 - \lambda^2)}\right) g(\Delta x | \boldsymbol{\sigma} = \Sigma_x \sqrt{1 - \lambda^2}) g(\Delta y | \boldsymbol{\sigma} = \Sigma_y \sqrt{1 - \lambda^2}), \quad (7.3)$$

where g is the normalized Gaussian probability density function and λ is the correlation coefficient, which is related to the bunch lengths and the half crossing angles by

$$\lambda(\Sigma_x, \Sigma_y) = \frac{C_{xy}}{\Sigma_x \Sigma_y} + \bar{\lambda}, \quad C_{xy} = (\sigma_{z1}^2 + \sigma_{z2}^2) \tan \phi_x \tan \phi_y, \quad \bar{\lambda} = \frac{\lambda_1 + \lambda_2}{2}.$$
(7.4)

The variables $\lambda_{1,2}$ are the individual bunch correlation coefficients, which take into account a possible rotation of the principal axes of the bunches (around the direction of motion). It is seen from eq. (7.4) that C_{xy} is only non-zero when both ϕ_x and ϕ_y are non-zero, which occurs when the beam crossing plane is tilted (*i.e.* neither strictly horizontal nor vertical).

The overlap integral in the double Gaussian model, which is discussed in section 5.3, is given by eq. (5.12) with each partial overlap integral expressed as

$$\Omega_{i_x, i_y, j_x, j_y} = \Omega(\Delta x, \Delta y | \Sigma_{x, i_x j_x}, \Sigma_{y, i_y j_y}, \lambda(\Sigma_{x, i_x j_x}, \Sigma_{y, i_y j_y})),$$
(7.5)

where i_x , i_y , j_x , j_y take the values n and w. Here, it is implicitly assumed that all Gaussian components have the same centres, same bunch lengths and same correlation coefficients (*i.e.* C_{xy} and $\bar{\lambda}$). The weights in eq. (5.12) depend on the factorizability parameter f_j and two weight parameters $(w_{x,j} \text{ and } w_{y,j})$ for each beam j = 1, 2.

The rate that is measured in VDM scans only provides direct information on the beam overlap integral and not on the single beam densities. Therefore, it is convenient to parametrize the model of the overlap integral using the effective convolved widths rather than the underlying beam width parameters. Only three of the four effective convolved widths per plane are linearly independent, thus two parameters are sufficient to quantify their relative magnitudes, with the choice

$$R_m = \frac{\Sigma_{m,\text{ww}}}{\Sigma_{m,\text{nn}}}, \quad R_m \ge 1,$$
(7.6)

$$A_m = \frac{\sum_{m,\text{nw}}^2 - \sum_{m,\text{wn}}^2}{\sum_{m,\text{ww}}^2 - \sum_{m,\text{nn}}^2}, \quad A_m \in [-1,1],$$
(7.7)

for m = x, y. In terms of beam width parameters we have

$$A_m = -\frac{(\sigma_{m1,w}^2 - \sigma_{m1,n}^2) - (\sigma_{m2,w}^2 - \sigma_{m2,n}^2)}{(\sigma_{m1,w}^2 - \sigma_{m1,n}^2) + (\sigma_{m2,w}^2 - \sigma_{m2,n}^2)},$$
(7.8)

thus A_m describes the asymmetry between the differences in the widths of the wide and narrow components of the two beams. The special case $A_m = 1$ corresponds to $\sigma_{m1,n} = \sigma_{m1,w}$, *i.e.* the

distribution of beam 1 is Gaussian in the m axis. In the parametrization of eqs. (7.6) and (7.7), the effective convolved widths are expressed as

$$\begin{bmatrix} \Sigma_{m,nn}^2 & \Sigma_{m,nw}^2 \\ \Sigma_{m,wn}^2 & \Sigma_{m,ww}^2 \end{bmatrix} = \Sigma_{m,nn}^2 \begin{bmatrix} r_{m,nn}^2 & r_{m,nw}^2 \\ r_{m,wn}^2 & r_{m,ww}^2 \end{bmatrix}$$
$$= \Sigma_{m,nn}^2 \left(\begin{bmatrix} 1 & \frac{R_m^2 + 1}{2} \\ \frac{R_m^2 + 1}{2} & R_m^2 \end{bmatrix} + A_m \frac{R_m^2 - 1}{2} \begin{bmatrix} 0 & 1 \\ -1 & 0 \end{bmatrix} \right),$$
(7.9)

where the ratios $r_{m,ij}$ are defined as $r_{m,ij} \equiv \Sigma_{m,ij} / \Sigma_{m,nn}$.

For the actual fitting procedure, it is advantageous to use scale (or width) parameters that have a model-independent meaning. For example, the RMS of the luminosity profiles at $\Delta x = 0$ and $\Delta y = 0$ are such parameters, which can be easily estimated from the scan data to obtain starting values for the fit. Using the parameters defined above, the RMS S_x of the luminosity profile at $\Delta y = 0$ is given by

$$S_x^2 = \frac{\int \Delta x^2 \Omega(\Delta x, 0) d\Delta x}{\int \Omega(\Delta x, 0) d\Delta x} = \sum_{x,nn}^2 \frac{\sum_I \frac{w_I}{r_{y,iyjy}} r_{x,i_xj_x}^2 (1 - \lambda^2 (\sum_{x,nn} r_{x,i_xj_x}, \sum_{y,nn} r_{y,i_yj_y}))}{\sum_I \frac{w_I}{r_{y,iyj_y}}}$$
(7.10)

and similarly for the other coordinate. The values $\Sigma_{x,nn}$ and $\Sigma_{y,nn}$ are obtained by solving the system of equations defined by the above equation. Finally, in the double Gaussian model, the shape of the overlap integral as function of beam separation is parametrized with the following 14 parameters

$$S_x, S_y, R_x, R_y, A_x, A_y, f_1, f_2, w_{x,1}, w_{y,1}, w_{x,2}, w_{y,2}, C_{xy}, \lambda.$$
(7.11)

Only some of those parameters remain free for the cross-section determination as explained in the following section.

7.3 Cross-section determination

All colliding bunch pairs are analysed individually. The data are fitted simultaneously for pairs of x and y scans. The value of μ_{vis} at each step k is described with

$$\mu_{\text{vis}}(\Delta x_k, \Delta y_k) = \sigma_{\text{vis}} N_1 N_2 \Omega(\Delta x_k - \Delta x_0, \Delta y_k - \Delta y_0) + N_1 \mu_{\text{be}}^{\text{sp}}(t_k) + N_2 \mu_{\text{eb}}^{\text{sp}}(t_k), \quad (7.12)$$

where $N_{1,2}$ are the bunch intensities and $\mu_{be(eb)}^{sp}$ is the specific μ value of the beam-gas background for beam 1(2). Two position parameters, Δx_0 and Δy_0 , were introduced to account for the fact that the luminosity may reach a maximum at a non-zero nominal separation (Δx , Δy) due to an imperfect alignment of the beams.

The last two terms in eq. (7.12) are due to beam-gas interactions and are proportional to the beam intensity and the residual gas pressure. The value of $\mu_{be(eb)}^{sp}$ is determined using events from data taken simultaneously for non-colliding bunches for each scan independently and is typically 2×10^{-14} (1×10^{-14}). In most cases the pressure was very stable during VDM scans. During the first pair of scans in the July session a drop of about 10% was observed due to the neon gas injection system being used beforehand. In order to take this into account, an exponential dependence on time is assumed for the first pair of scans in July, while in all other cases constant specific background is assumed.

As already discussed above, due to the pattern of two orthogonal movements used in the scans, the data provide no information on the factorizability parameters and the linear correlation parameters. Therefore, in order to obtain the required additional information, we use the BGI measurements performed in the same fills as the VDM scans. The BGI analysis gives $f_{1,2}$ values compatible with zero. In the VDM analysis the $f_{1,2}$ parameters are fixed to zero. In this fully non-factorizable case, the two weight parameters w_x and w_y for each beam are perfectly anticorrelated as seen from eq. (5.11). Therefore, only one weight parameter per beam is used for the VDM analysis. The value of C_{xy} is computed using the BGI measurements of the convolved bunch length and crossing angles. No significant linear correlation in the beam distributions is observed, thus the value of $\overline{\lambda}$ is set to zero. The available number of counts per scan point and the relatively coarse scanning grid do not allow measuring both R_m and A_m . The parameter R_m describes the main features of $\Sigma_{m,ij}$, eq. (7.9), and is determined in the fit, while A_m is fixed to zero for the determination of the central value of σ_{vis} . The systematic uncertainties arising from the assumptions on the fit parameters are discussed in section 7.7. In total nine parameters remain free, including the visible cross-section.

Initially, the fit is performed using the uncertainty estimates obtained from the data $\delta \mu_{\text{vis}} = \sqrt{1/N_0 - 1/N}$, where N_0 is the number of empty events in a total of N events. For small values of μ_{vis} , these uncertainty estimates are correlated with the μ_{vis} values themselves, thus biasing the data weights and the fit result. To mitigate this problem, the data are fitted a second time using uncertainties $\delta \mu_{\text{vis}} = \sqrt{\exp(\hat{\mu}_{\text{vis}})/N - 1/N}$, which are based on the predictions $\hat{\mu}_{\text{vis}}$ of the first fit.

For presentation purposes we define a corrected specific interaction rate per crossing by rearranging eq. (7.12)

$$\mu_{\text{vis},i}^{\text{sp}} = \frac{\mu_{\text{vis}}(\Delta x_i, \Delta y_i) - N_1 \mu_{\text{be}}^{\text{sp}}(t_i) - N_2 \mu_{\text{eb}}^{\text{sp}}(t_i)}{N_1 N_2}.$$
(7.13)

Then, the function $\sigma_{vis}\Omega(\Delta x, \Delta y)$ represents the fit to the corrected data points $\mu_{vis,i}^{sp}$. An example of a fit for a single bunch pair is shown in figure 39 for one scan pair. The measured cross-section values and the values of χ^2 per degree of freedom, χ^2/ndf , obtained for all bunch and scan pairs are shown in figure 40. Higher values of σ_{vis} are obtained for the fifth scan pair in July. The associated uncertainty is discussed in detail in section 7.8. For the April scan session, the values of χ^2/ndf are on average higher than one. This can be explained by the smaller beam size and the fact that the uncertainty of the beam separation is not taken into account. Finally, the cross-section is obtained for each calibration session by calculating the weighted average of all measurements from reference scan pairs.

The sources of systematic effects that influence the result of the fit described with eq. (7.12) can be grouped into two major categories. First, there can be systematic effects related to the inputs of the fit, namely bunch intensities (described in section 3), rates (of signal and background) and beam positions (sections 7.4–7.6). Second, the influence of the VDM profile model, the technical aspects of the fitting procedure and discrepancies between repeated measurements are described in sections 7.7 and 7.8. The variation in the obtained cross-section among bunch pairs is used to estimate the uncertainty due to the relative bunch intensity measurement. The uncertainty due to a potential non-linearity of the FBCT device is estimated to be about 0.05% using the same method as described in section 6.6. Moreover, by using the independent measurements from the ATLAS



Figure 39. Fit of VDM profiles for a single bunch and scan pair in the (top) April and (bottom) July VDM sessions. The left (right) panels show data and fit predictions corresponding to the x (y) scan. The data points are represented with circles, while the fitted curve is shown as a solid line. The error bars are smaller than the symbol size. The fit pulls are displayed below each fit projection and show no systematic structure. There are three data points at $\Delta m = 0$ as the nominal point was measured three times for each scan.

BPTX system [38], a discrepancy in the final result of about 0.1% is found, which is assigned as an additional systematic uncertainty.

7.4 Rate measurement

For the absolute calibration, the beam-gas related backgrounds are subtracted taking into account the difference in bunch populations. There is a statistical uncertainty associated with the measured specific background per proton used for subtraction. The value of the specific background is shared among cross-section fits of bunch pairs and introduces a correlation, which is taken into account when combining the measurements. This uncertainty is estimated to be less than 0.1%.



Figure 40. (Top) measured cross-section values and (bottom) the corresponding values of χ^2/ndf for all bunch pairs and for non-offset scan pairs in the (left) April and (right) July VDM sessions. Measurements within one scan pair are separated from the rest by a larger distance.



Figure 41. Luminous region longitudinal position and size as function of beam separation for a scan pair in the (left) April and (right) July VDM sessions. The values for other scan pairs are similar. Markers indicate the mean while the bands between dotted lines contain 68% of the vertices. In April, the luminous region moves in both horizontal and vertical scans because of the crossing angle configuration. The inversion of the slope in the horizontal scan is due to the change of sign of the crossing angle between the two sessions.

The beam-beam related background is estimated to be 0.1–0.2% by taking the difference between the visible cross-sections of the restricted and the non-restricted *Track* observables. For the central value of the visible cross-section we use the less affected *Track* observable, while the full difference is taken as systematic uncertainty.

In the presence of a non-zero beam crossing angle, the luminous region position varies with transverse beam separation. The track reconstruction efficiency is not uniform as function of the primary vertex longitudinal position. Therefore, a correction to the μ_{vis} values is applied using the same principle as described in section 4.

The longitudinal position and size of the luminous region are measured for each step using a single Gaussian fit to the selected beam-beam vertices, taking into account the vertex reconstruction



Figure 42. Efficiency correction factors for the *Track* observable for a scan pair in the (left) April and (right) July VDM sessions. The values for other scan pairs are similar.

efficiency as a function of z. The quantities are linearly extrapolated to large separations, where the luminosity vanishes and not enough vertices are available (< 1000). The measured longitudinal parameters of the luminous region are shown in figure 41 and the calculated correction factors to the observed $\mu_{\rm vis}$ for each scan and step are shown in figure 42. It can be noted that the correction factors are approximately linear as function of separation. Therefore, no large effect on the VDM profile width is expected. The effect of the correction amounts to +0.32% (-0.03%) in the April (July) calibration. An estimate of the associated systematic uncertainty is made using a comparison with an unaffected observable (*Calo*). In the approximation of pure Gaussian beams, the rate at zero beam separation is proportional to $\sigma_{vis}/(\Sigma_x \Sigma_v)$, see eqs. (1.1), (1.2) and (5.5). Equivalently, the measured cross-section is proportional to the rate at zero separation and the product of the widths of the VDM profiles. It is useful to separate the correction factor dependence on ξ_{lz} (figure 10) in a linear and a higher order part. The former only affects the rate at zero separation, while the latter mainly affects the widths. The residual (after correction) slope of the ratio $\mu_{\text{Track}}/\mu_{\text{Calo}}$ as function of ξ_{lz} is measured and normalized to the ratio at $\xi_{lz} = 0$. The ξ_{lz} value at the working point is multiplied by that slope to obtain the uncertainty on the cross-section due to the linear part of the correction. The uncertainty due to the higher order part of the correction is estimated by comparing the product $\Sigma_x \Sigma_y$ measured using the *Track* and the *Calo* observable. The two uncertainties are summed linearly to obtain the systematic uncertainty of 0.13% (0.07%) for the April (July) calibration.

7.5 Length scale

The nominal beam separation values Δx and Δy are calculated from the LHC magnet currents at every scan step. An absolute calibration of the beam separation is made using the more precise VELO scale. In principle, each beam can have different calibration constants. However, since for all scans the beams were moved symmetrically (as opposed to one beam at a time), only the average (or common) length scale matters to first order for calibrating the separation. During the length scale calibration scans the beams were moved in several steps, with each measurement lasting from one to four minutes. The time intervals during which the beams were stationary are determined using the recorded states of the deflection magnets in the LHC logging data. Two methods are used in order to obtain the absolute length scale calibration. **Constant beam separation method.** Both beams are moved in five equidistant steps in Δx and Δy keeping their nominal separation constant. During a simultaneous parallel translation of both beams, the centre of the luminous region should follow the beam positions regardless of the bunch shapes. The luminous region centre can be determined using beam-beam vertices measured with the VELO.

Beam-beam vertices are selected using randomly triggered events in bb crossings. In the 2013 calibrations the average number of interactions per crossing was very low. Therefore, the standard trigger requirement for beam-beam events in bb crossings was used in order to collect a sufficient amount of data. To remove background from material interactions, only vertices that lie within 2mm from the median in each axis are retained. Potential beam-gas interaction background is reduced by applying a loose cut on the longitudinal position of the vertices |z| < 350mm and by requiring at least one track in each direction. The vertex coordinates are binned in *x* and *y*. The bin width is chosen automatically based on the data according to the Freedman–Diaconis rule.⁷ Data that deviates more than 5 σ from the median is discarded.

An empirical model is fitted to the histogram for each coordinate and the mean of the model distribution is used as an estimator for the position of the luminous region. A sum of two Gaussian functions, where the two means are not required to be equal, is found to fit the data well in all cases. The distribution of the luminous region is not expected to change during the scans if the beam separation is kept constant. Using this property as a constraint, an additional global fit per scan is performed, which has a single set of shape parameters with the exception of the mean, which is independent for each step.

A simultaneous drift of both beams in the same direction affects the length scale measurement. The drift is estimated using the steps before and after the scans, when the beams are nominally centred. Differences of the luminous region position before and after a scan are attributed entirely to simultaneous beam drift. A correction is applied for each step during the scan by using a linear interpolation. Since the beam separation is kept constant, beam-beam effects do not influence the measurement.

The measured luminous region position for each step is fitted against the average of the nominal beam positions using a weighted linear fit. The fit is performed simultaneously for all colliding bunch pairs using independent intercepts and a common slope. The results of this fit are shown in figure 43. The slope parameter of the fit is an estimate of the length scale calibration factor.

Beam-gas imaging method. Individual beam positions are measured using beam-gas interactions. The neon gas injection system was used to enhance the beam-gas interaction rate. The beams were moved similarly to VDM scans, but in a few large steps. Beam-gas vertices for each beam are selected using the corresponding standard trigger requirements for beam-empty and beam-beam crossings. In order to remove background from material interactions, only vertices that lie within 2 mm from the median in each axis are retained. Potential beam-beam interaction background is reduced by requiring no tracks in the opposite direction of the corresponding beam. Consistent comparison between measurements from colliding and non-colliding bunches is ensured by imposing a cut on the longitudinal position of the vertices |z| > 300 mm. To obtain the single-beam

⁷ The bin width is determined as two times the interquartile range (the difference between the 75^{th} and the 25^{th} percentiles) divided by the number of observations in the sample to the power of 1/3 [45].



Figure 43. Length scale calibration fits in the (left) x and (right) y axis in the July VDM session. The fitted intercepts are subtracted from both the fit curves and the data points as they have no significance. Each data point is an aggregate of the measurements from 16 colliding bunch pairs for the constant separation method and 35 individual bunches for the BGI method. The fit slope obtained using the constant separation method (solid line) should be compared with the average of the slopes obtained with the BGI method (dashed and dotted lines).

Table 9. Length scale calibration constants. All uncertainties are statistical.

Calibration	Method	$\left(\frac{\Delta x(\text{VELO})}{\Delta x(\text{LHC})} - 1\right) \times 10^2$	$\left(rac{\Delta y(\mathrm{VELO})}{\Delta y(\mathrm{LHC})}-1 ight) imes 10^2$
April 2012	Constant separation	-1.10 ± 0.02	-0.30 ± 0.02
July 2012	Constant separation	-0.14 ± 0.03	0.54 ± 0.04
July 2012	Beam-gas imaging	0.29 ± 0.06	0.69 ± 0.05

offsets, the x and y coordinates of the beam-gas vertices are fitted with a straight line as function of z. The weight assigned to the individual vertices takes into account the beam width and the vertex resolution.

Individual beam positions are measured before and after the scans, when the beams are nominally centred. No significant beam drifts are observed at the $0.5 \mu m$ level, thus no correction is made. The beam-beam deflection effect influences beam positions in bb bunch crossings. We take this into account as described in section 7.6, where Σ values are taken from the VDM scan measurements. Taking this effect into account improves the consistency between measurements from colliding and non-colliding bunches.

The measured beam position for each step is fitted against the nominal setting using a weighted linear fit. The fit is performed simultaneously for all bunches in each beam using one slope and one intercept parameter. The results of this fit are shown in figure 43. To obtain an effective common length scale, the average between both beams is taken.

The length scale calibration constants are summarized in table 9. A statistically significant discrepancy of about 0.5% is observed between the calibration factors for the *x* axis obtained using the two methods in July. The origin of this difference is not understood. Therefore, we assign 0.5% uncertainty to all length scale calibrations. It is assumed that this uncertainty is correlated between calibrations. The statistical uncertainties are small compared to the discrepancy and are neglected. The calibration constants used for July are those obtained by the constant separation method. Thus, we maintain consistency of treatment with other calibration sessions, for which we



Figure 44. Relative change of the beam-beam parameter as function of beam separation. The values are obtained using a simulation of the machine optic elements.

have not performed a beam-gas imaging calibration.

7.6 Beam-beam effects

The electromagnetic interaction between charged particles of two colliding bunches is called *beam-beam effect*. There are two aspects of this interaction that affect the VDM scan measurements. The first, called *dynamic* β *effect*, is the result of the mutual (de)focusing of the two colliding bunches. It leads to a change in β^* (and thus beam size), which depends on the transverse beam separation. Therefore, the transverse size of the bunches is not constant during the VDM scans. Secondly, the closed orbits of the bunches are distorted by the angular kick induced by their electromagnetic repulsion. This *beam-beam deflection* effect has a different magnitude depending on transverse separation, thus distorting the scan profiles.

The so-called *beam-beam parameter* that quantifies the strength of the dynamic β effect in the axis *m*, for a normally distributed bunch in beam *j* is given by [46]

$$\xi_{m,j} = \frac{\alpha \hbar c}{2\pi} \frac{N_{j'} Z_{j'} \beta_{m,j}^*}{E_j \sigma_{m,j'} (\sigma_{x,j'} + \sigma_{y,j'})} \xi_m^{\text{rel}} (\frac{\Delta x}{\sigma_{x,j}}, \frac{\Delta y}{\sigma_{y,j}}), \qquad (7.14)$$

where j' denotes the bunch in the opposite beam, $N_{j'}Z_{j'}$ is the bunch charge in units of elementary charge, $\beta_{m,j}^*$ is the value of the β function at the IP, E_j is the ring energy setting (particle energy divided by particle charge) and $\sigma_{m,j'}$ are the bunch sizes. The function ξ_m^{rel} was modelled [47] using the MAD-X optics software [48] and is shown in figure 44. It is independent of the machine optics parameters and the bunch properties. For small separations Δx and Δy the value of the ξ_m^{rel} is close to unity and approximately constant. Therefore, the beam-beam force is approximately linear, resembling the force of a quadrupole field. However, the force becomes non-linear for large separations. The above equation is also valid in the case of ion beams.

The ratio between the perturbed β^* and the unperturbed β_0^* as function of separation is given by

$$\frac{\beta_{m,j}^*}{\beta_{0,m,j}^*} = (1 + 4\pi\xi_{m,j}\cot(2\pi Q_{m,j}) - 4\pi^2\xi_{m,j}^2)^{-1/2},$$
(7.15)

where $Q_{m,j}$ is the machine tune of beam *j* in the axis *m*. While there is a recursive dependence between ξ and β^* , this is only a second order effect and can be neglected by using the nominal



Figure 45. Relative β^* change for beam 1 in the *x* axis as as function of beam separation for a scan pair in the (left) April and (right) July VDM sessions. The values for other scan pairs are similar. The values for beam 2 are similar owing to similar bunch intensities. The values for the *y* axis are also similar with *x* and *y* scans exchanged. The markers and the solid line represent the average, while the dotted lines correspond to the extremes among colliding bunches. The reference β^* is chosen such that the relative change is zero at the separation corresponding to maximum luminosity.

 β_0^* in eq. (7.14). Possible collisions at other IPs also contribute to the dynamic β independently from the local beam separation. Such contributions, provided beams are not moving at other IPs, are constant and effectively modify the nominal value of β^* . Since these modifications are much smaller than the uncertainty in the value of the nominal β^* itself, they are neglected.

To calculate the effect on the luminosity, the relative change of β^* is computed with respect to a reference value. Applying a correction to the rate (or the overlap integral) effectively removes the dependence of bunch shapes on separation. The choice of the reference β^* value is arbitrary and determines the values of the effective separation-independent bunch parameters. It can be shown that this choice has a negligible impact on the net correction. It is beneficial to use $\beta^*(0,0)$ as the reference, in order to enable consistent comparisons of VDM profile parameters from different scans. The relative β^* change (see figure 45) is calculated for each colliding bunch pair using the parameters of the nominal optics, the measured bunch intensities and the convolved widths. In this case, the shape of the bunches is approximated with a single Gaussian distribution and the size of the colliding bunches in each pair is assumed equal. The relative change of the luminosity is obtained using the perturbed bunch widths. Finally, the correction factor to the rate is obtained, which is shown in figure 46.

The beam-beam angular kick causes a deflection of beam j, which is calculated numerically using the formalism of ref. [49]

$$\theta_{y,j} + i \theta_{x,j} = -2\alpha \hbar c \frac{N_{j'} Z_{j'}}{E_j} F_0(\Delta x, \Delta y, \Sigma_x, \Sigma_y), \qquad (7.16)$$

where $N_{j'}Z_{j'}$ is the bunch charge in the opposite beam in units of elementary charge, *i* is the imaginary unit, F_0 is a complex function and Gaussian bunch profiles are assumed. The shift of the closed orbit in the *m* coordinate for beam *j* is given by

$$\delta_{m,j}^{\rm BB} = \theta_{m,j} \beta_{m,j}^* \frac{1}{2 \tan(\pi Q_{m,j})}, \qquad (7.17)$$



Figure 46. Rate correction factor for the dynamic beta effect for a scan pair in the (left) April and (right) July VDM sessions. The values for other scan pairs are similar. The markers and the solid line represent the average, while the dotted lines correspond to the extremes among colliding bunches. Owing to the dynamic β effect, the β 's and thus the beam sizes depend on the beam separation. To take this into account, the effect on the luminosity is compensated by multiplying the observed rates with the correction factors.

Figure 47. Correction to the nominal separation in the x axis for beam-beam-induced orbit shift for a scan pair in the (left) April and (right) July VDM sessions. The values for other scan pairs are similar. The values for the y axis are similar with x and y scans exchanged. The markers and the solid line represent the average, while the dotted lines correspond to the extremes among colliding bunches.

while the effect on the beam separation is the sum of the individual beam shifts

$$\delta^{\mathrm{BB}}\Delta m = \delta^{\mathrm{BB}}_{m,1} + \delta^{\mathrm{BB}}_{m,2}.$$
(7.18)

The corrections that are added to the nominal beam separation are shown in figure 47. The procedure outlined above was verified by the direct observation of beam-beam deflections with LHC orbit data [50], which were found to agree well with the expectations.

The systematic uncertainty of the correction for beam-beam effects is estimated by numerically propagating the uncertainties of the input parameters. In addition to β^* and machine tune, uncertainties are assigned to the ratio of bunch widths and the convolved bunch width to account for the assumption of identical bunch shapes and Gaussian bunches, respectively. The assumed uncertainties on input parameters and their correlations are listed in table 10. The uncertainty is similar for the two 8 TeV *pp* calibrations and amounts to about 0.3%.

Table 10.	Uncertainties	of input	parameters	to the	correction	for beam	-beam	effects.	All	quantities	are
correlated b	etween axes. 7	The quant	tities $\beta_{m,i}^*$ and	nd $Q_{m,j}$	are not co	rrelated b	etween	the two	bear	ms.	

Source	$\beta^*_{m,j}$	$Q_{m,j}$	$\sigma_{m,1}/\sigma_{m,2}$	Σ_m
Uncertainty	20%	0.02	50%	$\sim 10\%$

7.7 Fit model uncertainty

The choice of a particular model to describe the single beam shapes gives rise to a systematic uncertainty. The bias in the case of the BGI method is studied by simulating various deviations of the single-beam distributions from the assumed model as described in section 6.6. We assume that the same uncertainty of 0.5% applies to the VDM calibration and that it is fully correlated between the methods. In addition, the fitting procedure applied to VDM scan data might introduce a bias. The potential fit bias is estimated to be less than 0.2% by applying the analysis to simulated VDM scans with experimental conditions and beam properties similar to those of data.

The linear correlations in VDM scan profiles can be non-zero due to a tilt of the beam crossing plane or due to non-zero linear correlations with respect to the scanning axes of the transverse bunch distributions in eq. (7.4). The latter are found to be small and are neglected. The bunch crossing plane is nominally horizontal in all calibration fills, except in April, when the tilt was approximately -21.5° . Taking into account the correlation, the April calibration result changes by about +0.3%. The effect of the correlation is most pronounced if the scan pair crossing point is not centred at zero, as seen from the exponent factor in eq. (7.3). For the nominal scans, the correction from the exponent factor is negligible. However, the correction to the fourth (offset) scan pair in April amounts to -8%, improving significantly the consistency with the reference scans. Rather than propagating the uncertainty of the fixed parameter C_{xy} to the fit result, half of the effect of the correction is used as an estimate of the systematic uncertainty.

The major uncertainty of the VDM calibration is due to the assumptions made on the parameter values as described in section 7.3. In order to estimate the corresponding bias, the fits are redone without fixing parameters but using constraints obtained from BGI measurements. The information that is used from the BGI is limited to the factorizability of the beams, thus the VDM calibration is as independent as possible. In particular, constraints are not applied directly using dimensional parameters (widths) from BGI measurements, which are subject to resolution and alignment uncertainties. Moreover, for each scan pair, all bunch pair fits are subject to the same constraints, with their width taking into account the spread of the parameter values among bunch pairs.

First of all, Gaussian constraints on the factorizability parameters $f_{1,2}$ are added to the fit. However, these constraints are not sufficient to ensure that the beams are non-factorizable. This is easily seen by considering sets of degenerate parameters for which the beam distribution is Gaussian in at least one of the axes. Therefore, a generic measure of the transverse factorizability is introduced as

$$v_j = 1 - \frac{\int \rho_j(x, y = 0) dx \int \rho_j(x = 0, y) dy}{\rho_j(x = 0, y = 0)},$$
(7.19)

where ρ_j is the transverse beam distribution of beam j = 1, 2. The measure v_j is zero if ρ_j is factorizable. The values of $f_{1,2}$ and $v_{1,2}$ measured with BGI, as well as the predicted values at the time of the VDM scans are shown in figure 48. The values of $v_{1,2}$, being functions of the individual

Figure 48. Values of the (top) factorizability parameter f_j and (bottom) generic factorizability measure v_j in the (left) April and (right) July VDM sessions. The filled markers and the corresponding error bars indicate the average and the RMS of BGI measurements for individual bunch pairs. In the case of the July scans, BGI measurements are available before and after the VDM session, thus the values are linearly interpolated to obtain predictions at the time of VDM scan pairs (open markers). For April, as BGI data are only available after the VDM scans, the closest data points are used as a prediction, while the difference to the linear extrapolation is added in quadrature to the uncertainties. The trend of the generic factorizability measure shows that beam distributions become more factorizable as function of time.

beam distributions, cannot be computed using only the parameters describing the overlap integral from eq. (7.11). Therefore, two additional parameters for each coordinate m = x, y are added to the overlap fit model, which are sufficient to compute $v_{1,2}$:

$$C_m = (\sigma_{z1}^2 + \sigma_{z2}^2) \tan^2 \phi_m$$
 (7.20)

$$a_m = \frac{\sigma_{m1,n}^2}{\sigma_{m1,n}^2 + \sigma_{m2,n}^2} = \frac{\sigma_{m1,n}^2}{\Sigma_{m,nn}^2 - C_m}, \quad a_m \in [0,1].$$
(7.21)

Gaussian constraints of 10% around the measured value from BGI are applied to $C_{x,y}$. Additionally, to avoid unphysical sets of beam parameters, beams are required to have similar transverse sizes in each coordinate. This requirement is ensured by imposing a weak Gaussian constraint of 0 ± 0.1 on the asymmetry of the RMS of the two beam distributions. Such constraints are justified by the equal design β^* and emittance of the two beams. The asymmetries estimated from the BGI measurements for all bunch pairs lie within $\pm 1\sigma$, indicating that the width of the constraints is sufficiently large.

The VDM data for each bunch and scan pair are fitted using the additional parameters and the constraints described above. Moreover, profiles of χ^2/ndf are obtained as function of σ_{vis} . The


Figure 49. Examples of the constrained fit χ^2 /ndf profiles as function of σ_{vis} for (left) scan pair 1 in April and (right) scan pair 2 in July. Each curve corresponds to one bunch pair. The values of σ_{vis} obtained with direct fitting are shown with cross markers.

minima of the χ^2 /ndf profiles are very close to the values of σ_{vis} obtained with the fit where σ_{vis} is a free parameter as seen in figure 49. The new values of σ_{vis} are averaged for each scan pair and the obtained value is compared to the baseline cross-section for that pair. A difference of 0.6% and 0.9% is found for the April and the July calibration, respectively. The difference is attributed to the fact that $f_{1,2}$ and $A_{x,y}$ are fixed for the baseline fits and is assigned as a systematic uncertainty.

Another systematic uncertainty arises from the uncertainty in the central values of the constraints obtained from BGI analysis. To estimate this uncertainty, the constraints on $f_{1,2}$ and $v_{1,2}$ are varied by one standard deviation. The cross-section obtained with the modified constraints is compared to that obtained with the nominal constraints. The difference of 0.3% (for both April and July calibrations) is taken as an additional systematic uncertainty, which is considered fully correlated with the BGI result. The average χ^2 /ndf profiles for the reference scan pairs are shown in figure 50 together with the relative contributions of the VDM data and the constraints.

7.8 Reproducibility

An important source of non-reproducibility of VDM scan measurements is the drift of the beam orbits. In addition, in the presence of a non-zero crossing angle, a drift in the longitudinal bunch crossing point, z_{rf} , can influence the beam separation. While the effect of the latter is found to be negligible, the former has potentially a sizeable effect, which cannot be reliably corrected. For the following discussion data are averaged over bunch pairs since these effects are common for all bunches.

The relation between the position of the luminous region and the beam separation from eq. (5.7) can be used to estimate orbit drifts and z_{rf} values in the case of non-zero crossing angle. In the general case of a crossing angle in both xz and yz planes, the luminous region centre for Gaussian beams is

$$\xi_{lz} = z_{\rm rf} + k\Delta x', \qquad (7.22)$$

where k contains the beam widths and angles and $\Delta x'$ is the separation in the crossing plane, which is given by

$$\Delta x' = \Delta x \cos \psi + \Delta y \sin \psi, \quad \tan \psi = \frac{\tan \phi_y}{\tan \phi_x}.$$
 (7.23)



Figure 50. Average χ^2/ndf profiles as function of the relative difference to the baseline cross-section for the reference scan pairs in (top) April and (bottom) July shown with the top pair of lines (blue). The profiles corresponding to the nominal and modified constraints are shown with solid and dashed lines, respectively. The minima of the χ^2/ndf profiles are marked with a cross. The contribution of the VDM data, χ^2_{VDM} , normalized to the number of points in the VDM scan pair n_{VDM} is shown with the middle pair of lines (red). The contribution of the constraints, χ^2_c , normalized to the number of constraints n_c is shown with the bottom pair of lines (yellow).



Figure 51. Values of z_{rf} in (left) April and (right) July scans. The implied assumptions are that z_{rf} does not change and there is no orbit drift during a scan pair. No measurements are performed for offset scans. The deviations of the measurements from the weighted average (horizontal dotted line) are not statistically significant.

Here ψ is the tilt of the crossing plane, which is approximately -21.5° in April 2012 and negligible for all other calibrations.



Figure 52. Estimated "slow" drift of the beam separation in the (left) April and (right) July VDM sessions. The nominal separation where the luminosity is maximal gives a single estimate per scan of the drift in the corresponding direction (circles). The uncertainties are smaller than the marker size. The data for each coordinate are fitted with a smooth function (curve) and a prediction is made at each step of every scan (solid curve).

In accordance with eq. (5.7), the value of z_{rf} is measured by interpolating the longitudinal position of the luminous region ξ_{lz} at the separation that corresponds to the maximum luminosity. It is implicitly assumed that z_{rf} does not change and that there is no orbit drift during a scan pair. When the scanning axis is orthogonal to the beam crossing plane (*e.g.* y scans in July) this measurement cannot be made, as ξ_{lz} is not expected to change during the corresponding scan. The measured z_{rf} values, which are shown in figure 51, are found to be consistent between scans and with those obtained with the BGI analysis in the same fills. Therefore, it is assumed that z_{rf} values do not vary during a scan and the associated uncertainty is negligible.

The overall drift of the beam separation between scan pairs is automatically taken into account in the VDM analysis, since it is effectively only a shift of the nominal separation where the luminosity is maximal. However, the drift during a scan pair can introduce a bias to the measurement. It is useful to split the drift into two components: a "slow" one that corresponds to the time scale of a scan pair and a "fast" one that corresponds to the time scale of a step.

The slow beam separation drift can be estimated from the fitted position of the VDM profile maximum. One estimate per coordinate is obtained for each scan pair, as shown in figure 52.

The fast component of the beam separation drift is more difficult to estimate. The presence of a crossing angle enables an estimation of the drift in the beam separation in the crossing plane $\Delta x'$. This is possible because of the correlation between the $\Delta x'$ and the *z* position of the luminous region as seen in eq. (5.7). While the latter is strictly true only for Gaussian beams, the exact form of the function $\xi_{lz}(\Delta x')$ can be estimated by fitting the measured ξ_{lz} as function of $\Delta x'$ with a smooth function. The data and the estimated dependence are shown in figure 53. The deviations of the data points ξ_{lz} from the curve give an estimate of the beam drift. This approach is only reliable when the reference curve is obtained by averaging enough measurements. Therefore, the drift is estimated only for the three nominally head-on steps for each scan, as shown in figure 54.



Figure 53. Longitudinal position of the luminous region as function of separation in the crossing plane for reference scans in the (left) April and (right) July VDM sessions. It is assumed that there is no significant orbit and z_{rf} drift during the reference scans. The data for each scan direction are fitted with a smooth function. In July the crossing plane is orthogonal to the *y* axis, thus ξ_{lz} is not expected to change during the *y* scans and the data are omitted. The non-linearity of the curves is due to second order effects, which are not expected for pure Gaussian beams.



Figure 54. Estimated drift of the beam separation in the crossing plane $\Delta x'$ in the (left) April and (right) July VDM sessions. The drift is estimated only for each nominally head-on step.



Figure 55. Two example simulations of the beam separation drift in the *x* coordinate for the July VDM session. The separation drift is modelled with a Brownian motion that is constrained to the measured slow component of the drift (smooth black line) at the measurement points (crosses).



Figure 56. Cross-section bias caused by beam orbit drifts in the (left) April and (right) July VDM sessions. The baseline cross-section for each pair is shown with a solid circle and the average calibration cross-section (using the reference pairs) is shown with a horizontal dashed line. The average biased cross-section obtained with simulation of the beam drifts is shown with open circles. The error bars indicate the RMS of the biased cross-section of 400 independent simulations. The larger error bars for the offset scan pairs indicate that they are more sensitive to random drifts.

To estimate the effect of the orbit drifts on the cross-section measurement a simulation approach is used. The fast component of the drift in each coordinate is modelled with a probabilistic process according to a Brownian motion. The diffusion coefficient, which is the parameter of the model, is estimated from the measured fast component of the drift in $\Delta x'$. The value is found to be $0.005 \,\mu\text{m}^2\text{s}^{-1}$ and it is assumed to be equal for both coordinates and constant during a fill. The separation drift model is constrained to the measured slow component of the drift at the measurement points. Examples of the result of the simulation are shown in figure 55. The systematic uncertainty is evaluated by analysing 400 statistically independent simulations for each scan session, where the simulated drift is added to the nominal beam separation. The average bias on the cross-section and its RMS are summarized on figure 56. It is seen that the average bias, which is mainly due to the slow component of the drift, is small. On the other hand, the uncertainty in the bias, which is represented by the RMS of all simulations and is driven by the fast drift component, is sizeable.

Since the fast drifts are modelled to be independent for each scan pair, the uncertainty on the average cross-section is reduced. The uncertainty due to the slow component is assumed to be correlated, thus the estimates for individual scan pairs are averaged. The two uncertainties are added in quadrature to obtain a total uncertainty of 0.54% and 0.21% in April and July, respectively. The non-reproducibility can be estimated directly by the deviations from the average of the cross-section from individual scan pairs. The maximum deviations for reference scan pairs observed in April and July are 0.03% and 0.30%, respectively. Assuming that the deviations are mainly due to the drift, and in order to avoid double counting, only the larger value of the drift estimate and the maximum deviation from the average is taken as a systematic uncertainty.

It is seen in figure 56 that the scan pairs with an offset working point are more sensitive to random beam drifts, which can be explained by the large derivative of the VDM profile at non-zero beam separation. This is the main reason for excluding offset scans from the cross-section measurement. Moreover, since scan pair number five in July had a working point offset by about $30 \mu m$ in *x*, it is also more sensitive to beam drifts. Therefore, it is not included in the determination

ſ	
the baseline	2.
(in %). To o	obtain the effects, each correction is excluded, the fits are redone and the result is compared with
Table II. E	Effect of corrections on the result of the VDM scan calibrations at $\sqrt{s_{NN}}$ values of 8, 7 and 5 TeV

	8 TeV		7 TeV	5T	leV
Source	April	July		p Pb	Pb <i>p</i>
Ghost charge	+0.93	+0.68	+1.59	+0.80	+0.97
Satellites charge	+0.85	+0.43	+0.78	+0.19	+0.19
Efficiency of the observable	+0.32	-0.03	-0.04	+0.08	+0.11
Length scale	-1.39	+0.39	+0.06	-2.14	-1.03
Dynamic β	-0.39	-0.37	-0.35	-0.05	-0.05
Beam-beam deflection	+1.12	+1.12	+1.00	+0.15	+0.16
Linear correlation	+0.30	n.a.	n.a.	n.a.	n.a.

of the central value of the cross-section. Considering the reference scan pairs (one to four) in July, the uncertainties of the drift bias are too large compared to the observed fluctuations. This suggests that the value used for the simulation parameter, on which these uncertainties depend directly, is overestimated or the employed model does not describe well the beam drifts. Therefore, the higher cross-section measured from scan pair number five (see figure 40) cannot be explained by random beam drifts and the deviation from the average cross-section (using pairs one to four) of 0.8% is taken as a systematic uncertainty to account for a potential unknown source of non-reproducibility. The latter systematic uncertainty is assigned to all *pp* calibrations and is considered to be correlated among calibrations.

The effects of the applied corrections are summarized in table 11. The values for the other energies will be discussed below (section 7.10).

7.9 Results

The reference cross-section for pp collisions at $\sqrt{s} = 8$ TeV for the *Track* observable is determined by computing a weighted average of the results from the calibrations in the April and July fills. It is assumed that most of the systematic uncertainties are fully correlated in order to avoid underestimating the uncertainty on the combined result. The individual calibration results and the average reference cross-section are shown in table 12. A list of all uncertainties for the July and April calibrations is provided in table 13 along with estimates of their correlation. The values for the other energies are discussed below.

7.10 Summary of other van der Meer scan calibrations

The VDM analysis presented in this paper focuses on the calibration of the reference cross-section at $\sqrt{s} = 8 \text{ TeV} (pp)$. Comparable results are obtained with the VDM method for the 2011 pp calibration at $\sqrt{s} = 7 \text{ TeV}$ and the 2013 pPb and Pbp calibrations at $\sqrt{s_{NN}} = 5 \text{ TeV}$.

A VDM scan session was performed in fill 2234 for the calibration of the reference crosssection at $\sqrt{s} = 7 \text{ TeV} (pp)$. The operational procedure and the relevant trigger configuration were very similar to the one employed in 2012. The fill conditions and the scan parameters are listed in tables 1 and 8. Three symmetric x-y scan pairs were performed. The average decay time of the

Table 12. Individual ca	dibration results and average of the 8 IeV pp VDM scan sessions.	the part of the
relative uncertainty that	is correlated between the calibrations is shown in the fourth column	. The weights
used to obtain the average	ge are given in the last column.	

Session	$\sigma_{\rm T}$ (mb)	Relative	Correlated	Weight
56331011	Ofrack (IIIO)	uncertainty	part	weight
April 2012	60.70 ± 0.92	1.52%	1.34%	0.53
July 2012	60.54 ± 0.93	1.54%	1.51%	0.47
Average	60.62 ± 0.89	1.47%		

bunch population product amounts to 70 hours and the luminosity drop caused by emittance growth is negligible.

The VDM analysis at 7 TeV is performed in exactly the same manner as for the 8 TeV calibrations, which are discussed in detail. Given the similarity of the conditions, the systematic uncertainties on the fit model, the non-reproducibility and the length scale calibration are directly translated from the 2012 calibration. The corrections and the systematic uncertainty due to beambeam effects are found to be similar, owing to the similar beam energy and bunch intensities. The dominating systematic uncertainties are related to the constraints on the beam factorizability as obtained from the BGI analysis (section 6.7). By using these constraints instead of making parameter assumptions (see section 7.3), the measurement is shifted by 0.9%. The systematic uncertainty arising from the uncertainty in the central values of the constraints is found to be 0.8% and it is fully correlated with the BGI result.

Individual calibrations of the *p*Pb and Pb*p* reference cross-sections were performed in separate fills (3505, 3542). Due to the limited time of the ion runs, normal fills for physics data taking were used. Therefore, the filling schemes (see table 1) were not optimized as for dedicated luminosity calibration fills. The number of colliding bunches was 38, instead of usually 16 in dedicated fills. Moreover, the peak μ_{vis} value was very low, < 0.02. In order to decrease statistical uncertainties, the random trigger rate was doubled to 45 kHz during the *p*Pb calibration. The statistical uncertainties on the cross-section measurement are significantly larger compared to *pp* calibrations and amount to 0.25%.

The statistical uncertainty on the cross-section measurement per colliding bunch pair and per scan pair is about 2%. Therefore, a systematic structure in the fit residuals, which leads to a measurement bias of that order, may not be revealed by the χ^2 /ndf values of individual fits. The fit quality of a model is estimated from the sum of the residuals of all bunch pairs as function of beam separation. The aggregated residuals show a statistically significant structure if a pure Gaussian model is employed. It is found that two factorizable empirical models (a double Gaussian function with a negative weight for the narrow component and a Gaussian function multiplied by an even fourth order polynomial) fit the VDM scan profiles well and give similar cross-section values. In addition, two non-factorizable empirical models for the two-dimensional luminosity profiles have similar fit quality. The first non-factorizable model is a sum of two two-dimensional Gaussian functions with a negative weight for the narrow term. The second non-factorizable model is a *xy* rotationally symmetric Gaussian function multiplied by an even fourth order polynomial for the narrow term. The second non-factorizable model is a *xy* rotationally symmetric Gaussian function multiplied by an even fourth order polynomial for the narrow term. The second non-factorizable model is a *xy* rotationally symmetric Gaussian function multiplied by an even fourth order polynomial and scaled by the *x* and *y* profile widths. The two non-factorizable models give similar cross-section values,

Table 13. Relative systematic uncertainties on the reference cross-section for the VDM calibrations at $\sqrt{s_{NN}}$ values of 8, 7 and 5 TeV (in %). The uncertainties are divided into groups affecting the description of the VDM profile (sections 7.5–7.8), the measurement of the rate (section 7.4) and the bunch population product measurement (section 3). The fourth (eight) column indicates whether the uncertainties are correlated between the two pp calibrations at $\sqrt{s} = 8$ TeV (the pPb and Pbp calibrations). Empty cells indicate that the corresponding source is not applicable or the uncertainty is negligible. For the pPb and Pbp calibrations, the uncertainties due to scan variation and drift, and non-reproducibility cannot be estimated separately as only two VDM scan pairs per calibration were performed.

		8 TeV		7 TeV		5 TeV	
Source	April	July	Corr.		pPb	Pb <i>p</i>	Corr.
VDM profile description							
Fit model	0.50	0.50	yes	0.50	1.31	1.34	yes
Fit bias	0.20	0.20	yes	0.20			
Linear correlation	0.15		no				
Parameter assumptions	0.60	0.90	yes	0.90			
Constraints from BGI	0.30	0.30	yes	0.80			
Length scale	0.50	0.50	yes	0.50	1.10	1.10	no
VELO transverse scale	0.05	0.05	yes	0.05	0.05	0.05	yes
Beam-beam effects	0.27	0.30	yes	0.29	0.06	0.05	yes
Scan variation and drift	0.54	0.30	no	0.20	0.67	1.31	no
Non-reproducibility	0.80	0.80	yes	0.80			
(fifth scan pair in July)							
Statistical	0.06	0.05	no	0.11	0.26	0.25	no
	Rate n	neasure	ment				
Beam-gas background	0.03	0.07	yes	0.08	0.34	0.21	yes
Beam-beam background	0.09	0.18	yes	0.09	0.65	0.71	yes
Efficiency of rate observable	0.13	0.07	yes	0.17	0.04	0.06	yes
Bunch population uncertainties							
DCCT population product	0.22	0.24	yes	0.24	0.31	0.34	no
FBCT offset	0.04	0.06	yes	0.01	0.21	0.09	no
BPTX cross-check	0.13	0.06	yes		0.14	0.14	no
Ghost charge	0.05	0.03	yes	0.07	0.14	0.19	no
Satellites charge	0.02	0.03	yes	0.25	0.07	0.09	no
No satellite measurements	0.43		no				
Total	1.52	1.54		1.71	2.05	2.36	

which are about 2% lower than the values from the factorizable models. The VDM scans performed provide no information on the factorizability. Therefore, the full range of obtained cross-section values is considered by taking the central value at the middle and assigning half of the span as a fit model uncertainty.

An independent length scale calibration was performed for both fills. The LHC optics setup was almost identical in the two proton-lead beam configurations, thus the calibration constants

are expected to be identical. However, a difference of about 1% is observed and is assigned as a systematic uncertainty to both calibrations.

The uncertainty due to the beam orbit drift is estimated to be 0.67% and 0.88% for the *p*Pb and the Pbp calibration, respectively. The deviation of the visible cross-section measurement from the average in repeated VDM scans amounts to 0.23% and 1.31% for the pPb and the Pbp calibration, respectively. Following the procedure from section 7.8, only the larger value of the uncertainty due to orbit drift and the deviation from the average is taken as a systematic uncertainty. Beam-beam effects have a small impact on the pPb and Pbp calibrations owing to the low bunch intensities of $(1.3-1.7) \times 10^{10}$ elementary charges. The low μ values effectively increase the fraction of beam-gas and beam-beam induced backgrounds. The corresponding uncertainties are evaluated as described in section 7.4 and amount to 0.3% and 0.7% for beam-gas and beam-beam background respectively. It is assumed that there is no correlation between the uncertainties of the pPb and Pbp calibrations when it is partial. This is done to avoid underestimating the uncertainty on the ratio of the luminosities of the two data samples, which enters into ratios of cross-section measurements. A summary of all uncertainties is provided in table 13. The reference cross-section for the Track observable is 2.126 ± 0.049 b and 2.120 ± 0.053 b for the *p*Pb and Pbp mode, respectively. The compatibility of the two results indicates that the Track observable has similar efficiency for the two beam modes.

8 Summary and conclusion

Since some of the luminosity calibrations have been performed with both the VDM and BGI method (*pp* cross-section at $\sqrt{s} = 8$ TeV and $\sqrt{s} = 7$ TeV), the best result is obtained by computing an average of the two methods, taking into account the correlation between the systematic error sources. A summary of the final reference cross-section results is presented in table 15.

The BGI and VDM calibrations of the visible pp cross-section at $\sqrt{s} = 8$ TeV achieve very similar precision. Therefore, for simplicity the two results are combined with equal weights; the correlated components of the uncertainties are averaged linearly and the others are averaged in quadrature. The different sources of uncertainty are compared in table 14, indicating also whether they are correlated or not. The result of the combination is given in table 15. If the uncertainty on the propagation to physics data (0.31%, see section 4) is included, the total uncertainty on the luminosity is 1.16%. The latter uncertainty is valid if the complete 2012 data set or a major part of it is used. In some cases, for small partial sets, the uncertainty may be different.

A weighted average of the two pp cross-section measurements at $\sqrt{s} = 7$ TeV is computed, taking into account that the VDM calibration is more precise. The correlated part of the uncertainty due to the bunch population measurements and the factorizability is estimated to be 1%. A weight of 0.87 is given to the VDM measurement based on the uncorrelated part of the uncertainties. The final result given in table 15 has a precision of 1.63%. If the uncertainty on the propagation to physics data (0.53%, see section 4) is also included, the total uncertainty on the luminosity is 1.71%. As for the 2012 data set, the total uncertainty on the luminosity is valid if a major part of the 2011 data set is used.

Measurements of the visible cross-section for the *Vertex* observable as function of centre-ofmass energy are shown in figure 57. The visible cross-section for proton-lead collisions at 5 TeV

Table 14. Systematic uncertainties of the BGI and VDM methods for *pp* interactions at 8 TeV. The fourth column indicates whether the uncertainties are correlated between the two calibrations. All values are given in %.

Source	BGI	VDM	Correlated			
Bunch population uncertainties (section 3)						
FBCT offset	0.04	0.05	yes			
BPTX cross-check	n.a.	0.09	yes			
DCCT population product	0.22	0.23	yes			
Ghost charge	0.02	0.04	yes			
Satellite charge	0.06	0.02	yes			
Missing satellite measurements	n.a.	0.23	no			
Rate measure	ment					
Background subtraction	0.20	0.14	yes			
Ratio of observables <i>Track</i> to <i>Vertex</i>	0.20	n.a.	no			
Efficiency of rate observables	negl.	0.09	no			
Fit model	0.	.50	yes			
VELO transverse scale	0.	.05	yes			
BGI specific (se	ction 6)					
Beam-beam resolution	0.93		no			
Beam-gas resolution	0.55		no			
Detector alignment	0.45		no			
Measurement spread	0.54		no			
Bunch length	0.05		no			
Reconstruction efficiency	0.04		no			
Pressure gradient	0.03		no			
VDM specific (se	ection 7)				
Length scale		0.50	no			
Beam-beam effects		0.28	no			
Fit bias		0.20	no			
Linear correlation		0.08	no			
Parameter assumptions		0.74	no			
Constraints from BGI		0.30	yes			
Scan variation and drift		0.32	no			
Non-reproducibility		0.80	no			
Statistical		0.04	no			
Uncorrelated	1.31	1.32				
Correlated	0.59	0.65				

is scaled by $A^{-2/3}$ (A = 208 for lead). A comparison is made with the luminosity-independent measurements of the *pp* inelastic cross-section by the TOTEM collaboration [9, 52] at 7 and 8 TeV, and direct measurements by the ALICE [53] and the ATLAS [54] experiments. The measurements from other experiments are scaled with the LHCb efficiency for inelastic events $\eta_{\text{Vertex,LHCb}}$, which is obtained from simulation with a negligible statistical uncertainty. The values of $\eta_{\text{Vertex,LHCb}}$ are 0.729, 0.768 and 0.758 for 2.76, 7 and 8 TeV, respectively. A measurement from CMS [55] is not shown as it does not include an extrapolation to the total inelastic cross-section. No systematic uncertainties are included to account for the $\eta_{\text{Vertex,LHCb}}$ scaling, nor for the scaling with $A^{-2/3}$.

Figure 57 also shows the result of a first luminosity calibration of the pp cross-section at 2.76 TeV which was performed using the VDM method in 2011 [51]. During this data taking period, the VELO was positioned with its sensitive area at a minimum distance of 13 mm from the

Table 15. Results of the luminosity calibration measurements. The total uncertainty on the luminosity calibration (last column) is the sum in quadrature of the absolute calibration uncertainty (fourth column) and the relative calibration uncertainty (fifth column). The weights used to obtain the average absolute calibration at 8 and 7 TeV (pp) are given in the third column. The part of the uncertainty that is correlated between VDM and BGI calibrations is shown in parentheses (fourth column).

Mathad	Absolute calibration			Relative calibration	Total	
Method	$\sigma_{\rm vis}~({\rm mb})$	Weight	Uncertainty	uncertainty	uncertainty	
pp at \sqrt{s}	= 8 TeV					
BGI	60.62 ± 0.87	0.50	1.43% (0.59%)			
VDM	60.63 ± 0.89	0.50	1.47% (0.65%)			
Average	60.62 ± 0.68		1.12%	0.31%	1.16%	
pp at \sqrt{s}	$=7 \mathrm{TeV}$					
BGI	63.00 ± 2.22	0.13	3.52% (1.00%)			
VDM	60.01 ± 1.03	0.87	1.71% (1.00%)			
Average	60.40 ± 0.99		1.63%	0.53%	1.71%	
pp at \sqrt{s}	= 2.76 TeV					
BGI	$52.7 \hspace{0.2cm} \pm 1.2$		2.20%	0.25%	2.21%	
<i>p</i> Pb at $$	$\overline{s_{NN}} = 5 \mathrm{TeV}$					
VDM	2126 ± 49		2.05%	1.03%	2.29%	
Pbp at $\sqrt{s_{NN}} = 5$ TeV						
VDM	2120 ± 53		2.36%	0.82%	2.50%	

beam instead of the nominal 8 mm. The corresponding drop in efficiency is estimated to be 5% from simulation. Taking this difference into account, as well as the unaccounted potential non-factorizability of the beams, a good agreement is found when comparing to the more precise BGI measurement from 2013.

An earlier measurement of the pp visible cross-section based on the VDM and BGI methods for the *Track* observable at $\sqrt{s} = 7$ TeV was reported in ref. [11]. The data available in the older publication were not sufficient to measure the factorizability of the beams and complete factorizability was assumed. It was shown here that neglecting effects of non-factorizability may cause an underestimate of the visible cross-section at the few percent level. Nevertheless, the value (58.8 ± 2.0 mb) is consistent with the significantly more precise result reported here. The results reported here supersede those of ref. [11].

In conclusion, several luminosity calibration measurements were performed at the LHC using the LHCb detector and two experimental methods, the VDM scan method and the BGI method. Ghost charge fractions were also measured using beam-gas interactions and the results are used in several luminosity calibrations by other LHC experiments. The LHCb luminosity calibrations were made for proton-proton collisions at three different centre-of-mass energies (2.76, 7 and 8 TeV) and for proton-lead collisions at the equivalent nucleon-nucleon centre-of-mass energy of 5 TeV. The analysis strategies and the results of the calibrations were presented in detail. Compared to the calibration performed in 2010, an improvement by an order of magnitude was achieved in the bunch population normalization, chiefly obtained by means of a thorough study of the LHC beam current



Figure 57. Measurements of the visible cross-section for the *Vertex* observable as function of centre-of-mass energy. The measurement at 2.76 TeV [51] performed using the VDM method (solid triangle) is corrected for the reduced efficiency due to the VELO being not fully closed. The visible cross-section for proton-lead collisions at 5 TeV is scaled by $A^{-2/3}$. A comparison is made with the luminosity-independent measurements of the *pp* inelastic cross-section by the TOTEM collaboration [9, 52] at 7 and 8 TeV, with direct measurements by the ALICE [53] and the ATLAS [54] experiments, and with a measurement by ATLAS from elastic *pp* scattering [10]. The measurements from other experiments are scaled with the LHCb efficiency for inelastic events $\eta_{\text{Vertex,LHCb}}$, obtained from simulation. The uncertainties of the direct measurements from ALICE and ATLAS are dominated by the extrapolation of the visible cross-section to the total inelastic cross-section and are not to be compared with the uncertainties of the LHCb measurements. The tick marks represent the uncertainty due to the luminosity calibration only. Data points at the same centre-of-mass energy are displaced horizontally for clarity.

measuring devices. This achievement opened the way to a global reduction of the systematic and statistical uncertainties for both the VDM scan method and the BGI method. A controlled gas injection into the LHC vacuum was employed to increase the beam-gas interaction rate by almost two orders of magnitude and detailed systematic studies (including offset scans and reproducibility checks) were conducted with the VDM scan method. Modelling the non-factorizability in the transverse distribution of the bunch particles is required and, if neglected, would have changed the calibration results by up to 3%. In the case of proton-proton collisions at 8 TeV, a precision of 1.47% is obtained with the VDM scan method and 1.43% with the BGI method. When combining the results, the precision obtained on the reference visible cross-section is 1.12%, which constitutes to date the most precise luminosity calibration at a bunched-beam hadron collider. The precision of the calibrations for the other beams and beam energies is close to 2%. The luminosity calibration results are used to determine a reference cross-section, which is employed in the LHCb physics data analysis to measure absolute cross-sections of various processes.

Acknowledgments

We express our gratitude to our colleagues in the CERN accelerator departments for the excellent performance of the LHC. We thank the technical and administrative staff at the LHCb institutes. We acknowledge support from CERN and from the national agencies: CAPES, CNPq, FAPERJ and

FINEP (Brazil); NSFC (China); CNRS/IN2P3 (France); BMBF, DFG, HGF and MPG (Germany); SFI (Ireland); INFN (Italy); FOM and NWO (The Netherlands); MNiSW and NCN (Poland); MEN/IFA (Romania); MinES and FANO (Russia); MinECo (Spain); SNSF and SER (Switzerland); NASU (Ukraine); STFC (United Kingdom); NSF (USA). The Tier1 computing centres are supported by IN2P3 (France), KIT and BMBF (Germany), INFN (Italy), NWO and SURF (The Netherlands), PIC (Spain), GridPP (United Kingdom). We are indebted to the communities behind the multiple open source software packages on which we depend. We are also thankful for the computing resources and the access to software R&D tools provided by Yandex LLC (Russia). Individual groups or members have received support from EPLANET, Marie Skłodowska-Curie Actions and ERC (European Union), Conseil général de Haute-Savoie, Labex ENIGMASS and OCEVU, Région Auvergne (France), RFBR (Russia), XuntaGal and GENCAT (Spain), Royal Society and Royal Commission for the Exhibition of 1851 (United Kingdom).

References

- [1] LHCb collaboration, *LHCb reoptimized detector design and performance: Technical Design Report*, CERN-LHCC-2003-030, LHCB-TDR-009.
- [2] LHCb collaboration, The LHCb Detector at the LHC, 2008 JINST 3 S08005.
- [3] LHCb collaboration, *Inclusive W and Z production in the forward region at* $\sqrt{s} = 7$ TeV, JHEP **06** (2012) 058 [arXiv:1204.1620].
- [4] LHCb collaboration, *Measurement of the forward W boson cross-section in pp collisions at* $\sqrt{s} = 7 \text{ TeV}, \text{ arXiv: } 1408.4354.$
- [5] LHCb collaboration, Updated measurements of exclusive J/ψ and $\psi(2S)$ production cross-sections in pp collisions at $\sqrt{s} = 7$ TeV, J. Phys. **G 41** (2014) 055002 [arXiv:1401.3288].
- [6] C. Møller, General properties of the characteristic matrix in the theory of elementary particles, K. Danske Vidensk. Selsk. Mat. -Fys. Medd. 23 (1945) 1.
- [7] O. Napoly, The luminosity for beam distributions with error and wake field effects in linear colliders, Part. Accel. 40 (1993) 181 CERN-SL-92-34-AP, CERN-CLIC-NOTE-173.
- [8] W. Herr and B. Muratori, Concept of luminosity, in Proceedings of the CERN Accelerator School (CAS) (2003) 361, Zeuthen, Germany [CERN-2006-002.361].
- [9] TOTEM collaboration, G. Antchev et al., *Luminosity-Independent Measurement of the Proton-Proton Total Cross section at* $\sqrt{s} = 8$ *TeV*, *Phys. Rev. Lett.* **111** (2013) 012001.
- [10] ATLAS collaboration, Measurement of the total cross section from elastic scattering in pp collisions at $\sqrt{s} = 7$ TeV with the ATLAS detector, Nucl. Phys. **B 889** (2014) 486 [arXiv:1408.5778].
- [11] LHCb collaboration, Absolute luminosity measurements with the LHCb detector at the LHC, 2012 JINST 7 P01010 [arXiv:1110.2866].
- [12] S. van der Meer, Calibration of the effective beam height in the ISR, CERN-ISR-PO-68-31 (1968).
- [13] C. Rubbia, *Measurement of the luminosity of* $p\overline{p}$ *collider with a (generalized) Van der Meer Method*, CERN-p \overline{p} -Note-38 (1977).
- [14] V. Balagura, Notes on van der Meer Scan for Absolute Luminosity Measurement, Nucl. Instrum. Meth. A 654 (2011) 634 [arXiv:1103.1129].

- [15] H. Burkhardt and P. Grafström, Absolute luminosity from machine parameters, CERN-LHC-PROJECT-Report-1019 (2007).
- [16] M. Ferro-Luzzi, Proposal for an absolute luminosity determination in colliding beam experiments using vertex detection of beam-gas interactions, Nucl. Instrum. Meth. A 553 (2005) 388.
- [17] ATLAS collaboration, Improved luminosity determination in pp collisions at $\sqrt{s} = 7$ TeV using the ATLAS detector at the LHC, Eur. Phys. J. C 73 (2013) 2518 [arXiv:1302.4393].
- [18] CMS collaboration, CMS luminosity based on pixel cluster counting summer 2013 update, CMS-PAS-LUM-13-001.
- [19] ALICE collaboration, Measurement of visible cross sections in proton-lead collisions at $\sqrt{s_{NN}} = 5.02 \text{ TeV}$ in van der Meer scans with the ALICE detector, 2014 JINST 9 P11003 [arXiv:1405.1849].
- [20] L. Evans and P. Bryant, LHC Machine, 2008 JINST 3 S08001.
- [21] LHCb collaboration, Prompt K_s^0 production in pp collisions at $\sqrt{s} = 0.9$ TeV, Phys. Lett. B 693 (2010) 69 [arXiv:1008.3105].
- [22] V. Balagura, Luminosity measurement at LHCb, in Proceedings of 45th Rencontres de Moriond (QCD and High Energy Interactions), La Thuile, Aosta Valley Italy, 13–20 March 2010, [LHCb-TALK-2010-021].
- [23] P. Hopchev, The Beam-gas method for luminosity measurement at LHCb, arXiv:1005.4398.
- [24] G. Anders et al., LHC bunch current normalisation for the April-May 2010 luminosity calibration measurements, CERN-ATS-Note-2011-004 PERF (2011).
- [25] G. Alici et al., LHC bunch current normalisation for the October 2010 luminosity calibration measurements, CERN-ATS-Note-2011-016 PERF (2011).
- [26] C. Barschel et al., Results of the LHC DCCT calibration studies, CERN-ATS-Note-2012-026 PERF (2012).
- [27] G. Anders et al., *Study of the relative LHC bunch populations for luminosity calibration*, CERN-ATS-Note-2012-028 PERF (2012).
- [28] A. Alici et al., *Study of the LHC ghost charge and satellite bunches for luminosity calibration*, CERN-ATS-Note-2012-029 PERF (2012).
- [29] R. Aaij et al., Performance of the LHCb Vertex Locator, 2014 JINST 9 09007 [arXiv:1405.7808].
- [30] R. Aaij et al., *The LHCb Trigger and its Performance in 2011*, 2013 JINST 8 P04022 [arXiv:1211.3055].
- [31] M. Kucharczyk, P. Morawski and M. Witek, *Primary Vertex Reconstruction at LHCb*, LHCb-PUB-2014-044.
- [32] P. Odier, M. Ludwig and S. Thoulet, *The DCCT for the LHC Beam Intensity Measurement*, in *Proceedings of DIPAC09 (Beam Diagnostics and Instrumentation for Particle Accelerators)*, Basel, Switzerland, 25–27 May 2009 [CERN-BE-2009-019].
- [33] D. Belohrad et al., The LHC fast BCT system: a comparison of design parameters with initial performance, in Proceedings of the Beam Instrumentation Workshop, BIW'10, Santa Fe, New Mexico, United States Of America, 2–6 May 2010, [CERN-BE-2010-010].

- [34] P. Baudrenghien et al., The LHC RF system experience with beam operation, in Proceedings of the 2nd International Particle Accelerator Conference, San Sebastian, Spain, 4–9 September 2011 [CERN-ATS-2011-048, CERN-ATS-2011-039].
- [35] A. Jeff et al., *Longitudinal density monitor for the LHC*, *Phys. Rev. ST Accel. Beams* **15** (2012) 032803.
- [36] A. Jeff, *A longitudinal density monitor for the LHC*, PhD thesis, Liverpool University, 2012, CERN-THESIS-2012-240.
- [37] C. Barschel, Precision luminosity measurements at LHCb with beam-gas imaging, PhD thesis, RWTH Aachen University, 2014, CERN-THESIS-2013-301.
- [38] C. Ohm and T. Pauly, The ATLAS beam pick-up based timing system, Nucl. Instrum. Meth. A 623 (2010) 558 [arXiv:0905.3648].
- [39] P. Hopchev, Absolute luminosity measurements at LHCb, PhD thesis, Grenoble University, 2011, CERN-THESIS-2011-210.
- [40] A. Boccardi, E. Bravin, M. Ferro-Luzzi and S. Mazzoni, LHC luminosity calibration using the longitudinal density monitor, CERN-ATS-Note-2013-034 TECH (2013).
- [41] N.Y. Zaitsev, *Study of the LHCb pile-up trigger and* $B_S \rightarrow J/\psi\phi$ *decay*, PhD thesis, Amsterdam University, 2000, [CERN-THESIS-2000-043].
- [42] R. Alemany-Fernandez, F. Follin and R. Jacobsson, *The LHCb online luminosity control and monitoring*, in *Proceedings of the 4th International Particle Accelerator Conference*, IPAC13, 12–17 May 2013, Shanghai, China [CERN-ACC-2013-0028].
- [43] F.J. Decker, Beam distributions beyond RMS, in Proceedings of the Beam Instrumentation Workshop, Vancouver, Canada, 3–6 October 1994 [SLAC-PUB-6684].
- [44] T.J. Hastie and R.J. Tibshirani, Generalized Additive Models, Chapman and Hall (1990).
- [45] D. Freedman and P. Diaconis, On the histogram as a density estimator: L2 theory, Prob. Theor. Relat. Fields 57 (1981) 453.
- [46] W. Herr, Beam-beam interactions, in Proceedings of the CERN Accelerator School (CAS) (2003) 379, Zeuthen, Germany [CERN-2006-002.379].
- [47] W. Herr, Beam-beam effects and dynamic β*, in LHC Lumi Days: LHC Workshop on LHC Luminosity Calibration, 29 Feb – 1 Mar 2012, CERN, Geneva, Switzerland.
- [48] MAD Methodical Accelerator Design, see http://cern.ch/mad.
- [49] V. Ziemann, Beyond Bassetti and Erskine: Beam-beam deflections for non-Gaussian beams, in Proceedings of the Beam Dynamics Workshop, Los Angeles, USA, 13–16 May 1991 [SLAC-PUB-5582].
- [50] W. Kozanecki, T. Pieloni and J. Wenninger, Observation of beam-beam deflections with LHC orbit data, CERN-ACC-NOTE-2013-0006 (2013).
- [51] LHCb collaboration, Measurement of J/ψ production in pp collisions at $\sqrt{s} = 2.76$ TeV, JHEP 02 (2013) 041 [arXiv:1212.1045].
- [52] TOTEM collaboration, G. Antchev et al., *Luminosity-independent measurements of total, elastic and inelastic cross-sections at* $\sqrt{s} = 7$ TeV, Europhys. Lett. **101** (2013) 21004.

- [53] ALICE collaboration, Measurement of inelastic, single- and double-diffraction cross sections in proton-proton collisions at the LHC with ALICE, Eur. Phys. J. C 73 (2013) 2456 [arXiv:1208.4968].
- [54] ATLAS collaboration, Measurement of the Inelastic Proton-Proton Cross-Section at $\sqrt{s} = 7$ TeV with the ATLAS Detector, Nature Commun. 2 (2011) 463 [arXiv:1104.0326].
- [55] CMS collaboration, Measurement of the inelastic proton-proton cross section at $\sqrt{s} = 7$ TeV, Phys. Lett. **B** 722 (2013) 5 [arXiv:1210.6718].

The LHCb collaboration

R. Aaij⁴¹, B. Adeva³⁷, M. Adinolfi⁴⁶, A. Affolder⁵², Z. Ajaltouni⁵, S. Akar⁶, J. Albrecht⁹, F. Alessio³⁸, M. Alexander⁵¹, S. Ali⁴¹, G. Alkhazov³⁰, P. Alvarez Cartelle³⁷, A.A. Alves Jr^{25,38}, S. Amato², S. Amerio²², Y. Amhis⁷, L. An³, L. Anderlini^{17,g}, J. Anderson⁴⁰, R. Andreassen⁵⁷, M. Andreotti^{16,f}, J.E. Andrews⁵⁸, R.B. Appleby⁵⁴, O. Aquines Gutierrez¹⁰, F. Archilli³⁸, A. Artamonov³⁵, M. Artuso⁵⁹, E. Aslanides⁶, G. Auriemma^{25,n}, M. Baalouch⁵, S. Bachmann¹¹, J.J. Back⁴⁸, A. Badalov³⁶, C. Baesso⁶⁰, W. Baldini¹⁶, R.J. Barlow⁵⁴, C. Barschel³⁸, S. Barsuk⁷, W. Barter⁴⁷, V. Batozskaya²⁸, V. Battista³⁹, A. Bay³⁹, L. Beaucourt⁴, J. Beddow⁵¹, F. Bedeschi²³, I. Bediaga¹, S. Belogurov³¹, K. Belous³⁵, I. Belyaev³¹, E. Ben-Haim⁸, G. Bencivenni¹⁸, S. Benson³⁸, J. Benton⁴⁶, A. Berezhnoy³², R. Bernet⁴⁰, M.-O. Bettler⁴⁷, M. van Beuzekom⁴¹, A. Bien¹¹, S. Bifani⁴⁵, T. Bird⁵⁴, A. Bizzeti^{17,i}, P.M. Bjørnstad⁵⁴, T. Blake⁴⁸, F. Blanc³⁹, J. Blouw¹⁰, S. Blusk⁵⁹, V. Bocci²⁵, A. Bondar³⁴, N. Bondar^{30,38}, W. Bonivento^{15,38}, S. Borghi⁵⁴, A. Borgia⁵⁹, M. Borsato⁷, T.J.V. Bowcock⁵², E. Bowen⁴⁰, C. Bozzi¹⁶, T. Brambach⁹, J. Bressieux³⁹, D. Brett⁵⁴, M. Britsch¹⁰, T. Britton⁵⁹, J. Brodzicka⁵⁴, N.H. Brook⁴⁶, H. Brown⁵², A. Bursche⁴⁰, J. Buytaert³⁸, S. Cadeddu¹⁵, R. Calabrese^{16,f}, M. Calvi^{20,k}, M. Calvo Gomez^{36,p}, P. Campana^{18,38}, D. Campora Perez³⁸, A. Carbone^{14,d}, G. Carboni^{24,l}, R. Cardinale^{19,38,j}, A. Cardini¹⁵, L. Carson⁵⁰, K. Carvalho Akiba², G. Casse⁵², L. Cassina²⁰, L. Castillo Garcia³⁸, M. Cattaneo³⁸, Ch. Cauet⁹, R. Cenci²³, M. Charles⁸, Ph. Charpentier³⁸, M. Chefdeville⁴, S. Chen⁵⁴, S.-F. Cheung⁵⁵, N. Chiapolini⁴⁰, M. Chrzaszcz^{40,26}, K. Ciba³⁸, X. Cid Vidal³⁸, G. Ciezarek⁵³, P.E.L. Clarke⁵⁰, M. Clemencic³⁸, H.V. Cliff⁴⁷, J. Closier³⁸, V. Coco³⁸, J. Cogan⁶, E. Cogneras⁵, L. Cojocariu²⁹, G. Collazuol²², P. Collins³⁸, A. Comerma-Montells¹¹, A. Contu^{15,38}, A. Cook⁴⁶, M. Coombes⁴⁶, S. Coquereau⁸, G. Corti³⁸, M. Corvo^{16, f}, I. Counts⁵⁶, B. Couturier³⁸, G.A. Cowan⁵⁰, D.C. Craik⁴⁸, M. Cruz Torres⁶⁰, S. Cunliffe⁵³, R. Currie⁵³, C. D'Ambrosio³⁸, J. Dalseno⁴⁶, P. David⁸, P.N.Y. David⁴¹, A. Davis⁵⁷, K. De Bruyn⁴¹, S. De Capua⁵⁴, M. De Cian¹¹, J.M. De Miranda¹, L. De Paula², W. De Silva⁵⁷, P. De Simone¹⁸, C.-T. Dean⁵¹, D. Decamp⁴, M. Deckenhoff⁹, L. Del Buono⁸, N. Déléage⁴, D. Derkach⁵⁵, O. Deschamps⁵, F. Dettori³⁸, A. Di Canto³⁸, H. Dijkstra³⁸, S. Donleavy⁵², F. Dordei¹¹, M. Dorigo³⁹, A. Dosil Suárez³⁷, D. Dossett⁴⁸, A. Dovbnya⁴³, K. Dreimanis⁵², G. Dujany⁵⁴, F. Dupertuis³⁹, P. Durante³⁸, R. Dzhelyadin³⁵, A. Dziurda²⁶, A. Dzyuba³⁰, S. Easo^{49,38}, U. Egede⁵³, V. Egorychev³¹, S. Eidelman³⁴, S. Eisenhardt⁵⁰, U. Eitschberger⁹, R. Ekelhof⁹, L. Eklund⁵¹, I. El Rifai⁵, Ch. Elsasser⁴⁰, S. Ely⁵⁹, S. Esen¹¹, H.-M. Evans⁴⁷, T. Evans⁵⁵, A. Falabella¹⁴, C. Färber¹¹, C. Farinelli⁴¹, N. Farley⁴⁵, S. Farry⁵², RF Fay⁵², D. Ferguson⁵⁰, V. Fernandez Albor³⁷, F. Ferreira Rodrigues¹, M. Ferro-Luzzi³⁸, S. Filippov³³, M. Fiore^{16,f}, M. Fiorini^{16,f}, M. Firlej²⁷, C. Fitzpatrick³⁹, T. Fiutowski²⁷, P. Fol⁵³, M. Fontana¹⁰, F. Fontanelli^{19, j}, R. Forty³⁸, O. Francisco², M. Frank³⁸, C. Frei³⁸, M. Frosini^{17, g}, J. Fu^{21,38}, E. Furfaro^{24, l} A. Gallas Torreira³⁷, D. Galli^{14,d}, S. Gallorini^{22,38}, S. Gambetta^{19,j}, M. Gandelman², P. Gandini⁵⁹, Y. Gao³, J. García Pardiñas³⁷, J. Garofoli⁵⁹, J. Garra Tico⁴⁷, L. Garrido³⁶, D. Gascon³⁶, C. Gaspar³⁸, R. Gauld⁵⁵, L. Gavardi⁹, A. Geraci^{21,v}, E. Gersabeck¹¹, M. Gersabeck⁵⁴, T. Gershon⁴⁸, Ph. Ghez⁴, A. Gianelle²², S. Gianì³⁹, V. Gibson⁴⁷, L. Giubega²⁹, V.V. Gligorov³⁸, C. Göbel⁶⁰, D. Golubkov³¹, A. Golutvin^{53,31,38}, A. Gomes^{1,a}, C. Gotti²⁰, M. Grabalosa Gándara⁵, R. Graciani Diaz³⁶, L.A. Granado Cardoso³⁸, E. Graugés³⁶, G. Graziani¹⁷, A. Grecu²⁹, E. Greening⁵⁵, S. Gregson⁴⁷, P. Griffith⁴⁵, L. Grillo¹¹, O. Grünberg⁶², B. Gui⁵⁹, E. Gushchin³³, Yu. Guz^{35,38}, T. Gys³⁸, C. Hadjivasiliou⁵⁹, G. Haefeli³⁹, C. Haen³⁸, S.C. Haines⁴⁷, S. Hall⁵³, B. Hamilton⁵⁸, T. Hampson⁴⁶, X. Han¹¹, S. Hansmann-Menzemer¹¹, N. Harnew⁵⁵, S.T. Harnew⁴⁶, J. Harrison⁵⁴, J. He³⁸, T. Head³⁸, V. Heijne⁴¹, K. Hennessy⁵², P. Henrard⁵, L. Henry⁸, J.A. Hernando Morata³⁷, E. van Herwijnen³⁸, M. Heß⁶², A. Hicheur¹, D. Hill⁵⁵, M. Hoballah⁵, C. Hombach⁵⁴, W. Hulsbergen⁴¹, P. Hunt⁵⁵, N. Hussain⁵⁵, D. Hutchcroft⁵², D. Hynds⁵¹, M. Idzik²⁷, P. Ilten⁵⁶, R. Jacobsson³⁸, A. Jaeger¹¹, J. Jalocha⁵⁵, E. Jans⁴¹, P. Jaton³⁹, A. Jawahery⁵⁸, F. Jing³, M. John⁵⁵, D. Johnson³⁸, C.R. Jones⁴⁷, C. Joram³⁸, B. Jost³⁸, N. Jurik⁵⁹, S. Kandybei⁴³, W. Kanso⁶, M. Karacson³⁸, T.M. Karbach³⁸, S. Karodia⁵¹, M. Kelsey⁵⁹, I.R. Kenyon⁴⁵, T. Ketel⁴², B. Khanji²⁰, C. Khurewathanakul³⁹, S. Klaver⁵⁴, K. Klimaszewski²⁸, O. Kochebina⁷, M. Kolpin¹¹, I. Komarov³⁹,

R.F. Koopman⁴², P. Koppenburg^{41,38}, M. Korolev³², A. Kozlinskiy⁴¹, L. Kravchuk³³, K. Kreplin¹¹, M. Kreps⁴⁸, G. Krocker¹¹, P. Krokovny³⁴, F. Kruse⁹, W. Kucewicz^{26,o}, M. Kucharczyk^{20,26,k}, V. Kudryavtsev³⁴, K. Kurek²⁸, T. Kvaratskheliya³¹, V.N. La Thi³⁹, D. Lacarrere³⁸, G. Lafferty⁵⁴, A. Lai¹⁵, D. Lambert⁵⁰, R.W. Lambert⁴², G. Lanfranchi¹⁸, C. Langenbruch⁴⁸, B. Langhans³⁸, T. Latham⁴⁸, C. Lazzeroni⁴⁵, R. Le Gac⁶, J. van Leerdam⁴¹, J.-P. Lees⁴, R. Lefèvre⁵, A. Leflat³², J. Lefrançois⁷, S. Leo²³, O. Leroy⁶, T. Lesiak²⁶, B. Leverington¹¹, Y. Li³, T. Likhomanenko⁶³, M. Liles⁵², R. Lindner³⁸, C. Linn³⁸, F. Lionetto⁴⁰, B. Liu¹⁵, S. Lohn³⁸, I. Longstaff⁵¹, J.H. Lopes², N. Lopez-March³⁹, P. Lowdon⁴⁰, H. Lu³, D. Lucchesi^{22,r}, H. Luo⁵⁰, A. Lupato²², E. Luppi^{16,f}, O. Lupton⁵⁵, F. Machefert⁷, I.V. Machikhiliyan³¹, F. Maciuc²⁹, O. Maev³⁰, S. Malde⁵⁵, A. Malinin⁶³, G. Manca^{15,e}, G. Mancinelli⁶, A. Mapelli³⁸, J. Maratas⁵, J.F. Marchand⁴, U. Marconi¹⁴, C. Marin Benito³⁶, P. Marino^{23,t}, R. Märki³⁹, J. Marks¹¹, G. Martellotti²⁵, A. Martens⁸, A. Martín Sánchez⁷, M. Martinelli³⁹, D. Martinez Santos^{42,38}, F. Martinez Vidal⁶⁴, D. Martins Tostes², A. Massafferri¹, R. Matev^{38,w}, Z. Mathe³⁸, C. Matteuzzi²⁰, B. Maurin³⁹, A. Mazurov⁴⁵, M. McCann⁵³, J. McCarthy⁴⁵, A. McNab⁵⁴, R. McNulty¹², B. McSkelly⁵², B. Meadows⁵⁷, F. Meier⁹, M. Meissner¹¹, M. Merk⁴¹, D.A. Milanes⁸, M.-N. Minard⁴, N. Moggi¹⁴, J. Molina Rodriguez⁶⁰, S. Monteil⁵, M. Morandin²², P. Morawski²⁷, A. Mordà⁶, M.J. Morello^{23,t}, J. Moron²⁷, A.-B. Morris⁵⁰, R. Mountain⁵⁹, F. Muheim⁵⁰, K. Müller⁴⁰, M. Mussini¹⁴, B. Muster³⁹, P. Naik⁴⁶, T. Nakada³⁹, R. Nandakumar⁴⁹, I. Nasteva², M. Needham⁵⁰, N. Neri²¹, S. Neubert³⁸, N. Neufeld³⁸, M. Neuner¹¹, A.D. Nguyen³⁹, T.D. Nguyen³⁹, C. Nguyen-Mau^{39,q}, M. Nicol⁷, V. Niess⁵, R. Niet⁹, N. Nikitin³², T. Nikodem¹¹, A. Novoselov³⁵, D.P. O'Hanlon⁴⁸, A. Oblakowska-Mucha^{27,38}, V. Obraztsov³⁵, S. Oggero⁴¹, S. Ogilvy⁵¹, O. Okhrimenko⁴⁴, R. Oldeman^{15,e}, C.J.G. Onderwater⁶⁵, M. Orlandea²⁹, J.M. Otalora Goicochea², P. Owen⁵³, A. Oyanguren⁶⁴, B.K. Pal⁵⁹, A. Palano^{13,c}, F. Palombo^{21,u}, M. Palutan¹⁸, J. Panman³⁸, A. Papanestis^{49,38}, M. Pappagallo⁵¹, L.L. Pappalardo^{16,f}, C. Parkes⁵⁴, C.J. Parkinson^{9,45}, G. Passaleva¹⁷, G.D. Patel⁵², M. Patel⁵³, C. Patrignani^{19, j}, A. Pearce⁵⁴, A. Pellegrino⁴¹, M. Pepe Altarelli³⁸, S. Perazzini^{14,d}, P. Perret⁵, M. Perrin-Terrin⁶, L. Pescatore⁴⁵, E. Pesen⁶⁶, G. Pessina²⁰, K. Petridis⁵³, A. Petrolini^{19, j}, E. Picatoste Olloqui³⁶, B. Pietrzyk⁴, T. Pilař⁴⁸, D. Pinci²⁵, A. Pistone¹⁹, S. Playfer⁵⁰, M. Plo Casasus³⁷, F. Polci⁸, A. Poluektov^{48,34}, E. Polycarpo², A. Popov³⁵, D. Popov¹⁰, B. Popovici²⁹, C. Potterat², E. Price⁴⁶, J.D. Price⁵², J. Prisciandaro³⁹, A. Pritchard⁵², C. Prouve⁴⁶, V. Pugatch⁴⁴, A. Puig Navarro³⁹, G. Punzi^{23,s}, W. Qian⁴, B. Rachwal²⁶, J.H. Rademacker⁴⁶, B. Rakotomiaramanana³⁹, M. Rama¹⁸, M.S. Rangel², I. Raniuk⁴³, N. Rauschmayr³⁸, G. Raven⁴², F. Redi⁵³, S. Reichert⁵⁴, M.M. Reid⁴⁸, A.C. dos Reis¹, S. Ricciardi⁴⁹, S. Richards⁴⁶, M. Rihl³⁸, K. Rinnert⁵², V. Rives Molina³⁶, P. Robbe⁷, A.B. Rodrigues¹, E. Rodrigues⁵⁴, P. Rodriguez Perez⁵⁴, S. Roiser³⁸, V. Romanovsky³⁵, A. Romero Vidal³⁷, M. Rotondo²², J. Rouvinet³⁹, T. Ruf³⁸, H. Ruiz³⁶, P. Ruiz Valls⁶⁴, J.J. Saborido Silva³⁷, N. Sagidova³⁰, P. Sail⁵¹, B. Saitta^{15,e}, V. Salustino Guimaraes², C. Sanchez Mayordomo⁶⁴, B. Sanmartin Sedes³⁷, R. Santacesaria²⁵, C. Santamarina Rios³⁷, E. Santovetti^{24,1}, A. Sarti^{18,m}, C. Satriano^{25,n}, A. Satta²⁴, D.M. Saunders⁴⁶, D. Savrina^{31,32}, M. Schiller⁴², H. Schindler³⁸, M. Schlupp⁹, M. Schmelling¹⁰, B. Schmidt³⁸, O. Schneider³⁹, A. Schopper³⁸, M. Schubiger³⁹, M.-H. Schune⁷, R. Schwemmer³⁸, B. Sciascia¹⁸, A. Sciubba²⁵, A. Semennikov³¹, I. Sepp⁵³, N. Serra⁴⁰, J. Serrano⁶, L. Sestini²², P. Seyfert¹¹, M. Shapkin³⁵, I. Shapoval^{16,43, f}, Y. Shcheglov³⁰, T. Shears⁵², L. Shekhtman³⁴, V. Shevchenko⁶³, A. Shires⁹, R. Silva Coutinho⁴⁸, G. Simi²², M. Sirendi⁴⁷, N. Skidmore⁴⁶, T. Skwarnicki⁵⁹, N.A. Smith⁵², E. Smith^{55,49}, E. Smith⁵³, J. Smith⁴⁷, M. Smith⁵⁴, H. Snoek⁴¹, M.D. Sokoloff⁵⁷, F.J.P. Soler⁵¹, F. Soomro³⁹, D. Souza⁴⁶, B. Souza De Paula², B. Spaan⁹, A. Sparkes⁵⁰, P. Spradlin⁵¹, S. Sridharan³⁸, F. Stagni³⁸, M. Stahl¹¹, S. Stahl¹¹, O. Steinkamp⁴⁰, O. Stenyakin³⁵, S. Stevenson⁵⁵, S. Stoica²⁹, S. Stone⁵⁹, B. Storaci⁴⁰, S. Stracka²³, M. Straticiuc²⁹, U. Straumann⁴⁰, R. Stroili²², V.K. Subbiah³⁸, L. Sun⁵⁷, W. Sutcliffe⁵³, K. Swientek²⁷, S. Swientek⁹, V. Syropoulos⁴², M. Szczekowski²⁸, P. Szczypka^{39,38}, T. Szumlak²⁷, S. T'Jampens⁴, M. Teklishyn⁷, G. Tellarini^{16, f}, F. Teubert³⁸, C. Thomas⁵⁵, E. Thomas³⁸, J. van Tilburg⁴¹, V. Tisserand⁴, M. Tobin³⁹, S. Tolk⁴², L. Tomassetti^{16, f}, D. Tonelli³⁸, S. Topp-Joergensen⁵⁵, N. Torr⁵⁵, E. Tournefier⁴, S. Tourneur³⁹, M.T. Tran³⁹, M. Tresch⁴⁰, A. Trisovic³⁸, A. Tsaregorodtsev⁶, P. Tsopelas⁴¹, N. Tuning⁴¹,

- M. Ubeda Garcia³⁸, A. Ukleja²⁸, A. Ustyuzhanin⁶³, U. Uwer¹¹, C. Vacca¹⁵, V. Vagnoni¹⁴, G. Valenti¹⁴,
- A. Vallier⁷, R. Vazquez Gomez¹⁸, P. Vazquez Regueiro³⁷, C. Vázquez Sierra³⁷, S. Vecchi¹⁶, J.J. Velthuis⁴⁶,
 M. Veltri^{17,h}, G. Veneziano³⁹, M. Vesterinen¹¹, B. Viaud⁷, D. Vieira², M. Vieites Diaz³⁷,
- with vertice 1, G. vertice ratio , M. vesterminen , B. vraudu , D. vreita , With viertes Draz ,
- X. Vilasis-Cardona^{36,p}, A. Vollhardt⁴⁰, D. Volyanskyy¹⁰, D. Voong⁴⁶, A. Vorobyev³⁰, V. Vorobyev³⁴,
- C. Voß⁶², J.A. de Vries⁴¹, R. Waldi⁶², C. Wallace⁴⁸, R. Wallace¹², J. Walsh²³, S. Wandernoth¹¹, J. Wang⁵⁹,
- D.R. Ward⁴⁷, N.K. Watson⁴⁵, D. Websdale⁵³, M. Whitehead⁴⁸, J. Wicht³⁸, D. Wiedner¹¹,
- G. Wilkinson^{55,38}, M.P. Williams⁴⁵, M. Williams⁵⁶, H.W. Wilschut⁶⁵, F.F. Wilson⁴⁹, J. Wimberley⁵⁸,
- J. Wishahi⁹, W. Wislicki²⁸, M. Witek²⁶, G. Wormser⁷, S.A. Wotton⁴⁷, S. Wright⁴⁷, K. Wyllie³⁸, Y. Xie⁶¹,
- Z. Xing⁵⁹, Z. Xu³⁹, Z. Yang³, X. Yuan³, O. Yushchenko³⁵, M. Zangoli¹⁴, M. Zavertyaev^{10,b}, L. Zhang⁵⁹,
- W.C. Zhang¹², Y. Zhang³, A. Zhelezov¹¹, A. Zhokhov³¹, L. Zhong³, A. Zvyagin³⁸.
 - ¹ Centro Brasileiro de Pesquisas Físicas (CBPF), Rio de Janeiro, Brazil
 - ² Universidade Federal do Rio de Janeiro (UFRJ), Rio de Janeiro, Brazil
 - ³ Center for High Energy Physics, Tsinghua University, Beijing, China
 - ⁴ LAPP, Université de Savoie, CNRS/IN2P3, Annecy-Le-Vieux, France
 - ⁵ Clermont Université, Université Blaise Pascal, CNRS/IN2P3, LPC, Clermont-Ferrand, France
 - ⁶ CPPM, Aix-Marseille Université, CNRS/IN2P3, Marseille, France
 - ⁷ LAL, Université Paris-Sud, CNRS/IN2P3, Orsay, France
 - ⁸ LPNHE, Université Pierre et Marie Curie, Université Paris Diderot, CNRS/IN2P3, Paris, France
 - ⁹ Fakultät Physik, Technische Universität Dortmund, Dortmund, Germany
 - ¹⁰ Max-Planck-Institut für Kernphysik (MPIK), Heidelberg, Germany
 - ¹¹ Physikalisches Institut, Ruprecht-Karls-Universität Heidelberg, Heidelberg, Germany
 - ¹² School of Physics, University College Dublin, Dublin, Ireland
 - ¹³ Sezione INFN di Bari, Bari, Italy
 - ¹⁴ Sezione INFN di Bologna, Bologna, Italy
 - ¹⁵ Sezione INFN di Cagliari, Cagliari, Italy
 - ¹⁶ Sezione INFN di Ferrara, Ferrara, Italy
 - ¹⁷ Sezione INFN di Firenze, Firenze, Italy
 - ¹⁸ Laboratori Nazionali dell'INFN di Frascati, Frascati, Italy
 - ¹⁹ Sezione INFN di Genova, Genova, Italy
 - ²⁰ Sezione INFN di Milano Bicocca, Milano, Italy
 - ²¹ Sezione INFN di Milano, Milano, Italy
 - ²² Sezione INFN di Padova, Padova, Italy
 - ²³ Sezione INFN di Pisa, Pisa, Italy
 - ²⁴ Sezione INFN di Roma Tor Vergata, Roma, Italy
 - ²⁵ Sezione INFN di Roma La Sapienza, Roma, Italy
 - ²⁶ Henryk Niewodniczanski Institute of Nuclear Physics Polish Academy of Sciences, Kraków, Poland
 - ²⁷ AGH University of Science and Technology, Faculty of Physics and Applied Computer Science, Kraków, Poland
 - ²⁸ National Center for Nuclear Research (NCBJ), Warsaw, Poland
 - ²⁹ Horia Hulubei National Institute of Physics and Nuclear Engineering, Bucharest-Magurele, Romania
 - ³⁰ Petersburg Nuclear Physics Institute (PNPI), Gatchina, Russia
 - ³¹ Institute of Theoretical and Experimental Physics (ITEP), Moscow, Russia
 - ³² Institute of Nuclear Physics, Moscow State University (SINP MSU), Moscow, Russia
 - ³³ Institute for Nuclear Research of the Russian Academy of Sciences (INR RAN), Moscow, Russia
 - ³⁴ Budker Institute of Nuclear Physics (SB RAS) and Novosibirsk State University, Novosibirsk, Russia
 - ³⁵ Institute for High Energy Physics (IHEP), Protvino, Russia
 - ³⁶ Universitat de Barcelona, Barcelona, Spain
 - ³⁷ Universidad de Santiago de Compostela, Santiago de Compostela, Spain
 - ³⁸ European Organization for Nuclear Research (CERN), Geneva, Switzerland
 - ³⁹ Ecole Polytechnique Fédérale de Lausanne (EPFL), Lausanne, Switzerland
 - ⁴⁰ Physik-Institut, Universität Zürich, Zürich, Switzerland

- ⁴¹ Nikhef National Institute for Subatomic Physics, Amsterdam, The Netherlands
- ⁴² Nikhef National Institute for Subatomic Physics and VU University Amsterdam, Amsterdam, The Netherlands
- ⁴³ NSC Kharkiv Institute of Physics and Technology (NSC KIPT), Kharkiv, Ukraine
- ⁴⁴ Institute for Nuclear Research of the National Academy of Sciences (KINR), Kyiv, Ukraine
- ⁴⁵ University of Birmingham, Birmingham, United Kingdom
- ⁴⁶ H.H. Wills Physics Laboratory, University of Bristol, Bristol, United Kingdom
- ⁴⁷ Cavendish Laboratory, University of Cambridge, Cambridge, United Kingdom
- ⁴⁸ Department of Physics, University of Warwick, Coventry, United Kingdom
- ⁴⁹ STFC Rutherford Appleton Laboratory, Didcot, United Kingdom
- ⁵⁰ School of Physics and Astronomy, University of Edinburgh, Edinburgh, United Kingdom
- ⁵¹ School of Physics and Astronomy, University of Glasgow, Glasgow, United Kingdom
- ⁵² Oliver Lodge Laboratory, University of Liverpool, Liverpool, United Kingdom
- ⁵³ Imperial College London, London, United Kingdom
- ⁵⁴ School of Physics and Astronomy, University of Manchester, Manchester, United Kingdom
- ⁵⁵ Department of Physics, University of Oxford, Oxford, United Kingdom
- ⁵⁶ Massachusetts Institute of Technology, Cambridge, MA, United States
- ⁵⁷ University of Cincinnati, Cincinnati, OH, United States
- ⁵⁸ University of Maryland, College Park, MD, United States
- ⁵⁹ Syracuse University, Syracuse, NY, United States
- ⁶⁰ Pontifícia Universidade Católica do Rio de Janeiro (PUC-Rio), Rio de Janeiro, Brazil, associated to ²
- ⁶¹ Institute of Particle Physics, Central China Normal University, Wuhan, Hubei, China, associated to ³
- ⁶² Institut für Physik, Universität Rostock, Rostock, Germany, associated to ¹¹
- ⁶³ National Research Centre Kurchatov Institute, Moscow, Russia, associated to ³¹
- ⁶⁴ Instituto de Fisica Corpuscular (IFIC), Universitat de Valencia-CSIC, Valencia, Spain, associated to ³⁶
- ⁶⁵ Van Swinderen Institute, University of Groningen, Groningen, The Netherlands, associated to ⁴¹
- ⁶⁶ Celal Bayar University, Manisa, Turkey, associated to ³⁸
- ^a Universidade Federal do Triângulo Mineiro (UFTM), Uberaba-MG, Brazil
- ^b P.N. Lebedev Physical Institute, Russian Academy of Science (LPI RAS), Moscow, Russia
- ^c Università di Bari, Bari, Italy
- ^d Università di Bologna, Bologna, Italy
- ^e Università di Cagliari, Cagliari, Italy
- ^f Università di Ferrara, Ferrara, Italy
- ^g Università di Firenze, Firenze, Italy
- ^h Università di Urbino, Urbino, Italy
- ⁱ Università di Modena e Reggio Emilia, Modena, Italy
- ^j Università di Genova, Genova, Italy
- ^k Università di Milano Bicocca, Milano, Italy
- ¹ Università di Roma Tor Vergata, Roma, Italy
- ^m Università di Roma La Sapienza, Roma, Italy
- ⁿ Università della Basilicata, Potenza, Italy
- ^o AGH University of Science and Technology, Faculty of Computer Science, Electronics and Telecommunications, Kraków, Poland
- ^p LIFAELS, La Salle, Universitat Ramon Llull, Barcelona, Spain
- ^q Hanoi University of Science, Hanoi, Viet Nam
- ^r Università di Padova, Padova, Italy
- ^s Università di Pisa, Pisa, Italy
- ^t Scuola Normale Superiore, Pisa, Italy
- ^u Università degli Studi di Milano, Milano, Italy
- v Politecnico di Milano, Milano, Italy
- ^w Faculty of Physics, University of Sofia, Sofia, Bulgaria